UC San Diego UC San Diego Electronic Theses and Dissertations

Title

Characterizing EGFRvIII-independent glioblastoma maintenance

Permalink

https://escholarship.org/uc/item/1df9w4pj

Author Taylor, Tiffany Elizabeth

Publication Date 2016

Peer reviewed|Thesis/dissertation

UNIVERSITY OF CALIFORNIA, SAN DIEGO

Characterizing EGFRvIII-independent glioblastoma maintenance

A dissertation submitted in partial satisfaction of the requirements for the degree Doctor of Philosophy

in

Biomedical Sciences

by

Tiffany Elizabeth Taylor

Committee in charge:

Professor Frank Furnari, Chair Professor Webster Cavenee Professor Karen Oegema Professor Dwayne Stupack Professor Inder Verma Professor Jing Yang

2016

Copyright

Tiffany Elizabeth Taylor, 2016

All rights reserved.

The Dissertation of Tiffany Elizabeth Taylor is approved, and it is acceptable in quality and form for publication on microfilm and electronically:

Chair

University of California, San Diego

2016

DEDICATION

This dissertation is dedicated to my family, friends and teachers, who have been my greatest sources of support, encouragement, and accountability through every endeavor of my life.

My Lord and my God, I am, because You are and Your grace towards me did not prove vain.

EPIGRAPH

Therefore do not cast away your confidence, which has great reward. For you have need of endurance, so that after you have done the will of God, you may receive the promise.

Hebrews 10:35-36 (NKJV)

TABLE OF CONTENTS

SIGNATURE PAGEiii
DEDICATIONiv
EPIGRAPHv
TABLE OF CONTENTS
LIST OF FIGURES xii
LIST OF TABLES xiv
ACKNOWLEDGEMENTSxv
VITAxviii
ABSTRACT OF THE DISSERTATIONxx
Chapter 1 Introduction to Glioblastoma and the Epidermal Growth Factor Receptor (EGFR)1
Chapter 2 Overview of Established Mechanisms of Resistance to EGFR inhibition
Chapter 3 Overcoming oxidative DNA damage supports escape from dependence on EGFRvIII
Chapter 4 Introduction to the Kelch Repeat Superfamily120
Chapter 5 KLHDC8A attenuates the effects of elevated ROS in the absence of EGFRvIII
Chapter 6 Discussion

LIST OF ABBREVIATIONS

GBM	Glioblastoma
WHO	World Health Organization
CNS	central nervous system
BCNU	carmustine
CCNU	lomustine
ТМΖ	temozolomide
MGMT	O-6 methylguanine DNA methyltransferase
EGFR	epidermal growth factor receptor
ERBB	erythroblastic leukemia viral oncogene
HER	human epidermal growth factor receptor
RTK	receptor tyrosine kinase
тк	tyrosine kinase
РІЗК	phosphatidylinositol 3-kinase
Akt	protein kinase B
МАРК	mitogen-activated protein kinase
СЫ	Casitas B-lineage lymphoma
CR	cysteine rich
BBB	blood brain barrier
СТД	carboxyl terminus domain
GERT	radiotherapy, TMZ, and cetuximab
MAb	monoclonal antibody
MMAF	monomethyl auristatin F

DC	dendritic cell
CTL	cytotoxic T lymphocyte
NSCLC	non-small cell lung carcinoma
RISC	RNA-induced silencing complex
DexAMs	dendritic polyamine complexes
АТР	adenosine triphosphate
Bcl2	B-cell lymphoma 2
Bcl-xL	B-cell lymphoma extra large
uPAS	urokinase plasminogen activator system
BH3	Bcl-2 homology domain 3
HDAC	histone deacetylase
GSC	glioma stem cell
ABC	ATP-binding cassette
ABCB1	ABC subfamily B, member 1
ABCG2	ABC subfamily G, member 2
NSC	normal stem cell
ВМХ	bone marrow tyrosine kinase gene in chromosome X
STAT3	signal transducer and activator of transcription 3
POSTN	periostin
ανβ3	integrin alpha v beta 3
ТАМ	tumor-associated macrophage
TGFβ	transforming growth factor beta
IL-6	interleukin 6
T _{regs}	regulatory T cells

JAK	Janus kinase
ID3	inhibitor of differentiation 3
PDGFRα/β	platelet-derived growth factor receptor alpha/beta
IDH1	isocitrate dehydrogenase 1
NF1	neurofibromatosis type 1
TNF	tumor necrosis factor
CHI3L1	chitinase 3-like 1
MET	hepatocyte growth factor receptor
TP53	tumor protein p53
ЕМТ	epithelial to mesenchymal transition
SCLC	small cell lung cancer
IGF1R	insulin-like growth factor receptor 1
VEGF/R	vascular endothelial growth factor/receptor
mTOR	mammalian target of rapamycin
mTORC1	mTOR complex 1
PTEN	phosphatase and tensin homolog deleted on chromosome 10
PIP ₃	phosphatidylinositol (3,4,5)-trisphosphate
PIK3CA	phosphatidylinositol-4,5-bisphosphate 3-kinase catalytic subunit alpha
KRAS	Kirsten Rat Sarcoma Viral Oncogene Homolog
GTP	guanosine-5'-triphosphate
MEK	MAPK/ERK Kinase
DOX	doxycycline
DSB	double strand break

HDR	homology-directed recombinational repair
NHEJ	non-homologous end joining
HR	homologous recombination
DNA-PKCs	DNA-dependent protein kinase, catalytic subunit
53BP1	p53 binding protein 1
АТМ	ataxia telangiectasia mutated kinase
Ink4a/arf	alternate reading frame protein product of the CDKN2A locus
үН2А.Х	gamma H2A histone family, member X
DMSO	dimethyl sulfoxide
ROS	reactive oxygen species
SSB	single strand break
NFκB	nuclear factor kappa B
KLHDC8A	Kelch domain-containing 8a
ORF1	open reading frame 1
втв	bric a brac, tramtrack, and Broad-complex domain
POZ	portions of zing finger proteins domain
BACK	BTB and C-terminal Kelch domain
CTLH	CRA C-terminal to Lish domain
HGNC	Hugo Gene Nomenclature Committee
BBK	BTB-Back-Kelch
SCF	Skp1-Cullin-F-box
IPP	actin-binding protein IPP
V(D) J	variable, diversity, joining recombination

NRP/B	nuclear-restricted protein/brain
ENC-1	ectodermal-neural cortex
pRb	Retinoblastoma protein
Keap1	Kelch-like ECH-associated protein 1
Nrf2	nuclear factor (erythroid 2-related) factor 2
NADPH	Nicotinamide adenine dinucleotide phosphate
cHL	classical Hodgkin's lymphoma
EBV	Epstein-Barr Virus
UTR	untranslated region
SNP	single nucleotide polymorphism
NLS	nuclear localization sequence
ARE	antioxidant response element
qPCR	quantitative polymerase chain reaction
PDX	patient-derived xenograft
LC-MS/MS	liquid chromatography-mass spectrometry
BirA*	E. coli DNA binding biotin ligase
TRiC/CCT	TCP-1 Ring Complex/ Chaperonin-containing TCP-1
EV	empty vector
INQ	intranuclear quality control compartment
FITC	fluorescein isothiocyanate
TRITC	tetramethylrhodamine
JUNQ	juxtanuclear quality control compartment
HSF	heat shock factor

LIST OF FIGURES

Figure 1.1. Schematic of wild-type EGFR and mutant EGFRvIII protein structures
Figure 2.1. Failure to up-regulate the pro-apoptotic protein BIM is an established mechanism of resistance in glioma
Figure 2.2. In vivo shRNA screen to identify genes that are essential to EGFRvIII-independent tumor maintenance
Figure 2.3. Periostin is an essential gene in EGFRvIII-independence tumor maintenance
Figure 2.4. Summary of the predominant mechanisms of resistance to EGFR inhibition
Figure 3.1. Doxycycline diet effectively silences EGFRvIII expression and glioma growth
Figure 3.2. An increase in the persistence of unrepaired DNA double strand breaks is associated with loss of EGFRvIII alone
Figure 3.3. Gefitinib-mediated EGFRvIII inhibition increases p-H2AX foci in GBM
Figure 3.4. EGFRvIII silencing leads to elevated ROS activity in GBM97
Figure 3.5. Pharmacological inhibition of EGFRvIII is also associated with elevated
Figure 3.6 Increased DNA oxidation correlates with increased DNA damage burden upon EGFRvIII silencing
Figure 3.7. Increased DNA oxidation is associated with pharmacological inhibition of EGFRvIII
Figure 3.8. Overcoming EGFRvIII blockade is reflected by a significant reduction in the persistence of DNA DSBs
Figure 3.9. Escaper populations display significantly less ROS activity105
Figure 3.10 A decrease in DNA oxidation accompanies diminished reliance on EGFRvIII
Figure 3.11. Gefitinib resistance is associated with significantly less ROS activity

Figure 3.12 Diminished reliance on EGFRvIII receptor function is associated with decreased DNA oxidation
Figure 4.1. KLHDC8A is a 7-bladed β -propeller only containing protein135
Figure 5.1. Up-regulation of KLHDC8A is common to in several GBM models.
Figure 5.2. KLHDC8A interacts with the TRiC/CCT chaperonin complex149
Figure 5.3. KLHDC8A is found in the nucleus independent of EGFRvIII in GBM
Figure 5.4. siRNA-mediated knockdown of KLHDC8A promotes the persistence of unrepaired DNA double strand breaks in Escapers154
Figure 5.5. Inhibition of KLHDC8A restores sensitivity to EGFRvIII inhibition in the form of increased mitochondrial ROS levels
Figure 5.6. Cytoplasmic ROS activity is not affected when KLHDC8A is inhibited
Figure 5.7. Loss of KLHDC8A is associated with an increase in oxidative DNA damage
Figure 6.1. Predicted model of KLHDC8A-mediated resistance to EGFRvIII inhibition
Figure 6.2. Some escapers do not rely on intact DNA repair pathways180

LIST OF TABLES

Table 1.1.	EGFR/EGFRvIII-targeted	therapies in	n clinical	development	for the
treatment of	of GBM				24

ACKNOWLEDGEMENTS

I would like to acknowledge my thesis mentor Frank Furnari. Thank you for choosing me back in 2010 to join your group, and for the opportunity to work on what surely became a challenging project. I have grown and learned so much throughout my time in the lab and I appreciate your devotion to being always available to help and your continued support of me chasing my dreams.

I am also very thankful to my committee members who stuck with me throughout the many detours that this project took. At every committee meeting, I always felt challenged, but at the same time very supported. Your advice and instruction were invaluable to getting me to this point. I am especially grateful for Dwayne Stupack, who though he was not even my formal advisor, met with me often, and really sowed into my PhD training immeasurably more than expected.

Thank you to all of my lab members/family, both past and present. Every one of you made my tenure in the lab very memorable and a delight. Thank you for all of your words of encouragement when I needed them most, your genuine care for me as a person first before a researcher, and your selfless offerings to help out with this project, when many of you had very little time to give. I will definitely miss you all.

Thank you to my grandma, aunts, uncles, mom, dad, sisters, brother, cousins, who though you don't know exactly what I do, you believe in me and have always been my biggest cheerleaders. You all keep me grounded and it is because of you that I have the courage to step out into unchartered territory so often.

My time here in San Diego would not have been as rich as it was without my church family and friends. Your continued support and encouragement have truly

ΧV

been a source of strength to me, especially over the past 3 years and I am so blessed to have you all in my life. Thank you for sticking this out with me, and never growing weary of my constant requests for your prayers. There are not enough pages allowed in this dissertation to mention you all by name, but I am most grateful for you individually and corporately.

Special thanks to my roommate Wendy, who has seen the good, the bad, and the ugly days and still thinks so highly of me. Thank you for reminding me of what is important in life, for the daily nourishment ⁽ⁱ⁾, and for your constant encouragement and prayers, when I needed them the most. I couldn't have imagined surviving this past year without you.

Lastly, I would like to thank my best friend Dr. LaTeira Haynes. I really don't know if I could have joyfully made it to this point without your constant support and encouragement. You always helped me to remain hopeful at the times when I was most discouraged. Thank you for making time to refresh me, even in the busyness of your first year of teaching. You are truly a gem that I am most grateful to have discovered during my graduate years.

Chapters 1 contains portions of material as it appears in Current Cancer Drug Targets, Taylor, T. E., Furnari, F. B., Cavenee, W. K. Targeting EGFR for Treatment of Glioblastoma: Molecular Basis to Overcome Resistance. 12:197-209 (2012). The dissertation author is the first author of this review article. This work was supported by the NIH F31 NS076343 to (T.E. Taylor) and the NIH R01 NS080939 to (F. B. Furnari).

Chapter 2 contains portions of material as it appears in Current Cancer Drug Targets, Taylor, T. E., Furnari, F. B., Cavenee, W. K. Targeting EGFR for Treatment of Glioblastoma: Molecular Basis to Overcome Resistance. 12:197-209 (2012). The

xvi

dissertation author is the first author of this review article. Additional data presented in chapter 2 have not been published previously and will not be submitted for publication. This work was supported by the NIH F31 NS076343 to (T.E. Taylor) and the NIH R01 NS080939 to (F. B. Furnari).

Chapter 3 contains data that have not been published previously, and are currently being prepared for submission for publication. The work in chapter 3 was completed with the help of Jianhui Ma, M.D, PhD., and Antonia Boyer of the laboratory of Frank Furnari. This work was supported by the NIH R01 NS08939 to (F.B. Furnari).

Chapter 5 contains data that have not been published previously, and are currently being prepared for submission for publication. The work in chapter 5 was completed with the help of Jill Wykosky, PhD., a former member of the laboratory of Frank Furnari and Jason Liang, a fellow PhD Candidate in the laboratory of Huilin Zhou at the Ludwig Institute for Cancer Research. This work was supported by the NIH F31 NS076343 to (T.E. Taylor) and the NIH R01 NS080939 to (F. B. Furnari).

Chapter 6 contains data that have not been published previously, and are currently being prepared for submission for publication. The dissertation author solely completed the work presented in chapter 6. This work was supported by NIH R01 NS080939 to (F. B. Furnari).

xvii

VITA

Education

2016	Doctor of Philosophy, Biomedical Sciences University of California, San Diego
2008	Bachelor of Science, Biology

Howard University

Publications

Wykosky, J., Hu, J., Gomez, G., **Taylor, T.,** Villa, G., Pizzo, D., VandenBerg, S., Thorne, A., Chen, C., Mischel, P., Gonias, S., Cavenee, W., Furnari, F. (2015). A urokinase receptor-Bim signaling axis emerges during EGFR inhibitor resistance in mutant EGFR glioblastoma. *Cancer Research* **75**: 394-404.

Taylor, T. E., Furnari, F. B., Cavenee, W. K. (2012). Targeting EGFR for Treatment of Glioblastoma: Molecular Basis to Overcome Resistance. *Current Cancer Drug Targets* **12**:197-209. (Review Article)

Rankin EB, Fuh KC, **Taylor TE**, Krieg AJ, Musser M, Yuan J, Wei K, Kuo CJ, Longacre TA, and Giaccia AJ. (2010). Axl is an essential factor and therapeutic target for metastatic ovarian cancer. *Cancer Research* **70**: 7570-7579.

Burnett, G. T., Weathersby, D. C., **Taylor, T. E**., and Bremner, T. A. (2008). Regulation of inflammation- and angiogenesis-related gene expression in breast cancer cells and co-cultured macrophages. *Anticancer Research* **28**: 2093-2100.

Awards and Honors

Grad Slam Champion, UCSD, April 2016

Edward A. Bouchet Graduate Honor Society Membership, UCSD, April 2013

High Scoring Abstract, Society for Neuro-Oncology, November 2012

Outstanding Poster Award, UCSD Biomedical Sciences Graduate Program, September 2012

HHMI Med-Into-Grad Fellow in Hematology/Oncology, UCSD, September 2011-April 2012

Ruth L. Kirschstein Individual Predoctoral NRSA Recipient, NINDS, July 2011

UCSD UJIMA Graduate Student Award, February 2010

Howard University, Summa Cum Laude, May 2008

Phi Beta Kappa Honor Society Membership, Howard University, April 2008

Thurgood Marshall Scholarship, Howard University, September 2006

Phi Sigma Pi National Honor Fraternity, Inc. Alpha Tau Chapter, Howard University, Spring 2006

Legacy Academic Scholarship, Howard University, September 2004-May 2008

ABSTRACT OF THE DISSERTATION

Characterizing EGFRvIII-independent glioblastoma maintenance

by

Tiffany Elizabeth Taylor

Doctor of Philosophy in Biomedical Sciences

University of California, San Diego, 2016

Professor Frank Furnari, Chair

Amplification and mutation of the epidermal growth factor receptor (EGFR) gene are common genetic hallmarks of glioblastoma (GBM). The most common mutation is an in-frame deletion of exons 2-7, resulting in a constitutively active variant of the receptor, EGFRvIII. Indeed, these molecules are proven drivers of gliomagenesis; yet, therapies that are directed against them, such as small molecule tyrosine kinase inhibitors (TKIs), have been ineffective due to both upfront and

acquired drug resistance. In a genetic model of tetracycline-regulated EGFRvIII expression, it was confirmed that this receptor is essential for the maintenance of glioma growth in vivo. However, similar to a clinical situation of acquired drug resistance, some tumors eventually regained aggressive growth, and these breakthrough tumors persisted despite sustained suppression of EGFRvIII. In this study, we establish that tumor recurrence in the absence of EGFRvIII is in part, afforded by the ability to overcome oxidative DNA damage. From the approach of characterizing the phenotypes that are indicative of therapeutic sensitivity, we found that these phenotypes are typically overturned upon relapse. Specifically, we established for the first time that an increase in DNA damage burden reflects sensitivity to EGFRvIII inhibition in GBM, and that populations that overcome the need for EGFRvIII receptor function, recover from this DNA damage. Interestingly, blocking EGFRvIII alone was sufficient to induce significant DNA damage and we determined that an increase in endogenous reactive oxygen species (ROS) activity contributed to this phenotype. Likewise, we also observed that the resistant populations displayed significantly lower ROS levels.

These results suggested that an adaptive response mechanism might have been activated to neutralize ROS, therefore affording escape. A key genetic change that was common in multiple resistant models was up-regulation of KLHDC8A. This gene has already been shown to be required for the in vivo maintenance of some EGFRvIII-independent populations, but a functional role beyond this was unknown. Here, we show for the first time that loss of KLHDC8A restores sensitivity to EGFRvIII inhibition, by causing an increase in ROS activity above the toxic threshold. Consequently, this resulted in increased DNA oxidation and persistent DNA damage

xxi

in the form of double strand breaks. Thus, we have established a novel link between KLHDC8A and ROS homeostasis. Furthermore, our work demonstrates that the characterization of sensitive phenotypes might uncover novel mechanisms that are essential for progressive escape from EGFRvIII blockade, and may also define novel therapeutic windows, whereby greater tumor response to standard of care may be achievable in GBM.

Chapter 1

Introduction to Glioblastoma and the Epidermal Growth Factor Receptor (EGFR)

1.1 Introduction

Each year approximately 29 out of 100,000 adults (aged 20 years or older) are diagnosed with primary brain tumors in the United States, and about one-third of these tumors are malignant (1-3). These malignant tumors represent ~2% of all cancers diagnosed each year and are the cause of 2% of all deaths from cancer in the United States, with a predicted death toll of ~17,000 Americans in 2016 (1,3). Malignant gliomas are among the deadliest of human cancers because they are highly invasive and neurologically destructive (4). The median survival of patients with the most common and most aggressive of these, grade IV glioblastoma (GBM), is 12-15 months, with a 5-year survival rate that remains at less than 5%, despite the use of intensive treatment modalities (4).

The World Health Organization (WHO) classification of Tumors of the Central Nervous System (CNS) distinguishes gliomas based on their histological appearances, where the grade indicates the level of malignancy (5-7). Specific histological features that distinguish GBMs include necrotic areas surrounded by anaplastic cells (pseudopalisading necrosis) and the presence of hyperplastic blood vessels that facilitate microvascular proliferation (6,8). These features contribute to the therapeutically intractable nature and rapid lethality associated with this tumor grade (4,6).

GBMs account for 60-70% of the malignant gliomas diagnosed in American adults between ages 46-74, and are more frequently diagnosed in men than in women (1). Two major subclasses of GBM (primary and secondary) have been established based on the clinical properties and the chromosomal and genetic aberrations that are unique to each (4,9). Primary GBM arises *de novo* from normal glial cells, or their precursors, and commonly occurs in patients above the age of 45 (2). In contrast, secondary GBM arises from the progressive transformation of lower grade tumors, and are generally seen in younger patients (4). Primary GBM is believed to account for 95% of all GBMs, while only 5% are believed to occur secondarily, and so with consideration for the predominant clinical situation, our work addresses primary GBM cases (10).

1.2 Standard of Care

Standard treatment for almost all GBM cases begins with surgery, with the goal of achieving gross total resection of the tumor to alleviate GBM symptoms and to facilitate treatment of any residual tumor (11). Yet, even with the recent technological developments in surgical techniques, the vast majority of patients are not cured by surgical resection. Tumors often infiltrate the normal brain parenchyma, and this invasive nature necessitates the use of adjuvant radiotherapy that is either combined with or is subsequently followed by chemotherapy (4). This regimen was established following results from pivotal clinical trials conducted in the 1970s and 1980s to evaluate the benefits of adjuvant therapy, where it was demonstrated that patients with malignant glioma treated after surgery with radiotherapy, or radiotherapy in combination with chemotherapy, displayed an increase in overall survival (12,13). Radiotherapy involves the administration of usually 50 to 60 Gy of irradiation to the whole brain following surgery, and for several decades, nitrosoureas, particularly carmustine (BCNU) and lomustine (CCNU), were the most common chemotherapeutics used alone or in combination with radiotherapy (11,14,15).

In 1999, a novel chemotherapy drug, temozolomide (TMZ), was approved for the treatment of GBM in combination with radiotherapy, and soon became a candidate for the standard treatment of other solid cancers (16-18). Temozolomide is an alkylating agent and its principle mechanism of action is to induce abnormal methylation of DNA bases, such as the formation of O6-methylguanine in DNA (16). This DNA adduct is resistant to successful excision by DNA mismatch repair enzymes, which leads to the production of single and double-strand breaks in the DNA and subsequent activation of cell killing pathways (16,19). Clinical studies later demonstrated that the median survival in newly diagnosed GBM patients treated with radiation plus TMZ was 14.6 months as opposed to 12.1 months for the radiation only group (20,21). Furthermore, a determinant of the survival advantage observed in GBM patients was found to be the methylation status of a DNA repair gene, O-6 methylguanine DNA methyltransferase (MGMT), in which GBMs containing a methylated MGMT promoter, which silences this gene, was associated with the most favorable prognosis (22,23).

The clinical benefit acquired through the addition of TMZ to the standard treatment strategy for GBM signified a therapeutic breakthrough that had not been experienced in the field in over two decades. Yet, despite such an incredible improvement to the adjuvant therapy regimen, the overall survival rate for GBM patients has remained relatively the same as it was more than fifteen years ago. Tumors inevitably recur, and once they do, life expectancy drastically diminishes and only limited therapeutic options are available.

Major advances in molecular and cell biology have afforded the development of novel second and third line therapies for cancer patients in the realm of targeted therapy. So far, targeted therapy has been used effectively against breast tumors driven by growth factor receptor dysregulation, and in the case of GBM, therapies directed against a number of growth factors and their respective receptors are in various stages of clinical development (24-33). The most common and most clinically advanced of these therapies target the epidermal growth factor receptor gene (EGFR) and a mutant form of this receptor, EGFRvIII. The following sections will explore these two molecules and the current status of the major targeted therapies that have been developed against them to treat GBM (**Table 1.1**).

<u>1.3 The Epidermal Growth Factor Receptor (EGFR)</u>

EGFR (also referred to as ERBB1 or HER1) is a member of the HER superfamily of receptor tyrosine kinases (RTKs) together with ERBB2, ERBB3, and ERBB4 (34). The structure of each member comprises: a ligand-binding ectodomain containing two cysteine-rich regions; a single transmembrane region; and a cytoplasmic tyrosine kinase (TK) domain (Figure 1.1). Binding of a cognate ligand to ligand-binding receptors the site of HER induces receptor homoor heterodimerization, resulting in a conformational change that activates the intracellular TK domain. This results in autophosphorylation of the cytoplasmic tail of the receptor, induction of downstream signaling (through the phosphatidylinositol 3kinase (PI3K)/Akt and the ras-raf-mitogen-activated protein kinase (MAPK) pathways, among others), and transcription of genes controlling pleiotropic cellular responses (34).

The most common ligands for HER receptors are members of the EGF family of growth factors (i.e., epidermal growth factor, heparin binding EGF-like growth factor, amphiregulin, epiregulin, epigen, betacellulin, and transforming growth factor a) (35). Interestingly, there is no known ligand for ERBB2, which is believed to undergo ligand-independent activation (34). HER receptors are localized at the surfaces of many types of epithelial, mesenchymal, and neuronal cells such that signal transduction from these receptors into the intracellular compartment actuates several cellular processes including cell differentiation, metabolism, proliferation, and survival (36).

EGFR was the first receptor to be sequenced and discovered to possess tyrosine kinase activity (36). In 1984, it was revealed that the sequence of EGFR was

closely related to that of a known oncoprotein, the *erbB* tyrosine kinase, previously revealed to be associated with the onset of erythroleukemia (37). Since then, EGFR has been shown to be frequently overexpressed or hyperactivated in a number of tumors (9). Indeed, amplification of EGFR is the most common genetic aberration associated with primary GBM, with a frequency of about 50% in newly diagnosed cases (4). The downstream signaling effects of such aberrations lead to impaired apoptosis, enhanced proliferation, angiogenesis, necrosis, metastatic spread and treatment refractoriness, suggesting a causative relationship between receptor dysregulation and the pathobiology of many cancers (38).

1.4 EGFR Mutations in GBM

Other mechanisms besides gene amplification are implicated in EGFR dysregulation in primary GBM, such as intrinsic alterations of the receptor structure as a result of mutation (i.e., missense and deletions). EGFR is among the most frequently mutated genes in GBM and recent RNA sequencing efforts have afforded comprehensive characterization of these alterations, which vary in frequency, type and structure (39). A number of EGFR point mutations: R108K, A289V/D/T, and G598V among others, have been identified and are predominantly found within the extracellular domain of the receptor (39). These missense mutations are encountered at an allelic frequency of about 14% in GBM and are predicted to promote an active EGFR conformation, thereby affording constitutive receptor activity (39,40).

Deletion mutations in EGFR have also been described, and the variant diversity that arises from such somatic alterations contributes to the RTK heterogeneity that is inherent to primary GBM. An extensive truncation mutation in the amino-terminal domain of EGFR gives rise to the type I variant (EGFRvI), while deletion of exons 14-15 produces the type II variant (EGFRvII) (40). The clinical relevance of EGFRvI is still undefined, but more recent re-characterization of the EGFRvII variant through single-cell genomic analysis and other functional studies revealed that it is indeed oncogenic, overturning previous claims of its irrelevance in gliomagenesis (41). The cytoplasmic tail mutants, EGFRvIV (exons 25-27 deletion) and EGFRvV (exons 25-28 deletion) are the least frequently encountered variants, but are exclusive to GBM. Additionally, these deletion mutations confer protection from ubiquitin-mediated degradation, likely due to the absence of the Tyrosine (Y) 1045 residue (found in exon 27) that is engaged by Casitas-B-lineage proteins (Cbl)

to promote receptor turnover (42).

The most common EGFR mutant encountered in about 30% of primary GBMs is EGFRvIII (also referred to as de2-7EGFR or Δ EGFR). It is naturally co-expressed with amplified wild-type EGFR (wtEGFR) in about 50% of GBMs, but can also be present independently of this aberration (43-45). EGFRvIII arises through an in-frame deletion of 801 base pairs that encode amino acids 6-273 (exons 2-7) of the extracellular domain of EGFR (40,46). Additionally, this truncation produces a novel glycine residue insertion and confers ligand-independent constitutive, but low, tyrosine kinase activity (46) (**Figure 1.1**). Disrupted ligand binding capacity and low level signaling, especially reduced phosphorylation at Y1045, afford evasion from Cbl-mediated receptor internalization, rendering increased stability of EGFRvIII at the cell surface and amplified mitogenic effects (42,47).

EGFRvIII is tumor-specific, as it has not been found in normal tissues, and its overexpression is associated with enhanced tumorigenicity, resistance to radio- and chemotherapy, and the maintenance of intratumoral heterogeneity (42,46,48-51). Additionally, a number of its oncogenic properties are shared with wtEGFR such as promoting proliferation, angiogenesis, tumor invasion and reducing apoptosis (52-54). EGFRvIII is heterogeneously expressed among cells within a given tumor, being detected at both very high and very low protein levels in GBM (55). At the DNA level, it is also found to be amplified in the form of double minutes, which are circularized double-stranded DNA fragments located outside of chromosomes (56). Such intratumoral heterogeneity supports the notion that different genetic subdomains within GBM tumors purposefully exist, perhaps, to contribute to the tumor's overall fitness and to cultivate a drug-resistant tumor microenvironment.

Indeed, previous work from our lab has demonstrated that inter-clonal cooperation between populations of tumor cells is a predominant mechanism by which GBM aggressive growth is maintained (50). Mutant EGFRvIII was shown to drive this process through the "impartation" of its own intrinsic tumorigenic abilities, via a cytokine circuit, to the less aggressive neighboring cells, the majority of which expressed amplified wtEGFR (50). In the case of EGFRvIII on double minutes, the heterogeneous nature of its expression here is entirely distinct from the mechanism that is observed in the absence of EGFR-specific treatments. Surprisingly, the tumor's ability to rapidly eliminate EGFRvIII from double minutes affords an advantage to evade therapies that target oncogenes residing on extrachromosomal DNA (56). Thus, it appears that the dynamics of EGFRvIII expression within GBM tumors is regulated such that its loss on extrachromosomal DNA upon exposure to drug and its concentrated expression in only a subpopulation of cells are both optimal conditions for growth and survival of the tumor mass. Nevertheless, the presence of EGFRvIII confers sensitivity to EGFR-targeting agents, suggesting that specifically targeting this molecule is still a rational therapeutic strategy (57,58).



Figure 1.1. Schematic of wild-type EGFR and mutant EGFRvIII protein structures. EGFR is composed of three distinct domains: a ligand-binding extracellular domain containing two cysteine-rich (CR) regions, a transmembrane domain, and a cytoplasmic tyrosine kinase catalytic domain. Mutant EGFRvIII retains the intact intracellular architecture of EGFR, but much of the extracellular region is absent due to the 801 base pair in-frame deletion. Additionally, a distinguishable novel glycine residue exists in this mutant.

1.5 EGFR-Targeting Monoclonal Antibodies

Cancer immunotherapy seeks to manipulate a person's own immune system to recognize and specifically destroy tumor cells, using target-specific antigenic proteins and peptides (59). Although tumor immunotherapy has shown some success in the treatment of renal cell carcinoma, melanoma, and hematologic cancers, the application of this approach to glioma presents more of a challenge (59). Limitations include hindrance in the drug's ability to cross the blood-brain barrier (BBB) and induction of potential autoimmunity, which may lead to severe and undesired effects, such as central nervous system toxicity (59,60).

In the early 1900s, Paul Erhlich was the first to propose the process of using monoclonal antibodies to target tumors, and later advances in antibody technology afforded the production of human monoclonal antibodies (60,61). One approach to inhibit EGFR-mediated signaling is to disrupt receptor-activating ligand binding (60). Blocking a ligand from binding to its cognate receptor could normalize growth rates, induce apoptosis, and increase tumor susceptibility to chemotherapeutic agents. Monoclonal antibodies, both unconjugated and conjugated, directed towards wtEGFR and EGFRvIII have been developed for therapeutic use in GBM. The most developed of the unconjugated antibodies is cetuximab (Erbitux[®]; Merck KGaA), which functions to prevent EGFR-mediated signal transduction by interfering with ligand binding and EGFR extracellular dimerization (60). Additionally, cetuximab is believed to trigger EGFR receptor internalization and destruction (62).

Mouse xenograft (intracranial and subcutaneous) studies demonstrated that treatment with cetuximab decreases tumor proliferation and increases cell death and overall survival (63). Yet, other pre-clinical data suggested that treatment with cetuximab alone had minimal impact on glioma tumorigenicity; but was synergistic with cytostatic drugs and radiotherapy (62,64). Based on the latter data, the efficacy of combining radiotherapy, TMZ and cetuximab, together known as GERT, in treating patients with primary GBM is being clinically evaluated (65). More recently though, cetuximab alone was shown to be effective against GBM tumors harboring a newly identified exon 27 deletion mutation in the carboxyl-terminus domain of EGFR referred to as EGFR-CTD (66). Particularly, cetuximab impaired tumorigenicity and prolonged the survival of xenograft mice harboring the oncogenic EGFR-CTD deletion mutant (66).

Antibodies are large in molecular weight; but despite this fact, these data suggest that cetuximab can traverse the blood-brain barrier. Thus, while cetuximab alone has on occasion displayed limited clinical effectiveness among GBM patients, it may be a more promising therapeutic for patients with GBMs specifically harboring the EGFR-CTD deletion mutation. Furthermore, it appears that patient stratification will be required to confidently establish the determinants of drug efficacy. Other unconjugated monoclonal antibodies in clinical development to treat GBM include panitumumab (Vectibix[®]; Amgen) and nimotuzumab (Theraloc[®]; YM BioSciences Inc.), which all function similarly to cetuximab (9,60). So far in glioma clinical trials, panitumumab does not effectively cross the blood brain barrier and nimotuzumab is associated with nervous system toxicity (67). Therefore, the fate of these therapies as treatments for GBM is uncertain.

The specific binding properties of antibodies can be exploited to deliver the toxic properties of either toxins or radioisotopes to target sites, and are referred to as conjugated antibodies. ¹²⁵I-MAb 425 is one of the most advanced of the radioisotope-
conjugated monoclonal anti-EGFR antibodies. Various phase II clinical trials have reported that ¹²⁵I-MAb 425, either administered alone or concomitant with radiotherapy or temozolomide, significantly improves median survival in GBM patients (68-70). The ligand-conjugate toxin composed of EGF and diphtheria toxin, DAB389EGF, represents another means to specifically transport toxins to the site of EGFR, and involves the use of cognate ligands as vectors. Results from studies assessing DAB389EGF are encouraging, in which significant tumor regression in subcutaneous glioblastoma xenografts was observed (71). Nonetheless, it still remains to be determined if the performance of these agents in the clinical setting will lead to long-lasting, desirable outcomes for patients.

Some monoclonal antibodies have been engineered to specifically target EGFRvIII. One such antibody is mAb806. which attenuates receptor autophosphorylation by binding to the short cysteine loop of the extracellular domain that is always exposed in EGFRvIII, but may also weakly target amplified wtEGFR, which transiently exposes this epitope during the switch from the inactive to the ligand-activated conformation (72-74). Surprisingly, it does not bind wtEGFR expressed on normal cells, representing a promising feature of this therapy, since many unwanted side effects may be alleviated (75). Pre-clinical data showed that mAb806 strongly inhibits the growth of tumor xenografts that express EGFRvIII and as expected, more weakly those that express wtEGFR (76,77).

To enable clinical development, a humanized version of mAb806 designated ABT-806 (Abbott) was generated (78). Pre-clinical characterization of ABT-806 demonstrated that its antitumor activity is superior to cetuximab against EGFRvIII-expressing tumors, and in the case of wtEGFR-expressing tumors, it is equal in

potency, but with less toxicity (78). These results prompted a Phase I clinical trial with ABT-806, and so far, it is well tolerated and displays excellent bio-distribution and specificity for its target in patients with advanced solid tumors, including glioblastoma (78,79). To date, it is also being clinically evaluated as a monomethyl auristatin F (MMAF) conjugated version called ABT-414, where the conjugate drug is a microtubule depolymerization agent (80). Therefore, both mitotic arrest and selective killing of tumor cells expressing high levels of wild type and mutant forms of EGFR can be achieved. ABT-414 was shown to be effective against recurrent or unresectable GBMs and a phase II clinical trial evaluating its efficacy against newly diagnosed GBMs when combined with TMZ and radiotherapy is currently underway. Of note though, GBMs that expressed low levels of these receptors responded poorly to ABT-414, indicating that EGFR amplification status is key in predicting the best clinical response (80).

1.6 EGFR-Targeting Vaccines

Antitumor vaccines have also been developed with the promise of precisely eradicating tumor cells while limiting toxicity. To date, vaccines in clinical development for the treatment of glioma consist of different combinations of dendritic cells (DCs), peptides, adjuvants, and even autologous tumors (59). The most encouraging of these vaccines are dendritic cell and peptide-based; other combinations have resulted in either no significant improvement over standard therapy or the induction of several adverse events (59). The EGFRvIII–specific vaccines are directed against the novel glycine epitope at the fusion junction that arises as a consequence of the in frame deletion of exons 2-7 from the extracellular domain of wtEGFR (81). The proposed mechanism of action for achieving tumor regression with these therapies begins with capture of the antigenic peptide by the antigen-presenting cell. The peptide is then relocated into the most proximal lymph nodes, where it is presented to circulating cytotoxic T-lymphocytes (CTLs). The CTLs are then activated upon recognition of the antigenic peptide via its T-cell receptor, and finally infiltrate the tumor to eliminate the respective cancer cells (82,83).

Tumorigenicity studies in rodents show that intracerebral treatment with an EGFRvIII synthetic peptide conjugated to keyhole limpet hemocyanin (rindopepimut/PEP-3-KLH/CDX-110[®], Celldex Therapeutics) reduces tumor size and increases overall survival (84). Prior to its development, major clinical responses were rarely obtained from previous peptide-based vaccine studies, but rindopepimut in combination with TMZ has been shown to improve both progression free survival (14.2 months vs 7.3 months) and median survival (26 months vs. 15.2 months) in GBM patients (9,82). Additionally, a vaccine comprised of DCs pulsed with

rindopepimut has been tested in the clinic and was shown to increase overall median survival time (22.8 months vs. 15.2 months) without severe adverse events (85). These results confirmed both the safety of rindopepimut and its ability to induce EGFRvIII-specific immune responses in patients with newly diagnosed primary GBM.

A vaccine that specifically recognizes wtEGFR has also been designed, in which a wtEGFR binding peptide is conjugated with a lytic-type peptide containing cationic-rich amino acids that kills the cancer cell by disintegration of the cell membrane (86). This vaccine was designed to serve as an adjuvant for either monoclonal antibodies or tyrosine kinase inhibitors to more effectively eradicate the cancer cells that are intractable to signaling inhibition, a characteristic that renders them resistant to these single-based therapy approaches. This EGFR-lytic peptide can destroy cancer cells as quickly as ten minutes after exposure and exerts strong cytotoxic activity on TKI-resistant glioma cells (86). Though this therapy has not been tested specifically in a glioblastoma xenograft model, it was also shown to be effective in suppressing the growth of breast, pancreatic, and oesophageal xenografts (86,87). Furthermore, other evidence has demonstrated that with just a single amino acid replacement in the lytic peptide sequence, specifically a histidine to an arginine, its anticancer activity is enhanced (88). Taken together, these data suggest a potential value for this therapy specifically against wtEGFR-expressing GBMs.

These pre-clinical and early clinical results are very encouraging; yet, a number of obstacles regarding the efficacy of epitope-specific vaccines remain. For instance, the likelihood that targeting a single heterogeneously expressed tumor antigen could potentially select for the survival and outgrowth of antigen-negative cells. In fact, in a multicenter phase II clinical trial to assess the immunogenicity of the

rindopepimut vaccine alone, tumor recurrence following a significant period of progression-free survival was observed (89). Indeed, 82% of the relapse tumors were completely EGFRvIII-negative, demonstrating that this vaccine can effectively eliminate EGFRvIII-expressing cells, but as a monotherapy, it lacks the long-term efficacy that is needed to eradicate this disease (89,90). Thus, overcoming the inherent heterogeneity associated with GBM demands that emerging therapeutic strategies meet criteria that go beyond antigen-specificity.

1.7 EGFR-Targeting RNA

Interference with transcription or translation is another mechanism through which receptor inhibition may be achieved, and some methods that have been developed over the last decade include antisense RNA, RNA interference (RNAi), and ribozymes (91). Antisense oligonucleotides, such as OGX-011, are already in advanced clinical development for the treatment of NSCLC and prostate cancer, and thus far, the clinical responses are very promising (92-94). These RNAs hybridize to the sense mRNA of the target, resulting in inhibition of translation and protein synthesis.

In an orthotopic xenograft model of human glioblastoma, intratumoral injection of a plasmid or viral vector expressing EGFRvIII-targeted antisense RNA was shown to cause a significant decrease in tumor growth compared with controls (95). In RNA interference methods, the suppression of homologous genes by small interfering RNAs (siRNAs) leads to sequence-specific target mRNA degradation. EGFR-specific siRNAs are directed against the TK domain, and were shown to cause 90% knockdown of EGFR mRNA in U251 glioma cells (96). Furthermore, siRNA mediatedknockdown of EGFR resulted in G2/M arrest and reduced proliferation (96). These findings were confirmed in an intracranial xenograft model, where treatment with EGFR-specific siRNAs increased overall survival by almost 90% (96).

One caveat of this approach is the overall safety of siRNAs as therapeutics, given that in other *in vivo* studies, a vast number of mice fatalities were observed, due to oversaturation of RNAi pathways (91). A strategy in place to overcome this obstacle is to use the lowest possible concentration of siRNAs that provides therapeutic efficacy by designing exogenous siRNAs with increasing length (91). This

would introduce them into the RNAi pathway upstream of the RNA-induced silencing complex (RISC) directly at the step of Dicer cleavage, resulting in enhanced RNAi activity at lower concentrations (91). Another strategy in place is the use of cyclodextrin-modified dendritic polyamine complexes (DexAMs) as a vehicle for translocating siRNAs (97). So far, DexAMs have been shown to deliver EGFRvIII siRNAs efficiently and selectively to glioblastoma cells with minimal toxicity (97). Furthermore, co-delivery of EGFRvIII-siRNA and the EGFR tyrosine kinase inhibitor erlotinib was found to significantly inhibit cell proliferation and induce apoptosis in glioblastoma cells (97).

A third method of RNA-based interference that has been explored involves the use of anti-EGFRvIII hairpin ribozymes. Ribozymes catalytically cleave certain RNA substrates in a sequence-specific manner, in which cleavage is mediated by a catalytic core (9). In pre-clinical *in vitro* studies, treatment with anti-EGFRvIII hairpin ribozymes was shown to reduce EGFRvIII mRNA by 90% and inhibit anchorage-independent growth of U87MG-EGFRvIII glioma cells (98). These encouraging pre-clinical outcomes along with the success of RNA-based approaches in other cancers suggest that upon optimization and refining, these therapies should be further explored in clinical trials for the treatment of GBM.

<u>1.8 EGFR-Targeting Tyrosine Kinase Inhibitors</u>

Small molecule tyrosine kinase inhibitors (TKIs) are the most clinically advanced of the EGFR-targeted therapies, and both reversible and irreversible inhibitors are in clinical trials. Some of the reversible inhibitors include erlotinib (Tarceva[®]/OSI774; Genentech/Roche/OSI), gefitinib (Iressa[®]/ZD1839; AstraZeneca), lapatanib (Tykerb[®]; GlaxoSmithKline) and PKI166 (Novartis), and the irreversible inhibitors include canertinib (Cl1033; Pfizer/Warner-Lambert) and pelitinib (EKB-569; Wyest-Ayerst) (27). Mechanistically, these TKIs compete with ATP for binding to the tyrosine kinase domain of EGFR (27). The irreversible and reversible nature of these inhibitors lead to the ablation of both phosphorylation of the receptor and downstream signaling (9). Though a number of these inhibitors have been developed, gefitinib erlotinib, and lapatanib represent the most explored TKI inhibitors in the clinical setting for the treatment of GBM.

In pre-clinical *in vitro* studies, erlotinib was shown to inhibit anchorageindependent growth of glioblastoma cell lines (99). More importantly, this inhibition was shown to correlate with suppressed induction of EGFR mRNA (99). Additionally, long-term exposure to erlotinib was found to down-regulate the expression of both EGFRvIII and molecular effectors of tumor invasion in transformed glioblastoma cell lines (100). The efficacy of erlotinib; however, is more characterized in other cancer cell types, where it has been shown to inhibit cell-cycle progression by inducing G_1/S phase arrest (101,102). Moreover, erlotinib is able to induce apoptosis in colon and pancreatic cancer cell lines by stimulating DNA fragmentation and decreasing the expression of the anti-apoptotic Bcl-2 family members, respectively (101,102). Data from *in vivo* studies demonstrate that erlotinib also displays anti-angiogenic activity by suppressing vessel formation in pancreatic tumor xenografts (102).

Erlotinib was first clinically tested for the treatment of advanced and metastatic NSCLC, and was shown to significantly improve median survival in patients with these tumors by 42.5% in a phase III randomized trial (9). Yet in a more recent phase II clinical trial for GBM therapy, erlotinib was well tolerated, but only demonstrated a modest effect over placebo (103). These results underscore the notion that differences in tissue-specific biology and/or signaling networks coupled to EGFR greatly influence TKI efficacy. Nevertheless, in the same study testing the efficacy of cetuximab against GBM tumors harboring novel EGFR mutations, erlotinib was also shown to be very effective against the EGFR CTD mutant-expressing GBMs (66).

In the case of gefitinib, its antitumor activity is independent of the expression level of EGFR, but is heavily impacted by its ability to inhibit anti-apoptotic signals (57). In ways similar to erlotinib, it is effective at inhibiting the *in vitro* growth of a variety of human cancer cell (104). Additionally, treatment with gefitinib inhibits the survival and proliferative capacity of adenocarcinoma cancer cells and induces G_0/G_1 arrest in pancreatic cancer cells (102). In the *in vivo* context, gefitinib has been shown to hinder the growth of human breast and ovarian tumor xenografts (104).

Despite its success in other cancer cell types, gefitinib has been less effective against glioblastoma. Similar to erlotinib, the clinical efficacy of gefitinib fails to go beyond the pre-clinical studies, in which in one phase II clinical trial, gefitinib was well tolerated and displayed anti-tumor activity, but the median overall survival time in GBM patients was only 38.4 weeks from treatment initiation (105). A more recent phase II trial revealed that gefitinib reaches high concentrations in tumor tissue and efficiently dephosphorylates its target; however, more dominant regulatory circuits that promote sustained downstream signal transduction independent of EGFR phosphorylation nullify these effects (30). Gefitinib and erlotinib appear to work best against tumors expressing EGFR with mutations in exons 19 and 21 of the TK domain, but to date, such EGFR mutants have not been found in GBM (9).

Lapatanib is a third tyrosine kinase inhibitor to be explored as a treatment option for GBM. It is currently approved for use in the treatment of HER2-positive breast cancer cells and has generated encouraging results when used as a combination therapy with capecitabine (106). This success led to a phase I/II clinical trial to assess response rate, and pharmacokinetics of lapatinib in patients with recurrent GBM (107). The results from this study revealed that lapatanib alone displays no significant antitumor activity in GBM patients. Yet, not all hope in lapatanib has been lost, as it is currently being evaluated in the context of newly diagnosed GBM, as part of a combination therapy with TMZ and radiation. Perhaps this approach will match the efficacy achieved in breast cancer, but again even these results will not negate the fact that these agents as monotherapies are not durable in GBM. Additionally, though the molecular weights of small molecule inhibitors are within the size limit for molecules that are allowed to cross the BBB, recent studies have shown that plasma concentrations of gefitinib and erlotinib following therapy were only 6-11% of the starting dose (9,58). Thus, insufficient delivery of drug to the target site may be another cause of the disappointing clinical responses to these TKIs.

Table 1.1. EGFR/EGFRvIII-targeted therapies in clinical development for the treatment of GBM. Included here are the major anti-EGFR agents: monoclonal antibodies, tumor-antigen specific peptide vaccines, and small molecule tyrosine kinase inhibitors, along with their current status in clinical trials.

Phase II ongoing	GlaxoSmithKline	EGFR/HER2	Tykerb	Lapatinib	
Phase II completed	Genentech Inc.	EGFR/EGFRvIII	Tarceva	Erlotinib	
Phase II completed	AstraZeneca Pharmaceuticals	EGFR	Iressa	Geftinib	Tyrosine Kinase Inhibitors
Phase III (w/ GM-CSF + TMZ) ongoing	Celldex Therapeutics	EGFRvIII	Rindopepimut/ CDX-110	PEP-3-KLH	Vaccines
Phase II (w/radiation +TMZ) ongoing	AbbVie	EGFR/EGFRvIII	ABT-414	mAb 806- MMAF	
Phase II completed	Fox Chase Cancer Center	EGFR		¹²⁵ I-MAb 425	
Phase III (w/ radiation + TMZ) completed	YM BioSciences Inc.	EGFR	Theraloc	Nimotuzumab	
Phase II (w/ Irinotecan) terminated	Amgen	EGFR	Vectibix	Panitumumab	
Phase I/II (as GERT) ongoing	Merck KGaA	EGFR	Erbitux	Cetuximab	Monoclonal Antibodies
Status in Clinical Trials	Company/Institution	Target(s)	Brand Name	Substance	Therapy

<u>1.9 Summary and Key Questions</u>

Both wtEGFR and EGFRvIII are bonafide oncogenes that are prevalent in primary GBM, making them attractive targets for therapeutic strategies. Indeed, EGFR overexpression is a poor prognostic factor, correlating with decreased overall survival in GBM patients, and the presence of EGFRvIII also confers a less favorable prognosis in these patients (45,108). Yet, the clinical outcome for GBM patients treated with the many anti-EGFR agents remains poor due to both upfront and acquired resistance. Thus, while knowledge of inherent genetic alterations is pertinent in determining rational therapeutic targets, the monotherapies that have emerged from this knowledge are inadequate for generating a durable response in GBM patients.

GBM heterogeneity supports the likelihood that there are secondary factors that become relevant once the primary oncogenic event has been silenced. The identities of the key factors involved in tumor recurrence following EGFR/EGFRvIII inhibition in GBM have not been thoroughly established. Lack of such knowledge is a critical issue, because without it, the development of efficacious combination therapies will be limited. In this dissertation, we will explore the adeptness of GBMs at overcoming blockade of EGFRvIII receptor function and the predominant "escape" routes that should be considered when exploring strategies to combat therapeutic resistance.

Chapters 1 contains portions of material as it appears in Current Cancer Drug Targets, Taylor, T. E., Furnari, F. B., Cavenee, W. K. Targeting EGFR for Treatment of Glioblastoma: Molecular Basis to Overcome Resistance. 12:197-209 (2012). The dissertation author is the first author of this review article. This work was supported by the NIH F31 NS076343 to (T.E. Taylor) and the NIH R01 NS080939 to (F. B. Furnari).

1.10 References

- 1. Wen, P. Y., and Kesari, S. (2008) Malignant gliomas in adults. *The New England journal of medicine* **359**, 492-507
- Ladha, J., Donakonda, S., Agrawal, S., Thota, B., Srividya, M. R., Sridevi, S., Arivazhagan, A., Thennarasu, K., Balasubramaniam, A., Chandramouli, B. A., Hegde, A. S., Kondaiah, P., Somasundaram, K., Santosh, V., and Rao, S. M. (2010) Glioblastoma-specific protein interaction network identifies PP1A and CSK21 as connecting molecules between cell cycle-associated genes. *Cancer research* 70, 6437-6447
- Ostrom, Q. T., Gittleman, H., Fulop, J., Liu, M., Blanda, R., Kromer, C., Wolinsky, Y., Kruchko, C., and Barnholtz-Sloan, J. S. (2015) CBTRUS Statistical Report: Primary Brain and Central Nervous System Tumors Diagnosed in the United States in 2008-2012. *Neuro-oncology* 17 Suppl 4, iv1-iv62
- Furnari, F. B., Fenton, T., Bachoo, R. M., Mukasa, A., Stommel, J. M., Stegh, A., Hahn, W. C., Ligon, K. L., Louis, D. N., Brennan, C., Chin, L., DePinho, R. A., and Cavenee, W. K. (2007) Malignant astrocytic glioma: genetics, biology, and paths to treatment. *Genes & development* 21, 2683-2710
- 5. Scheithauer, B. W., Erdogan, S., Rodriguez, F. J., Burger, P. C., Woodruff, J. M., Kros, J. M., Gokden, M., and Spinner, R. J. (2009) Malignant peripheral nerve sheath tumors of cranial nerves and intracranial contents: a clinicopathologic study of 17 cases. *Am J Surg Pathol* **33**, 325-338
- 6. Maher, E. A., Furnari, F. B., Bachoo, R. M., Rowitch, D. H., Louis, D. N., Cavenee, W. K., and DePinho, R. A. (2001) Malignant glioma: genetics and biology of a grave matter. *Genes & development* **15**, 1311-1333
- Louis, D. N., Perry, A., Reifenberger, G., von Deimling, A., Figarella-Branger, D., Cavenee, W. K., Ohgaki, H., Wiestler, O. D., Kleihues, P., and Ellison, D. W. (2016) The 2016 World Health Organization Classification of Tumors of the Central Nervous System: a summary. *Acta neuropathologica*
- 8. Jones, T. S., and Holland, E. C. (2010) Molecular Pathogenesis of Malignant Glial Tumors. *Toxicol Pathol* **000**, 1-9

- 9. Karpel-Massler, G., Schmidt, U., Unterberg, A., and Halatsch, M. E. (2009) Therapeutic inhibition of the epidermal growth factor receptor in high-grade gliomas: where do we stand? *Mol Cancer Res* **7**, 1000-1012
- Ohgaki, H., Dessen, P., Jourde, B., Horstmann, S., Nishikawa, T., Di Patre, P. L., Burkhard, C., Schuler, D., Probst-Hensch, N. M., Maiorka, P. C., Baeza, N., Pisani, P., Yonekawa, Y., Yasargil, M. G., Lutolf, U. M., and Kleihues, P. (2004) Genetic pathways to glioblastoma: a population-based study. *Cancer research* 64, 6892-6899
- 11. Silvani, A., Gaviani, P., Lamperti, E. A., Eoli, M., Falcone, C., Dimeco, F., Milanesi, I. M., Erbetta, A., Boiardi, A., Fariselli, L., and Salmaggi, A. (2009) Cisplatinum and BCNU chemotherapy in primary glioblastoma patients. *Journal of neuro-oncology* **94**, 57-62
- 12. Walker, M. D. (1978) The contemporary role of chemotherapy in the treatment of malignant brain tumor. *Clin Neurosurg* **25**, 388-396
- Walker, M. D., Green, S. B., Byar, D. P., Alexander, E., Jr., Batzdorf, U., Brooks, W. H., Hunt, W. E., MacCarty, C. S., Mahaley, M. S., Jr., Mealey, J., Jr., Owens, G., Ransohoff, J., 2nd, Robertson, J. T., Shapiro, W. R., Smith, K. R., Jr., Wilson, C. B., and Strike, T. A. (1980) Randomized comparisons of radiotherapy and nitrosoureas for the treatment of malignant glioma after surgery. *N Engl J Med* **303**, 1323-1329
- 14. La Rocca, R. V., and Mehdorn, H. M. (2009) Localized BCNU chemotherapy and the multimodal management of malignant glioma. *Curr Med Res Opin* **25**, 149-160
- 15. Grossman, S. A., and Batara, J. F. (2004) Current management of glioblastoma multiforme. *Seminars in oncology* **31**, 635-644
- 16. Agarwala, S. S., and Kirkwood, J. M. (2000) Temozolomide, a novel alkylating agent with activity in the central nervous system, may improve the treatment of advanced metastatic melanoma. *The oncologist* **5**, 144-151
- 17. Miller, J. L. (1999) Oral temozolomide approved for refractory brain tumor. American journal of health-system pharmacy : AJHP : official journal of the American Society of Health-System Pharmacists **56**, 1910

- 18. Sansom, C. (1999) Temozolomide presents breakthrough in glioblastoma multiforme treatment. *Pharmaceutical science & technology today* **4**, 131-133
- 19. Viel, T., Monfared, P., Schelhaas, S., Fricke, I. B., Kuhlmann, M. T., Fraefel, C., and Jacobs, A. H. (2013) Optimizing glioblastoma temozolomide chemotherapy employing lentiviral-based anti-MGMT shRNA technology. *Molecular therapy : the journal of the American Society of Gene Therapy* **21**, 570-579
- 20. Stupp, R., Mason, W. P., van den Bent, M. J., Weller, M., Fisher, B., Taphoorn, M. J., Belanger, K., Brandes, A. A., Marosi, C., Bogdahn, U., Curschmann, J., Janzer, R. C., Ludwin, S. K., Gorlia, T., Allgeier, A., Lacombe, D., Cairncross, J. G., Eisenhauer, E., Mirimanoff, R. O., European Organisation for, R., Treatment of Cancer Brain, T., Radiotherapy, G., and National Cancer Institute of Canada Clinical Trials, G. (2005) Radiotherapy plus concomitant and adjuvant temozolomide for glioblastoma. *The New England journal of medicine* **352**, 987-996
- 21. Stupp, R., van den Bent, M. J., and Hegi, M. E. (2005) Optimal role of temozolomide in the treatment of malignant gliomas. *Current neurology and neuroscience reports* **5**, 198-206
- Hegi, M. E., Diserens, A. C., Gorlia, T., Hamou, M. F., de Tribolet, N., Weller, M., Kros, J. M., Hainfellner, J. A., Mason, W., Mariani, L., Bromberg, J. E., Hau, P., Mirimanoff, R. O., Cairncross, J. G., Janzer, R. C., and Stupp, R. (2005) MGMT gene silencing and benefit from temozolomide in glioblastoma. *The New England journal of medicine* **352**, 997-1003
- 23. Rivera, A. L., Pelloski, C. E., Gilbert, M. R., Colman, H., De La Cruz, C., Sulman, E. P., Bekele, B. N., and Aldape, K. D. (2010) MGMT promoter methylation is predictive of response to radiotherapy and prognostic in the absence of adjuvant alkylating chemotherapy for glioblastoma. *Neurooncology* **12**, 116-121
- 24. Slamon, D., and Pegram, M. (2001) Rationale for trastuzumab (Herceptin) in adjuvant breast cancer trials. *Semin Oncol* **28**, 13-19
- 25. Norden, A. D., Drappatz, J., and Wen, P. Y. (2008) Antiangiogenic therapy in malignant gliomas. *Curr Opin Oncol* **20**, 652-661

- 26. Norden, A. D., Drappatz, J., and Wen, P. Y. (2008) Novel anti-angiogenic therapies for malignant gliomas. *Lancet Neurol* **7**, 1152-1160
- 27. Mischel, P. S., and Cloughesy, T. F. (2003) Targeted molecular therapy of GBM. *Brain Pathol* **13**, 52-61
- Wikstrand, C. J., Hale, L. P., Batra, S. K., Hill, M. L., Humphrey, P. A., Kurpad, S. N., McLendon, R. E., Moscatello, D., Pegram, C. N., Reist, C. J., Traweek, S. T., Wong, A. J., Zalutsky, M. R., and Bigner, D. D. (1995) Monoclonal antibodies against EGFRvIII are tumor specific and react with breast and lung carcinomas and malignant gliomas. *Cancer Res* 55, 3140-3148
- 29. Lv, S., Teugels, E., Sadones, J., De Brakeleer, S., Duerinck, J., Du Four, S., Michotte, A., De Greve, J., and Neyns, B. (2012) Correlation of EGFR, IDH1 and PTEN status with the outcome of patients with recurrent glioblastoma treated in a phase II clinical trial with the EGFR-blocking monoclonal antibody cetuximab. *International journal of oncology* **41**, 1029-1035
- Hegi, M. E., Diserens, A. C., Bady, P., Kamoshima, Y., Kouwenhoven, M. C., Delorenzi, M., Lambiv, W. L., Hamou, M. F., Matter, M. S., Koch, A., Heppner, F. L., Yonekawa, Y., Merlo, A., Frei, K., Mariani, L., and Hofer, S. (2011) Pathway analysis of glioblastoma tissue after preoperative treatment with the EGFR tyrosine kinase inhibitor gefitinib--a phase II trial. *Molecular cancer therapeutics* **10**, 1102-1112
- Kesavabhotla, K., Schlaff, C. D., Shin, B., Mubita, L., Kaplan, R., Tsiouris, A. J., Pannullo, S. C., Christos, P., Lavi, E., Scheff, R., and Boockvar, J. A. (2012) Phase I/II study of oral erlotinib for treatment of relapsed/refractory glioblastoma multiforme and anaplastic astrocytoma. *Journal of experimental therapeutics & oncology* **10**, 71-81
- Schuster, J., Lai, R. K., Recht, L. D., Reardon, D. A., Paleologos, N. A., Groves, M. D., Mrugala, M. M., Jensen, R., Baehring, J. M., Sloan, A., Archer, G. E., Bigner, D. D., Cruickshank, S., Green, J. A., Keler, T., Davis, T. A., Heimberger, A. B., and Sampson, J. H. (2015) A phase II, multicenter trial of rindopepimut (CDX-110) in newly diagnosed glioblastoma: the ACT III study. *Neuro-oncology* **17**, 854-861
- Gatson, N. T., Weathers, S. P., and de Groot, J. F. (2016) ReACT Phase II trial: a critical evaluation of the use of rindopepimut plus bevacizumab to treat EGFRvIII-positive recurrent glioblastoma. *CNS oncology* 5, 11-26

- 34. Hynes, N. E., and Lane, H. A. (2005) ERBB receptors and cancer: the complexity of targeted inhibitors. *Nature reviews. Cancer* **5**, 341-354
- 35. Hasselbalch, B., Lassen, U., Poulsen, H. S., and Stockhausen, M. T. (2010) Cetuximab insufficiently inhibits glioma cell growth due to persistent EGFR downstream signaling. *Cancer Invest* **28**, 775-787
- 36. Wells, A. (1999) EGF receptor. Int J Biochem Cell Biol 31, 637-643
- 37. Weinberg, R. A. (2007) *The Biology of Cancer*, Garland Science, New York
- 38. Tortora, G., Bianco, R., Daniele, G., Ciardiello, F., McCubrey, J. A., Ricciardi, M. R., Ciuffreda, L., Cognetti, F., Tafuri, A., and Milella, M. (2007) Overcoming resistance to molecularly targeted anticancer therapies: Rational drug combinations based on EGFR and MAPK inhibition for solid tumours and haematologic malignancies. *Drug resistance updates : reviews and commentaries in antimicrobial and anticancer chemotherapy* **10**, 81-100
- Brennan, C. W., Verhaak, R. G., McKenna, A., Campos, B., Noushmehr, H., Salama, S. R., Zheng, S., Chakravarty, D., Sanborn, J. Z., Berman, S. H., Beroukhim, R., Bernard, B., Wu, C. J., Genovese, G., Shmulevich, I., Barnholtz-Sloan, J., Zou, L., Vegesna, R., Shukla, S. A., Ciriello, G., Yung, W. K., Zhang, W., Sougnez, C., Mikkelsen, T., Aldape, K., Bigner, D. D., Van Meir, E. G., Prados, M., Sloan, A., Black, K. L., Eschbacher, J., Finocchiaro, G., Friedman, W., Andrews, D. W., Guha, A., Iacocca, M., O'Neill, B. P., Foltz, G., Myers, J., Weisenberger, D. J., Penny, R., Kucherlapati, R., Perou, C. M., Hayes, D. N., Gibbs, R., Marra, M., Mills, G. B., Lander, E., Spellman, P., Wilson, R., Sander, C., Weinstein, J., Meyerson, M., Gabriel, S., Laird, P. W., Haussler, D., Getz, G., Chin, L., and Network, T. R. (2013) The somatic genomic landscape of glioblastoma. *Cell* 155, 462-477
- 40. Furnari, F. B., Cloughesy, T. F., Cavenee, W. K., and Mischel, P. S. (2015) Heterogeneity of epidermal growth factor receptor signalling networks in glioblastoma. *Nature reviews. Cancer* **15**, 302-310
- Francis, J. M., Zhang, C. Z., Maire, C. L., Jung, J., Manzo, V. E., Adalsteinsson, V. A., Homer, H., Haidar, S., Blumenstiel, B., Pedamallu, C. S., Ligon, A. H., Love, J. C., Meyerson, M., and Ligon, K. L. (2014) EGFR variant heterogeneity in glioblastoma resolved through single-nucleus sequencing. *Cancer discovery* 4, 956-971

- 42. Huang, H. S., Nagane, M., Klingbeil, C. K., Lin, H., Nishikawa, R., Ji, X. D., Huang, C. M., Gill, G. N., Wiley, H. S., and Cavenee, W. K. (1997) The enhanced tumorigenic activity of a mutant epidermal growth factor receptor common in human cancers is mediated by threshold levels of constitutive tyrosine phosphorylation and unattenuated signaling. *The Journal of biological chemistry* **272**, 2927-2935
- 43. Gan, H. K., Kaye, A. H., and Luwor, R. B. (2009) The EGFRvIII variant in glioblastoma multiforme. *J Clin Neurosci* **16**, 748-754
- 44. Mukasa, A., Wykosky, J., Ligon, K. L., Chin, L., Cavenee, W. K., and Furnari, F. (2010) Mutant EGFR is required for maintenance of glioma growth in vivo, and its ablation leads to escape from receptor dependence. *Proceedings of the National Academy of Sciences of the United States of America* **107**, 2616-2621
- Shinojima, N., Tada, K., Shiraishi, S., Kamiryo, T., Kochi, M., Nakamura, H., Makino, K., Saya, H., Hirano, H., Kuratsu, J., Oka, K., Ishimaru, Y., and Ushio, Y. (2003) Prognostic value of epidermal growth factor receptor in patients with glioblastoma multiforme. *Cancer research* 63, 6962-6970
- Nishikawa, R., Ji, X. D., Harmon, R. C., Lazar, C. S., Gill, G. N., Cavenee, W. K., and Huang, H. J. (1994) A mutant epidermal growth factor receptor common in human glioma confers enhanced tumorigenicity. *Proceedings of the National Academy of Sciences of the United States of America* 91, 7727-7731
- 47. Schmidt, M. H., Furnari, F. B., Cavenee, W. K., and Bogler, O. (2003) Epidermal growth factor receptor signaling intensity determines intracellular protein interactions, ubiquitination, and internalization. *Proceedings of the National Academy of Sciences of the United States of America* **100**, 6505-6510
- 48. Nagane, M., Levitzki, A., Gazit, A., Cavenee, W. K., and Huang, H. J. (1998) Drug resistance of human glioblastoma cells conferred by a tumor-specific mutant epidermal growth factor receptor through modulation of Bcl-XL and caspase-3-like proteases. *Proceedings of the National Academy of Sciences of the United States of America* **95**, 5724-5729

- 50. Inda, M. M., Bonavia, R., Mukasa, A., Narita, Y., Sah, D. W., Vandenberg, S., Brennan, C., Johns, T. G., Bachoo, R., Hadwiger, P., Tan, P., Depinho, R. A., Cavenee, W., and Furnari, F. (2010) Tumor heterogeneity is an active process maintained by a mutant EGFR-induced cytokine circuit in glioblastoma. *Genes* & development 24, 1731-1745
- 51. Tang, C. K., Gong, X. Q., Moscatello, D. K., Wong, A. J., and Lippman, M. E. (2000) Epidermal growth factor receptor vIII enhances tumorigenicity in human breast cancer. *Cancer research* **60**, 3081-3087
- 52. Nagane, M., Coufal, F., Lin, H., Bogler, O., Cavenee, W. K., and Huang, H. J. (1996) A common mutant epidermal growth factor receptor confers enhanced tumorigenicity on human glioblastoma cells by increasing proliferation and reducing apoptosis. *Cancer research* **56**, 5079-5086
- 53. Bonavia, R., Inda, M. M., Vandenberg, S., Cheng, S. Y., Nagane, M., Hadwiger, P., Tan, P., Sah, D. W., Cavenee, W. K., and Furnari, F. B. (2012) EGFRvIII promotes glioma angiogenesis and growth through the NF-kappaB, interleukin-8 pathway. *Oncogene* **31**, 4054-4066
- 54. Feng, H., Hu, B., Vuori, K., Sarkaria, J. N., Furnari, F. B., Cavenee, W. K., and Cheng, S. Y. (2014) EGFRvIII stimulates glioma growth and invasion through PKA-dependent serine phosphorylation of Dock180. *Oncogene* **33**, 2504-2512
- 55. Bonavia, R., Inda, M. M., Cavenee, W. K., and Furnari, F. B. (2011) Heterogeneity maintenance in glioblastoma: a social network. *Cancer research* **71**, 4055-4060
- Nathanson, D. A., Gini, B., Mottahedeh, J., Visnyei, K., Koga, T., Gomez, G., Eskin, A., Hwang, K., Wang, J., Masui, K., Paucar, A., Yang, H., Ohashi, M., Zhu, S., Wykosky, J., Reed, R., Nelson, S. F., Cloughesy, T. F., James, C. D., Rao, P. N., Kornblum, H. I., Heath, J. R., Cavenee, W. K., Furnari, F. B., and Mischel, P. S. (2014) Targeted therapy resistance mediated by dynamic regulation of extrachromosomal mutant EGFR DNA. *Science* 343, 72-76
- 57. Mellinghoff, I. K., Wang, M. Y., Vivanco, I., Haas-Kogan, D. A., Zhu, S., Dia, E. Q., Lu, K. V., Yoshimoto, K., Huang, J. H., Chute, D. J., Riggs, B. L., Horvath,

S., Liau, L. M., Cavenee, W. K., Rao, P. N., Beroukhim, R., Peck, T. C., Lee, J. C., Sellers, W. R., Stokoe, D., Prados, M., Cloughesy, T. F., Sawyers, C. L., and Mischel, P. S. (2005) Molecular determinants of the response of glioblastomas to EGFR kinase inhibitors. *The New England journal of medicine* **353**, 2012-2024

- Vivanco, I., Robins, H. I., Rohle, D., Campos, C., Grommes, C., Nghiemphu, P. L., Kubek, S., Oldrini, B., Chheda, M. G., Yannuzzi, N., Tao, H., Zhu, S., Iwanami, A., Kuga, D., Dang, J., Pedraza, A., Brennan, C. W., Heguy, A., Liau, L. M., Lieberman, F., Yung, W. K., Gilbert, M. R., Reardon, D. A., Drappatz, J., Wen, P. Y., Lamborn, K. R., Chang, S. M., Prados, M. D., Fine, H. A., Horvath, S., Wu, N., Lassman, A. B., DeAngelis, L. M., Yong, W. H., Kuhn, J. G., Mischel, P. S., Mehta, M. P., Cloughesy, T. F., and Mellinghoff, I. K. (2012) Differential sensitivity of glioma- versus lung cancer-specific EGFR mutations to EGFR kinase inhibitors. *Cancer discovery* 2, 458-471
- 59. Johnson, L. A., and Sampson, J. H. (2010) Immunotherapy approaches for malignant glioma from 2007 to 2009. *Current neurology and neuroscience reports* **10**, 259-266
- 60. Weiner, L. M., Surana, R., and Wang, S. (2010) Monoclonal antibodies: versatile platforms for cancer immunotherapy. *Nat Rev Immunol* **10**, 317-327
- 61. Ehrlich, P. (1906) *Collected Studies on Immunity*, J. Wiley & Sons, New York
- 62. Rivera, F., Vega-Villegas, M. E., and Lopez-Brea, M. F. (2008) Cetuximab, its clinical use and future perspectives. *Anticancer Drugs* **19**, 99-113
- 63. Eller, J. L., Longo, S. L., Hicklin, D. J., and Canute, G. W. (2002) Activity of anti-epidermal growth factor receptor monoclonal antibody C225 against glioblastoma multiforme. *Neurosurgery* **51**, 1005-1013; discussion 1013-1004
- Eller, J. L., Longo, S. L., Kyle, M. M., Bassano, D., Hicklin, D. J., and Canute, G. W. (2005) Anti-epidermal growth factor receptor monoclonal antibody cetuximab augments radiation effects in glioblastoma multiforme in vitro and in vivo. *Neurosurgery* 56, 155-162; discussion 162
- 65. Combs, S. E., Heeger, S., Haselmann, R., Edler, L., Debus, J., and Schulz-Ertner, D. (2006) Treatment of primary glioblastoma multiforme with cetuximab, radiotherapy and temozolomide (GERT)--phase I/II trial: study protocol. *BMC Cancer* **6**, 133

- 66. Cho, J., Pastorino, S., Zeng, Q., Xu, X., Johnson, W., Vandenberg, S., Verhaak, R., Cherniack, A. D., Watanabe, H., Dutt, A., Kwon, J., Chao, Y. S., Onofrio, R. C., Chiang, D., Yuza, Y., Kesari, S., and Meyerson, M. (2011) Glioblastoma-Derived Epidermal Growth Factor Receptor Carboxyl-Terminal Deletion Mutants Are Transforming and Are Sensitive to EGFR-Directed Therapies. *Cancer Res* **71**, 1-10
- 67. Padfield, E., Ellis, H. P., and Kurian, K. M. (2015) Current Therapeutic Advances Targeting EGFR and EGFRvIII in Glioblastoma. *Frontiers in oncology* **5**, 5
- 68. Emrich, J. G., Brady, L. W., Quang, T. S., Class, R., Miyamoto, C., Black, P., and Rodeck, U. (2002) Radioiodinated (I-125) monoclonal antibody 425 in the treatment of high grade glioma patients: ten-year synopsis of a novel treatment. *Am J Clin Oncol* **25**, 541-546
- 69. Quang, T. S., and Brady, L. W. (2004) Radioimmunotherapy as a novel treatment regimen: 125I-labeled monoclonal antibody 425 in the treatment of high-grade brain gliomas. *Int J Radiat Oncol Biol Phys* **58**, 972-975
- 70. Li, L., Quang, T. S., Gracely, E. J., Kim, J. H., Emrich, J. G., Yaeger, T. E., Jenrette, J. M., Cohen, S. C., Black, P., and Brady, L. W. (2010) A Phase II study of anti-epidermal growth factor receptor radioimmunotherapy in the treatment of glioblastoma multiforme. *J Neurosurg* **113**, 192-198
- 71. Liu, T. F., Hall, P. D., Cohen, K. A., Willingham, M. C., Cai, J., Thorburn, A., and Frankel, A. E. (2005) Interstitial diphtheria toxin-epidermal growth factor fusion protein therapy produces regressions of subcutaneous human glioblastoma multiforme tumors in athymic nude mice. *Clin Cancer Res* **11**, 329-334
- 72. Mishima, K., Johns, T. G., Luwor, R. B., Scott, A. M., Stockert, E., Jungbluth, A. A., Ji, X. D., Suvarna, P., Voland, J. R., Old, L. J., Huang, H. J., and Cavenee, W. K. (2001) Growth suppression of intracranial xenografted glioblastomas overexpressing mutant epidermal growth factor receptors by systemic administration of monoclonal antibody (mAb) 806, a novel monoclonal antibody directed to the receptor. *Cancer Res* **61**, 5349-5354
- 73. Gan, H. K., Lappas, M., Cao, D. X., Cvrljevdic, A., Scott, A. M., and Johns, T. G. (2009) Targeting a unique EGFR epitope with monoclonal antibody 806

activates NF-kappaB and initiates tumour vascular normalization. *J Cell Mol Med* **13**, 3993-4001

- 74. Ng, Y. Z., Kannan, S., Lane, D. P., Fuentes, G., and Verma, C. S. (2015) mAb806 binding to epidermal growth factor receptor: a computational study. *Proteins* **83**, 153-168
- 75. Jungbluth, A. A., Stockert, E., Huang, H. J., Collins, V. P., Coplan, K., Iversen, K., Kolb, D., Johns, T. J., Scott, A. M., Gullick, W. J., Ritter, G., Cohen, L., Scanlan, M. J., Cavenee, W. K., and Old, L. J. (2003) A monoclonal antibody recognizing human cancers with amplification/overexpression of the human epidermal growth factor receptor. *Proceedings of the National Academy of Sciences of the United States of America* **100**, 639-644
- 76. Luwor, R. B., Johns, T. G., Murone, C., Huang, H. J., Cavenee, W. K., Ritter, G., Old, L. J., Burgess, A. W., and Scott, A. M. (2001) Monoclonal antibody 806 inhibits the growth of tumor xenografts expressing either the de2-7 or amplified epidermal growth factor receptor (EGFR) but not wild-type EGFR. *Cancer Res* **61**, 5355-5361
- 77. Scott, A. M., Lee, F. T., Tebbutt, N., Herbertson, R., Gill, S. S., Liu, Z., Skrinos, E., Murone, C., Saunder, T. H., Chappell, B., Papenfuss, A. T., Poon, A. M., Hopkins, W., Smyth, F. E., MacGregor, D., Cher, L. M., Jungbluth, A. A., Ritter, G., Brechbiel, M. W., Murphy, R., Burgess, A. W., Hoffman, E. W., Johns, T. G., and Old, L. J. (2007) A phase I clinical trial with monoclonal antibody ch806 targeting transitional state and mutant epidermal growth factor receptors. *Proc Natl Acad Sci U S A* **104**, 4071-4076
- Reilly, E. B., Phillips, A. C., Buchanan, F. G., Kingsbury, G., Zhang, Y., Meulbroek, J. A., Cole, T. B., DeVries, P. J., Falls, H. D., Beam, C., Gu, J., Digiammarino, E. L., Palma, J. P., Donawho, C. K., Goodwin, N. C., and Scott, A. M. (2015) Characterization of ABT-806, a Humanized Tumor-Specific Anti-EGFR Monoclonal Antibody. *Molecular cancer therapeutics* 14, 1141-1151
- Cleary, J. M., Reardon, D. A., Azad, N., Gandhi, L., Shapiro, G. I., Chaves, J., Pedersen, M., Ansell, P., Ames, W., Xiong, H., Munasinghe, W., Dudley, M., Reilly, E. B., Holen, K., and Humerickhouse, R. (2015) A phase 1 study of ABT-806 in subjects with advanced solid tumors. *Investigational new drugs* 33, 671-678

- Phillips, A. C., Boghaert, E. R., Vaidya, K. S., Mitten, M. J., Norvell, S., Falls, H. D., DeVries, P. J., Cheng, D., Meulbroek, J. A., Buchanan, F. G., McKay, L. M., Goodwin, N. C., and Reilly, E. B. (2016) ABT-414, an Antibody-Drug Conjugate Targeting a Tumor-Selective EGFR Epitope. *Molecular cancer therapeutics* 15, 661-669
- 81. Moscatello, D. K., Ramirez, G., and Wong, A. J. (1997) A naturally occurring mutant human epidermal growth factor receptor as a target for peptide vaccine immunotherapy of tumors. *Cancer research* **57**, 1419-1424
- 82. Kanaly, C. W., Ding, D., Heimberger, A. B., and Sampson, J. H. (2010) Clinical applications of a peptide-based vaccine for glioblastoma. *Neurosurg Clin N Am* **21**, 95-109
- 83. Yamanaka, R. (2009) Dendritic-cell- and peptide-based vaccination strategies for glioma. *Neurosurg Rev* **32**, 265-273; discussion 273
- 84. Heimberger, A. B., and Sampson, J. H. (2009) The PEPvIII-KLH (CDX-110) vaccine in glioblastoma multiforme patients. *Expert Opin Biol Ther* **9**, 1087-1098
- Sampson, J. H., Archer, G. E., Mitchell, D. A., Heimberger, A. B., Herndon, J. E., 2nd, Lally-Goss, D., McGehee-Norman, S., Paolino, A., Reardon, D. A., Friedman, A. H., Friedman, H. S., and Bigner, D. D. (2009) An epidermal growth factor receptor variant III-targeted vaccine is safe and immunogenic in patients with glioblastoma multiforme. *Molecular cancer therapeutics* 8, 2773-2779
- 86. Kohno, M., Horibe, T., Haramoto, M., Yano, Y., Ohara, K., Nakajima, O., Matsuzaki, K., and Kawakami, K. (2011) A novel hybrid peptide targeting EGFR-expressing cancers. *European journal of cancer* **47**, 773-783
- 87. Kikuchi, O., Ohashi, S., Horibe, T., Kohno, M., Nakai, Y., Miyamoto, S., Chiba, T., Muto, M., and Kawakami, K. (2016) Novel EGFR-targeted strategy with hybrid peptide against oesophageal squamous cell carcinoma. *Scientific reports* **6**, 22452
- 88. Tada, N., Horibe, T., Haramoto, M., Ohara, K., Kohno, M., and Kawakami, K. (2011) A single replacement of histidine to arginine in EGFR-lytic hybrid peptide demonstrates the improved anticancer activity. *Biochem Biophys Res Commun* **407**, 383-388

- Sampson, J. H., Heimberger, A. B., Archer, G. E., Aldape, K. D., Friedman, A. H., Friedman, H. S., Gilbert, M. R., Herndon, J. E., 2nd, McLendon, R. E., Mitchell, D. A., Reardon, D. A., Sawaya, R., Schmittling, R. J., Shi, W., Vredenburgh, J. J., and Bigner, D. D. (2010) Immunologic Escape After Prolonged Progression-Free Survival With Epidermal Growth Factor ReceptorVariant III Peptide Vaccination in Patients With NewlyDiagnosed Glioblastoma. *Journal of clinical oncology : official journal of the American Society of Clinical Oncology* 28, 4722-4729
- 90. Swartz, A. M., Li, Q. J., and Sampson, J. H. (2014) Rindopepimut: a promising immunotherapeutic for the treatment of glioblastoma multiforme. *Immunotherapy* **6**, 679-690
- 91. Castanotto, D., and Rossi, J. J. (2009) The promises and pitfalls of RNAinterference-based therapeutics. *Nature* **457**, 426-433
- Sowery, R. D., Hadaschik, B. A., So, A. I., Zoubeidi, A., Fazli, L., Hurtado-Coll, A., and Gleave, M. E. (2008) Clusterin knockdown using the antisense oligonucleotide OGX-011 re-sensitizes docetaxel-refractory prostate cancer PC-3 cells to chemotherapy. *BJU Int* **102**, 389-397
- 93. Zielinski, R., and Chi, K. N. (2012) Custirsen (OGX-011): a second-generation antisense inhibitor of clusterin in development for the treatment of prostate cancer. *Future oncology* **8**, 1239-1251
- 94. Laskin, J. J., Nicholas, G., Lee, C., Gitlitz, B., Vincent, M., Cormier, Y., Stephenson, J., Ung, Y., Sanborn, R., Pressnail, B., Nugent, F., Nemunaitis, J., Gleave, M. E., Murray, N., and Hao, D. (2012) Phase I/II trial of custirsen (OGX-011), an inhibitor of clusterin, in combination with a gemcitabine and platinum regimen in patients with previously untreated advanced non-small cell lung cancer. *Journal of thoracic oncology : official publication of the International Association for the Study of Lung Cancer* **7**, 579-586
- 95. Shir, A., and Levitzki, A. (2002) Inhibition of glioma growth by tumor-specific activation of double-stranded RNA-dependent protein kinase PKR. *Nat Biotechnol* **20**, 895-900
- 96. Kang, C. S., Zhang, Z. Y., Jia, Z. F., Wang, G. X., Qiu, M. Z., Zhou, H. X., Yu, S. Z., Chang, J., Jiang, H., and Pu, P. Y. (2006) Suppression of EGFR expression by antisense or small interference RNA inhibits U251 glioma cell growth in vitro and in vivo. *Cancer Gene Ther* **13**, 530-538

- Halatsch, M. E., Schmidt, U., Botefur, I. C., Holland, J. F., and Ohnuma, T. (2000) Marked inhibition of glioblastoma target cell tumorigenicity in vitro by retrovirus-mediated transfer of a hairpin ribozyme against deletion-mutant epidermal growth factor receptor messenger RNA. *Journal of neurosurgery* 92, 297-305
- 99. Halatsch, M. E., Gehrke, E. E., Vougioukas, V. I., Botefur, I. C., F, A. B., Efferth, T., Gebhart, E., Domhof, S., Schmidt, U., and Buchfelder, M. (2004) Inverse correlation of epidermal growth factor receptor messenger RNA induction and suppression of anchorage-independent growth by OSI-774, an epidermal growth factor receptor tyrosine kinase inhibitor, in glioblastoma multiforme cell lines. *Journal of neurosurgery* **100**, 523-533
- 100. Lal, A., Glazer, C. A., Martinson, H. M., Friedman, H. S., Archer, G. E., Sampson, J. H., and Riggins, G. J. (2002) Mutant epidermal growth factor receptor up-regulates molecular effectors of tumor invasion. *Cancer Res* **62**, 3335-3339
- 101. Moyer, J. D., Barbacci, E. G., Iwata, K. K., Arnold, L., Boman, B., Cunningham, A., DiOrio, C., Doty, J., Morin, M. J., Moyer, M. P., Neveu, M., Pollack, V. A., Pustilnik, L. R., Reynolds, M. M., Sloan, D., Theleman, A., and Miller, P. (1997) Induction of apoptosis and cell cycle arrest by CP-358,774, an inhibitor of epidermal growth factor receptor tyrosine kinase. *Cancer research* 57, 4838-4848
- Quatrale, A. E., Porcelli, L., Silvestris, N., Colucci, G., Angelo, A., and Azzariti, A. (2011) EGFR tyrosine kinases inhibitors in cancer treatment: in vitro and in vivo evidence. *Front Biosci* 16, 1962-1972
- 103. Raizer, J. J., Abrey, L. E., Lassman, A. B., Chang, S. M., Lamborn, K. R., Kuhn, J. G., Yung, W. K., Gilbert, M. R., Aldape, K. A., Wen, P. Y., Fine, H. A., Mehta, M., Deangelis, L. M., Lieberman, F., Cloughesy, T. F., Robins, H. I., Dancey, J., and Prados, M. D. (2010) A phase II trial of erlotinib in patients with recurrent malignant gliomas and nonprogressive glioblastoma multiforme postradiation therapy. *Neuro-oncology* **12**, 95-103

- 104. Knight, L. A., Di Nicolantonio, F., Whitehouse, P., Mercer, S., Sharma, S., Glaysher, S., Johnson, P., and Cree, I. A. (2004) The in vitro effect of gefitinib ('Iressa') alone and in combination with cytotoxic chemotherapy on human solid tumours. *BMC cancer* **4**, 83
- 105. Rich, J. N., Rasheed, B. K., and Yan, H. (2004) EGFR mutations and sensitivity to gefitinib. *N Engl J Med* **351**, 1260-1261; author reply 1260-1261
- 106. Geyer, C. E., Forster, J., Lindquist, D., Chan, S., Romieu, C. G., Pienkowski, T., Jagiello-Gruszfeld, A., Crown, J., Chan, A., Kaufman, B., Skarlos, D., Campone, M., Davidson, N., Berger, M., Oliva, C., Rubin, S. D., Stein, S., and Cameron, D. (2006) Lapatinib plus capecitabine for HER2-positive advanced breast cancer. *The New England journal of medicine* **355**, 2733-2743
- Thiessen, B., Stewart, C., Tsao, M., Kamel-Reid, S., Schaiquevich, P., Mason, W., Easaw, J., Belanger, K., Forsyth, P., McIntosh, L., and Eisenhauer, E. (2010) A phase I/II trial of GW572016 (lapatinib) in recurrent glioblastoma multiforme: clinical outcomes, pharmacokinetics and molecular correlation. *Cancer chemotherapy and pharmacology* 65, 353-361
- 108. Jones, T. S., and Holland, E. C. (2011) Molecular pathogenesis of malignant glial tumors. *Toxicol Pathol* **39**, 158-166

Chapter 2

Overview of Established Mechanisms of Resistance to EGFR inhibition

2.1 Introduction

Despite advances in our current knowledge of glioma biology and genetics, this disease remains largely incurable. For the reasons discussed in the previous chapter, many of the EGFR-targeting agents hold great potential, as their specific nature is a majorly desirable therapeutic component. However, the reality is that successful treatment of malignant gliomas with these agents alone has proven to be unachievable (1,2). Mechanisms causing resistance to EGFR tyrosine kinase inhibitors are well established and have been implicated in a number of solid tumors. Some of the documented ways include the acquisition of secondary EGFR point mutations, co-activation and/or amplification of other RTKs, heterogeneity, and upregulation of drug efflux pumps (2). Many of these have been established in nonsmall cell lung carcinoma (NSCLC) and breast cancer; however, a comprehensive understanding of the mechanisms of therapeutic resistance that are specific to gliomas is undefined (2,3).

Malignant brain tumors are typified by resistance to apoptosis and diffuse invasion into the normal brain parenchyma (4). Once tumor cells disseminate into the normal brain regions, they are protected by an intact BBB, which can significantly limit the efficacy of therapeutic agents that cannot traverse this barrier (5). Furthermore, despite being robustly angiogenic, the tortuous vasculature in malignant brain tumors critically limits drug penetration (5). It is clear that GBMs are recalcitrant by nature with demonstrated ability to escape the need for receptor function when challenged with receptor-targeted therapeutics (6). In this chapter, we will explore some of the established mechanisms of resistance to EGFR inhibition in GBM and other solid tumors that are essential to this "escaper" phenotype.

2.2 Cross-Resistance

A primary mechanism by which some EGFR TKIs (i.e., gefitinib, erlotinib) exert their antitumor effects is through induction of apoptosis (6). During invasion, brain tumor cells often arrest in mitosis, rendering them refractory to radiotherapy and chemotherapy, where the mechanism of antitumor activity is to trigger DNA damage-induced apoptosis in proliferating cells (5). In essence, brain tumor cells that have developed a way to evade chemotherapy and radiotherapy-induced cell death may also be intractable to these TKIs or vice versa, as a result of cross-resistance. Cross-resistance is defined as tolerance to a normally toxic agent, resulting from exposure to a similarly acting substance.

Oncogenic EGFRvIII has been demonstrated to confer resistance to both radiotherapy and chemotherapy by promoting the induction of the anti-apoptotic protein, B-cell lymphoma-extra large (Bcl-xL) (7). Additionally, in a U373MG glioma model harboring tetracycline-regulatable EGFRvIII, Mukasa *et al.* established that despite sustained repression of EGFRvIII, some GBM tumors eventually recur (8). Indeed, these recurrent tumors called "Escapers" are significantly less apoptotic and display restored Bcl-xL protein expression (8). Furthermore, they are resistant to EGFR TKI's even when EGFRvIII is re-expressed, reflecting a completely EGFRvIII-independent phenotype. This resistance manifests as failure to up-regulate the pro-apoptotic protein BIM (**Figure 2.1**), due to the urokinase plasminogen activator system (uPAS), a mechanism of resistance that we recently established in different glioma models (9).

In NSCLC, it's been shown that disruption of *BIM* function can also arise from a deletion polymorphism in intron 2 of the gene, which leads to preferential splicing of

exon 3 over exon 4 (10,11). The resulting gene product lacks the Bcl-2 homology domain 3 (BH3) (located in exon 4), which is indispensable for the pro-apoptotic function of *BIM*, and this deficiency is sufficient to confer inherent resistance to EGFR TKI's. Fortunately, this resistance can be circumvented with treatment cocktails, consisting of EGFR TKI's plus either a BH3 mimetic drug or a histone deacetylase (HDAC) inhibitor (11,12). However, in GBM, this polymorphism has not been detected.

These results suggest that EGFR-targeting monotherapies that solely function through the induction of pro-apoptotic signaling may not reach maximum therapeutic efficacy in GBM (8,13). GBMs that initially exhibit a predominant anti-apoptotic phenotype by way of EGFRvIII receptor function appear to regain this property even in the absence of this gene. These tumors robustly respond, ensuring that a compensatory pathway to dampen cell killing activation is already in place even before exposure to EGFR TKI's. In the case of acquired resistance, this property is likely afforded by adjuvant therapy following surgical resection. Thus, achieving durable drug efficacy with alternative agents may require mechanisms of antitumor activity that are entirely different from the components of the standard of care regimen.



Figure 2.1. Failure to up-regulate the pro-apoptotic protein BIM is an established mechanism of resistance in glioma. When EGFRvIII is inhibited, either genetically or pharmacologically, BIM up-regulation is not observed in "Escapers" at both the mRNA and protein levels. In contrast, the EGFRvIII-dependent populations significantly induce BIM upon EGFRvIII inhibition, suggestive of proapoptotic signaling and response. n.s-not significant; C-DMSO control ; G-gefitinib; E-

erlotinib

2.3 Glioma Stem Cells

Genetic pathways that are involved in the development of malignant gliomas are now reasonably well characterized; however, the cellular origins of these tumors are poorly understood (14-16). There is growing evidence that glioma stem cells are major contributors to resistance to standard treatments. Several studies have demonstrated that malignant brain tumors that are enriched with cancer stem cells (GSCs) are more resistant to radiotherapy and chemotherapy due to their ability to alter both the DNA damage checkpoint and DNA repair pathways (5,17).

Many conventional therapies specifically target rapidly proliferating cells, while sparing the quiescent, tumor cell compartment. This approach emerged from the notion that proliferation is the main problem to address for successful cancer treatment (18). Indeed, rapidly proliferating cells make up the bulk of the tumor; however, if the GSCs that are left behind can repopulate the tumor, then the question that inevitably arises is: Do the current anti-cancer therapies account for the tumor in its entirety?

Cancer stem cells are thought to be the source of tumor relapse due to their intrinsic and acquired properties, such as increased genetic stability, decreased oxidative stress, and enhanced ATP-binding cassette (ABC) transporter activity (19). In the latter case, activation of these transporters would lead to increased efflux of anti-EGFR agents from the tumor, resulting in decreased intracellular concentrations of drug. Indeed in NSCLC, binding of the EGFR TKI lapatanib to the substrate binding sites of ABC subfamily B, member 1 (ABCB1) and ABC subfamily G, member 2 (ABCG2), actually induces the overexpression of these ABC transporters (20).

The genetic stability of cancer stem cells would render them unperturbed by the pro-apoptotic signals generated by EGFR TKIs. The challenge therefore remains in designing novel therapies that specifically target the GSC population or GSCassociated factors to eradicate these sources of tumor maintenance. Delineating the properties that are most essential to their survival is required, but more importantly, understanding the biological differences between normal stem cells and GSCs is required to develop selective therapies that would spare normal brain cells.

In a study designed to test the efficacy of bortezomib, a protease inhibitor, in treating multiple low- and high-grade GSC cultures, it was revealed that bortezomib can reduce GSC populations by 80% with minimal effects on normal stem cell (NSC) populations (21). Additionally, erlotinib was shown to produce similar results as bortezomib, while TMZ and cisplatin were more toxic to NSCs and less effective against GSCs (21). Thus, it appears that combining newer, promising agents with TKIs, rather than with older chemotherapeutic agents, could result in more durable responses in GBM patients. Furthermore, this could potentially reduce the likelihood of cross-resistance, since the removal of older chemotherapeutics would also eliminate tumor pre-exposure to agents where induction of apoptosis is the predominant cytotoxic property.

The bone marrow tyrosine kinase gene in chromosome X protein (BMX) has also been implicated in maintaining self-renewal and tumorigenic potential of GSC populations by activating the signal transducer and activator of transcription 3 (STAT3) (22). BMX silencing leads to potent inhibition of STAT3 activation and inhibits growth of GSC-derived intracranial tumors (22). Additionally, GSCs can preferentially secrete factors such as periostin (POSTN) to promote the malignant growth of GBMs (23). Periostin functions as a chemoattractant to recruit monocyte derived macrophages from the peripheral blood into the brain via integrin alpha V Beta 3 ($\alpha_v\beta_3$) signaling (23). Once in the brain, they are transformed into the tumor-supportive M2 subtype of tumor associated macrophages (TAMs) in part by periostin local function (23). Silencing POSTN reduces the recruitment of the M2 subtype of TAMs, leading to inhibition of GBM tumor growth and massive apoptosis in vivo (23).

Interestingly, periostin has been implicated as a driver of emergence from tumor dormancy and activation of metastatic outgrowth in breast cancer (24). Brain tumor cells do not typically metastasize to distant organs; however, the dormancy period observed in breast cancer could be similar to the stasis period observed following EGFRvIII inhibition in our U373MG-tetO-EGFRvIII model (8). The "Escapers" that emerged in our model would then reflect a departure from tumor dormancy. Indeed, in an shRNA *in vivo* screening method to triage putative Escaperassociated genes identified by microarray analysis (8), we found that POSTN supports EGFRvIII-independent tumor maintenance (**Figures 2.2 and 2.3**). Thus, it appears that the expression of a GSC-associated factor confers resistance to EGFR inhibition, and identifying additional factors that may preserve this quiescent population to then ultimately afford escape, is clearly warranted to overcome the contribution of GSCs to therapeutic resistance.


Figure 2.2. In vivo shRNA screen to identify genes that are essential to EGFRvIII-independent tumor maintenance. An outline of the experimental procedure that was used, from the point of transducing the cells with shRNA lentiviral vectors to sample collection and sequencing. The shRNAs used were specific to the 19 "escaper"-associated genes already identified by microarray analysis in Mukasa *et al.*, and then 6 internal control genes to substantiate methodology. Hairpin representation values, identified by sequence barcodes, in the resulting *in vivo* tumor samples were subtracted from respective representation values in the pre-injection reference samples to conclude if a gene was essential or not. gDNA-genomic DNA



Figure 2.3. Periostin is an essential gene in EGFRvIII-independence tumor maintenance. Three independent hairpin libraries were created to carry out the experiment in triplicate. No significant difference was observed in the growth rate of Escaper 5 tumors resulting from cells transduced with the three libraries. Cells transduced with shRNAs specific to POSTN were significantly underrepresented in the resulting in vivo tumor samples. Relative abundance value reflects the average from the three experiments.

2.4 Tumor Microenvironment

Malignant glioma patients are known to be profoundly immunosuppressed. Thus, even if EGFR-specific monoclonal antibodies or peptide vaccines generate systemic immune responses, they could be negated in the tumor microenvironment by a variety of immunosuppressive growth factors and cytokines (25). In contrast to EGFR TKIs, the mechanisms of resistance to these therapies in any cancer are poorly defined (26). Cytokines, such as the transforming growth factor beta (TGF β) and interleukin-6 (IL-6), have been shown to promote tumor escape from immunosurveillance, and specifically high plasma levels of TGF β correlate with a negative prognosis in a variety of cancers (27).

TGFβ confers the transformation of vascular endothelial cells to the proangiogenic phenotype that is associated with GBM. Additionally, the production of TGFβ can lead to the accumulation of CD3⁺, CD4⁺, CD25⁺, FOXP3⁺ T_{Reg} cells. Regulatory T cells (T_{Regs}) are a specialized subpopulation of T cells that function to actively suppress the immune system in order to maintain system homeostasis and to prevent pathological self-reactivity. T_{Regs} are enriched in the peripheral blood of GBM and also in the tumor microenvironment, and though a prognostic role of T_{Regs} present in glioma has not been thoroughly evaluated, T_{Regs} are associated with poor clinical outcome in other systemic tumors (25). In a syngeneic murine model of glioma, depletion of T_{Regs} resulted in complete tumor rejection and enhanced survival (28). Thus, T_{Regs} can antagonize EGFR-specific immunotherapy agents by promoting immunosuppression.

Aside from its involvement in GSC maintenance via BMX, STAT3 also mediates TGF β production, which has been linked to potent inhibition of both innate

and acquired immune responses (25). Specifically, STAT3 induces immune tolerance via T_{Reg} activity and inhibiting STAT3 impedes T_{Reg} function (29). STAT3 is overexpressed in GBM, and its elevated activity correlates with decreased overall survival in patients (29). Based on these findings, inhibitors of Janus kinase (JAK)/STAT activation have been explored as a potential strategy to overcome T_{Reg} -mediated immunosuppression (25). Presumably, if STAT3 coordinates the activity of T_{Regs} with other cell populations in the tumor microenvironment (e.g. tumor stem cells), then such an approach is expected to be efficacious.

STAT3 is a protumorigenic factor in EGFRvIII-expressing GBMs, and a predicted nuclear interaction with EGFRvIII that alters the binding of STAT3 to DNA has been proposed as the mechanism (30). Currently, GC1008, a TGF β -specific antibody, and WP1066, a STAT3 inhibitor, are both in clinical development for the treatment of various cancers. Given the role of these factors in immunosuppression, either agent in combination with EGFR-specific immunotherapy agents could potentially be more efficacious in the treatment of malignant glioma.

The secretion of IL-6 by stromal cells into the tumor microenvironment can also promote tumor survival and block apoptosis, thereby resulting in therapeutic resistance (31). These effects are mediated through the IL6R/JAK1/STAT3 pathway and via Bcl-xL function (31). In the former case, signaling through IL6R/JAK1/STAT3 confers *de novo* resistance to irreversible EGFR TKI's in NSCLCs harboring a *T790M* secondary mutation (32). Additionally, TGF β -dependent IL-6 secretion confers resistance to the EGFR TKI erlotinib in lung tumor cells (33). In GBM, Inda *et al.* demonstrated that EGFRvIII-expressing cells secrete IL-6 in order to activate the wtEGFR-expressing cells through a paracrine cytokine signaling circuit (4). The

secretion of IL-6 was also correlated with enhanced tumorigenic growth of U87 glioma cells (4). An additional mechanism whereby EGFR mediates the production of IL-6 via the inhibitor of differentiation 3 (ID3) was shown to promote tumor cell heterogeneity in GSC populations. These populations as discussed above, are major sources of therapeutic resistance due to their dominant impact on the tumor microenvironment in order to maintain their "stemness" (34). IL-6 has been implicated in drug resistance and survival signaling through both the Bcl-xL and STAT3 pathways in prostate cancer (35). Interestingly, these effects were significantly attenuated by transfection with anti-sense Bcl-xL olignonucleotides, further supporting the notion that RNA-based therapies should be more closely considered as a therapeutic option in GBM therapy (35). Taken together, IL-6 secretion into the microenvironment reflects a common mechanism of acquired resistance to EGFR inhibition and strategies to overcome its effects are clearly warranted.

Stromal cells (e.g., endothelial cells) can also act independently in immunosuppression-mediated resistance. Work from Ricci-Vitiani et al. demonstrated that a large proportion of endothelial cells are found to harbor the same mutations as tumor cells (36). Thus, it is presumed that GBM-associated endothelial cells arise from tumor stem cells, and the stem cells that survive therapeutic insult acquire resistance through cell-cell interactions with such stromal cells in the microenvironment. Likewise, others have shown that extracellular matrix proteins as well as hypoxic conditions can also impart radioresistance to the stem cells (17). These findings suggest that conquering immunosuppression-mediated resistance requires targeting both the antigen-expressing tumor cells and the immunosuppressive activities afforded by the tumor microenvironment.

2.5 Histological Heterogeneity

Glioblastomas are considerably heterogeneous, and this intratumoral diversity is becoming increasingly appreciated as a predominant mechanism underlying anti-EGFR therapeutic resistance and tumor recurrence (37). Within these tumors lie mixed cytological subtypes, regional differences in gene expression, and varying representation of key genetic mutations and chromosomal alterations (4). Indeed, the WHO classification of CNS tumors has been restructured in terms of nomenclature to account for this variability (38). Furthermore, concerted efforts at providing a global description of the genetic abnormalities that are present in GBMs resulted in a geneexpression based classification of these tumors into four molecular subtypes: Proneural, Neural, Classical and Mesenchymal (39-41).

The Proneural subtype is associated with two major features: alterations of platelet-derived growth factor receptor alpha gene (*PDGFRA*) and mutations in isocitrate dehydrogenase 1 (*IDH1*) (39,42). Most of the known secondary GBMs are found in this class; consistent with the finding that younger age is also associated with this subtype. Additionally, global transcriptomic analysis revealed that the Proneural tumors resemble oligodendrocytes and express several genes involved in development (39). The Neural subgroup is characterized by expression of neuron markers and genes associated with biological processes such as neuron projection and axon and synaptic transmission. Interestingly, the expression patterns found within this class are most similar to normal brain, and its gene signature can only be concluded as reflecting a cell with a differentiated phenotype (39). Its association with the genetic signatures of multiple cell types (i.e., neural, astrocytes, oligodendrocytes) further demonstrates this genetic ambiguity.

The Mesenchymal subtype is predominantly made up of GBMs harboring aberrations in the neurofibromatosis type 1 gene (*NF1*) and is most reminiscent of astroglial cells (39). Genes associated with inflammation and the tumor necrosis factor (TNF) superfamily are highly expressed in this subtype, which correlates with the high percentage of necrosis that is evident in these tumors (39). Other distinguishable genetic abnormalities found in this subtype include high expression of the mesenchymal markers, chitinase 3-like 1 protein (CHI3L1) and hepatocyte growth factor receptor gene (MET), events from which the group name is derived (41).

The Classical gene signature is strongly associated with astrocytes and is distinguished as the subtype containing the most common genomic aberrations encountered in GBM. The majority of EGFR-amplified or mutated GBMs are found within this subtype, manifesting as polysomy and amplification of chromosome 7 (43). Loss of chromosome 10 (monosomy 10) is another highly frequent event in this subtype, but the overwhelming percentage of EGFR-altered tumors suggests that there is a focused preference for this genomic alteration in GBMs (39,40). This is confirmed by the frequent lack in additional common abnormalities in *TP53*, *NF1*, *PDGFRA* or *IDH1*, giving credence to the wave of "EGFR-driven" therapeutic development in GBM (39).

More than one subtype may be found within the same tumor mass, reflecting spatial heterogeneity and another layer of complexity that would challenge patient stratification and treatment simply based on molecular features found within a single tumor specimen (44). Tumor recurrence is observed across all subtypes, but a difference in the degree of clinical response to aggressive therapy by subtype has been observed, ranging from the greatest reduction in mortality seen in the Classical and Mesenchymal subtypes down to no altered survival benefit in the Proneural subtype (39). In essence, some tumors may be inherently resistant to EGFR-targeted therapy simply because the bulk of the tumor is not majorly dependent on its receptor function.

In NSCLC, histological transformation has even been observed and is attributed to EGFR TKI resistance. Specifically, an epithelial to mesenchymal transition (EMT) and in rare cases, a conversion to small cell lung carcinoma (SCLC) can occur at the time of the development of resistance (12,45). EMT is characterized by loss of cell polarity and cell-cell adhesion, extensive actin cytoskeletal remodeling, and the acquisition of mesenchymal components (46). This process promotes tumor invasion, migration, and progression, and is mediated by a diverse list of cytokines, chemokines and growth factors. Co-targeting the relevant molecular pathways and EGFR can reverse EMT-mediated resistance (47). In the context of the NSCLC conversion to SCLC, the original EGFR activating mutation persists, suggesting that this transformation does not arise from *de novo* clones, but rather emerges from preexisting cancer cells (48). Cytotoxic chemotherapy and radiation, which is the standard of care for SCLC, is an effective treatment against this switch and in some cases actually primes the tumor to redevelop EGFR TKI susceptibility (45). GBMs; however, do not display evidence of phenotypic transformation upon tumor recurrence (39).

2.6 Compensatory RTK Pathways

Receptor tyrosine kinase (RTK) coactivation or crosstalk is a process in cancer cells defined by synchronous activation of two or more RTKs that supports network robustness and increases the diversity of signaling outcomes, using only a limited repertoire of intracellular signaling factors (49). This process reflects another form of intratumoral heterogeneity that may contribute to the maintenance of individual cell growth within GBMs. Additionally, it is presumed to be an acquired mechanism to reduce dependency on a single RTK, and thus plays a critical role in tumor response to targeted therapeutics (49,50). Indeed, other RTKs, such as the insulin-like growth factor 1 receptor (IGF1R), MET, and PDGFR α/β are altered in GBM, and their ability to compensate for EGFR has been implicated in the persistent activation of downstream survival signaling, even in the presence of EGFR inhibitors (50-53).

As mentioned in chapter 1, EGFR is a member of the HER family, which consists of other structurally similar members prone to form homo- and heterodimers with one another and display functional redundancy. In NSCLC, both HER2 and HER3 are involved in resistance to EGFR TKI's (12). HER2-mediated resistance is afforded by genetic alterations of the gene, while HER3 active heterodimers trigger pro-survival signaling via the PI3K/Akt pathway (54). Interestingly, GSC resistance to EGFR inhibition is mediated by sustained activation of AKT and mitogen activated protein kinase (MAPK) signaling pathways by way of HER2 and HER3 (55). Afatanib, а HER2 specific irreversible inhibitor, is effective against HER2overexpressing/mutated gastric cancers, but has shown limited activity as a single agent against recurrent GBMs (56). So far, HER3-targeting monoclonal antibodies

look promising based on preclinical studies in others cancers (i.e., head and neck, breast and lung), but have not been evaluated in GBM (12,57).

c-Met is a transmembrane RTK that is frequently amplified in human cancers and in the past was established as the second most common mechanism of resistance to EGFR-TKI's in NSCLC (12). *c-Met* amplified clones exist even before EGFR TKI treatment, suggesting that they are selected out by treatment (58). Interestingly, c-Met can also promote HER3-PI3K/AKT signaling through the stabilization of HER3 phosphorylation (59). This mechanism of action occurs even in the presence of gefitinib, reflecting a bonafide EGFR kinase-independent bypass pathway (60). In GBM, the ability for c-Met to compensate for EGFR is demonstrated in that treatment with erlotinib alone had no discernible effect on inhibiting the activation of the PI3K/AKT pathway in U87MG-EGFRvIII cells. However, when used in combination with the MET inhibitor SU11274, downstream signaling as measured by activated Akt and S6 ribosomal protein was significantly inhibited and anchorageindependent growth of these cells was also reduced (50,61). Similarly, activation of c-Met expression in response to EGFR inhibition leads to the survival of GBM tumor cells in a mouse model of glioblastoma and inhibiting MET reverses this phenotype (62).

IGF1R signaling through PI3K is another common compensatory pathway that mediates resistance to EGFR TKI's (63). Concomitant inhibition of both EGFR and IGF1R is required to effectively abort PI3K/Akt signaling in resistant cells and restore sensitivity to gefitinib (64). Aside from the PI3K/Akt pathway, IGF1R signal transduction also regulates the function of cyclin-dependent kinase inhibitor 1B (p27), by altering its activation and subcellular localization (65). Thus, IGF1R-p27-mediated cell proliferation and motility may represent another pathway that cancer cells utilize to circumvent EGFR inhibition.

The vascular endothelial growth factor and receptors (VEGF/VEGFRs) are the most common pro-angiogenic factors in GBM. Angiogenic signaling mediated by these molecules induces tumor cell proliferation and migration and suppresses apoptosis and drug penetration. In several human cancers, acquired resistance to EGFR inhibition is associated with increased secretion of VEGF (66). Additionally, the VEGF/VEGFR2 signaling pathway can promote the secretion of more VEGF through a feed-forward loop mechanism involving the PI3K/ mammalian target of rapamycin (mTOR) pathway (12). On the basis of the highly angiogenic nature of aggressive cancers, VEGFR targeting inhibitors have been evaluated for clinical use in many cancers, including GBM. So far, as monotherapies, durable responses have not been observed; however, dual inhibition of EGFR and VEGFR produces antitumor effects in several human cancers (66). This was observed even in the cases of acquired resistance to EGFR TKI's, validating VEGFR as a legitimate EGFR substitute.

PDGFRβ, which is usually suppressed by EGFRvIII through Akt/mTOR complex 1 (mTORC1) signaling, becomes transcriptionally de-repressed upon EGFR inhibition to restore GBM growth and survival, reflecting another established mechanism of resistance (67). Combined abrogation of EGFRvIII and PDGFRβ is then required to block glioma in vivo growth. Alternatively, coactivation of EGFR and PDGFRα occurs in GBM (50). Blocking PDGFRα alone results in only modest antitumor activity, yet PDGFRα inhibition is still required to completely eradicate survival signaling through the PI3K/Akt pathway in GBM. Thus, in multiple cancer types, the signaling outputs from RTKs converge and activate similar signaling

pathways such as PI3K/Akt. This suggests that oncogene "switching" afforded by RTK coactivation within the same tumor, may be a principal mechanism to achieve chemoresistance.

2.7 Aberrant Downstream Pathways

Deletions or decreased function in tumor suppressor genes may also lead to therapeutic resistance due to aberrant signaling. Persistent activation of the PI3K/Akt/mTOR pathway can also be promoted by loss of the phosphatase and tensin homolog deleted on chromosome 10 (*PTEN*) gene, occurring in around 40-50% of GBMs (68). Specifically, the *PTEN* gene encodes a dual specificity lipid and protein phosphatase that dephosphorylates PIP₃ to impede signal flux through the PI3K pathway. Co-expression of EGFRvIII and wild type PTEN has been shown to correlate with tumor sensitivity to EGFR TKIs; however, there is considerable variability in patient response despite the presence of this genetic signature (6,69). In fact, Fenton *et al.* demonstrated that PTEN that is intact, but is phosphorylated at tyrosine 240, is frequently encountered in GBM and confers resistance to EGFR inhibitors (70). Hence, PTEN function may prove to be a more superior determinant compared to simply its expression status.

PTEN-deficient glioblastoma cell lines can also employ the autophagic process as a survival pathway of escape (71-73). Specifically, overexpression and accumulation of α B-crystallin following erlotinib treatment was observed in DBTRG-05 and U87 glioma cells (71). This molecule is a heat shock protein that can impair caspase activation, and thus block apoptosis. Inhibiting either autophagy or autophagosome maturation was shown to increase the death-inducing activity of both erlotinib and an mTOR inhibitor, PI-103 (71,72).

Activating mutation and/or amplification of the *PIK3CA* gene, which encodes the p110 α catalytic subunit of PI3K, has also been linked to resistance to EGFR inhibition (69). These aberrations confer increased PI3K activity that is unchecked by

PTEN regulation, resulting in cellular transformation and amplified mitogenic effects via Akt (74). Interestingly, *PIK3CA* mutations are mutually exclusive to the PTEN deletions, but are collectively the cause of mutational activation of the PI3K/Akt pathway that is encountered in about 50% of glioblastomas (75). Taken together, the antitumor activity of many EGFR inhibitors can ultimately be negated due to defects in components of the targeted pathway, suggesting that cooperation between TKIs and inhibitors of predictive compensatory mechanisms is required to achieve a more robust anti-tumor effect.

2.8 Secondary Mutations

Finally, the acquisition of secondary mutations in oncogenic drivers is highly associated with the onset of acquired resistance to anti-EGFR therapies. Currently, more than 90% of the known EGFR secondary mutations occur in exons 19-21 (12). In NSCLC, the most common mechanism of resistance is the emergence of *T790M*, a point mutation located in exon 20 of the tyrosine kinase domain of EGFR. This mutation is present in ~50-60% of resistance cases, and occurs at what is defined as a "gatekeeper" residue, which is important for regulating inhibitor specificity within the ATP binding pocket (12). A major consequence of this mutation is enhanced affinity for ATP, which would effectively compete against TKI's, whose target region is also the ATP binding pocket. Interestingly, tumors harboring the *T790M* mutation display extremely slow growth rates, which results in a better prognosis compared to recurrent tumors that lack this aberration (76).

Despite this prognostic advantage, the *T790M* mutation is still considered highly oncogenic, possessing enhanced phosphorylating activity and conferring resistance to both first-generation EGFR inhibitors (i.e., erlotinib, gefitinib) and second-generation EGFR inhibitors (i.e., afatanib) (12,77). For these reasons, T790M-specific third-generation covalent EGFR inhibitors, osimertinib and rociletinib, have been developed. So far, these agents have displayed significant efficacy in mouse models and in the clinic, are well tolerated and have led to response rates in about 60% of patients harboring the *T790M* mutation (77). Nevertheless, patients develop resistance once again even on this treatment plan, suggesting that the development of some sort of drug resistance to targeted therapy is inevitable. Other

NSCLC include *D761Y*, *L747S* and *T854A* (12). These mutations can coexist with *T790M*, but the structural basis underlying their individual contributions to resistance is still unknown.

Preexisting minor clones with the *T790M* mutation have also been identified within treatment-naïve tumors, reflecting an association between this aberration and inherent resistance to EGFR inhibition (76). In this context, it is believed that treatment with TKI's introduces a selective pressure that supports the survival of these clones, which overall manifests as acquired resistance. To this end, newer evidence has established that the *T790M* recurrent tumor consists of both preexisting and *de novo* drug-tolerant cells (78). This suggests that cancer cells that persist following initial therapy may serve as an important reservoir that affords the emergence of acquired resistance in the clinic.

Other secondary mutations that impact therapeutic response to anti-EGFR therapies involve the *KRAS* gene. KRAS is a part of the RAS protein family along with HRAS and NRAS, and these proteins are GTPases that act as molecular mediators for a number of critical signaling processes. Thus, mutations that occur within these molecules promote oncogenic activity, manifesting as constitutive activation of these proteins and uncontrollable cell proliferation and survival (79). The association between *KRAS* mutations and acquired resistance to anti-EGFR therapies has been established in the context of colorectal cancers, specifically in response to the anti-EGFR monoclonal antibody cetuximab (79). The most common mutations found are *G13D* and *G12R*, and both are sufficient to confer resistance to cetuximab. Combinatorial therapy that antagonizes both the EGFR and MEK pathways is then required to overcome this resistance (79).

Analogous to the patterns underlying the emergence of the *T790M* mutation in NSCLC, it is known that drug resistance as a result of *KRAS* mutant alleles is due to both preexistent *KRAS* mutant and amplified clones and *de novo* evolution promoted by ongoing mutagenesis (79). Interestingly, these mutant alleles could be detected in the plasma of patients as early as 10 months prior to radiographic evidence of disease progression, reflecting biomarkers for early detection of risk for this specific mechanism of resistance (79). Yet, as often as secondary mutations have been linked to resistance to EGFR inhibition in other solid tumors, no such mutations have been implicated in GBM.



Figure 2.4. Summary of the predominant mechanisms of resistance to EGFR inhibition. (1) Brain tumor cells that are intractable to DNA damage-induced apoptosis may also tolerate apoptotic cues driven by TKI-mediated inhibition of EGFR. Combinatorial therapy using agents that can antagonize anti-apoptotic activity within cancer cells may overcome cross-resistance. (2) Efflux of EGFR TKIs and increased genetic stability support to the maintenance of CSC populations and tumor relapse. Additionally, cell-cell communication between the glioma stem cells and the non-stem tumor cells affords escape from tumor dormancy. (3) More than one molecular subtype may be represented in a given GBM, reflecting both histological and spatial heterogeneity that confer resistance. (4) Enhanced immunosuppression mediated by circulating growth factors, cytokines and suppressor T cells can inhibit the systemic immune responses generated by anti-EGFR immunotherapies. Additionally, circulating IL-6 in the tumor microenvironment can facilitate resistance via the JAK/STAT3/Bcl-xL pathway. (5) Compensatory/Aberrant signaling within GBM tumors may drive resistance to EGFR monotherapies due to: RTK co-activation, PTEN deletion/mutations/modifications (Y240 PTEN), PIK3CA mutations and cell-cell cross talk via secreted factors. (6) Secondary mutations in EGFR that alter the binding affinity of small molecule inhibitors commonly promote therapeutic resistance.

2.9 Discussion

The successful application of EGFR-targeted therapy for the treatment of glioblastoma has proven to be very challenging. Malignant brain tumors require a very complex signaling network that is not driven by EGFR alone, and this complexity dictates tumor sensitivity to EGFR-targeted therapies. In comparison with the progress made to uncover mechanisms of therapeutic resistance in other cancers, the GBM field is still in the budding stage. A deeper understanding of the intricate inter-relationships that underlie the pathobiology of this disease is required to achieve stable therapeutic responses to targeted agents.

It is clear that GBM tumors possess a number of intrinsic variables that would render monotherapies inept; yet, lack in the knowledge of a comprehensive role for these variables has immobilized the design of more effective treatments. Thus, there is still a great need to identify more determinants of resistance as a basis for novel therapeutic strategies that are specific to how GBMs recur. In this dissertation, we EGFRvIII-dependent EGFRvIII-independent characterize both and glioma populations. Such an approach yielded a previously unidentified mechanism of resistance to EGFR blockade in GBM, which emerged from a therapeutic response phenotype that had to be overcome for re-initiation of tumor growth. It is the goal of this thesis to highlight an important association between characterizing a therapeutic response to predict mechanisms of resistance, and linking this information to development of a tailored combination therapy based on identified mechanisms.

Chapter 2 contains portions of material as it appears in Current Cancer Drug Targets, Taylor, T. E., Furnari, F. B., Cavenee, W. K. Targeting EGFR for Treatment of Glioblastoma: Molecular Basis to Overcome Resistance. 12:197-209 (2012). The dissertation author is the first author of this review article. Additional data presented in chapter 2 have not been published previously and will not be submitted for publication. This work was supported by the NIH F31 NS076343 to (T.E. Taylor) and the NIH R01 NS080939 to (F. B. Furnari).

2.10. References

- 1. Engelman, J. A., and Settleman, J. (2008) Acquired resistance to tyrosine kinase inhibitors during cancer therapy. *Curr Opin Genet Dev* **18**, 73-79
- 2. Wykosky, J., Mukasa, A., Furnari, F., and Cavenee, W. K. (2010) Escape from targeted inhibition: the dark side of kinase inhibitor therapy. *Cell Cycle* **9**, 1661-1662
- 3. Engelman, J. A., and Janne, P. A. (2008) Mechanisms of acquired resistance to epidermal growth factor receptor tyrosine kinase inhibitors in non-small cell lung cancer. *Clinical cancer research : an official journal of the American Association for Cancer Research* **14**, 2895-2899
- 4. Inda, M. M., Bonavia, R., Mukasa, A., Narita, Y., Sah, D. W., Vandenberg, S., Brennan, C., Johns, T. G., Bachoo, R., Hadwiger, P., Tan, P., Depinho, R. A., Cavenee, W., and Furnari, F. (2010) Tumor heterogeneity is an active process maintained by a mutant EGFR-induced cytokine circuit in glioblastoma. *Genes* & development 24, 1731-1745
- 5. Krakstad, C., and Chekenya, M. (2010) Survival signalling and apoptosis resistance in glioblastomas: opportunities for targeted therapeutics. *Mol Cancer* **9**, 135
- Mellinghoff, I. K., Wang, M. Y., Vivanco, I., Haas-Kogan, D. A., Zhu, S., Dia, E. Q., Lu, K. V., Yoshimoto, K., Huang, J. H., Chute, D. J., Riggs, B. L., Horvath, S., Liau, L. M., Cavenee, W. K., Rao, P. N., Beroukhim, R., Peck, T. C., Lee, J. C., Sellers, W. R., Stokoe, D., Prados, M., Cloughesy, T. F., Sawyers, C. L., and Mischel, P. S. (2005) Molecular determinants of the response of glioblastomas to EGFR kinase inhibitors. *The New England journal of medicine* **353**, 2012-2024
- 7. Nagane, M., Levitzki, A., Gazit, A., Cavenee, W. K., and Huang, H. J. (1998) Drug resistance of human glioblastoma cells conferred by a tumor-specific mutant epidermal growth factor receptor through modulation of Bcl-XL and caspase-3-like proteases. *Proceedings of the National Academy of Sciences of the United States of America* **95**, 5724-5729
- Mukasa, A., Wykosky, J., Ligon, K. L., Chin, L., Cavenee, W. K., and Furnari, F. (2010) Mutant EGFR is required for maintenance of glioma growth in vivo, and its ablation leads to escape from receptor dependence. *Proceedings of*

the National Academy of Sciences of the United States of America **107**, 2616-2621

- 9. Wykosky, J., Hu, J., Gomez, G. G., Taylor, T., Villa, G. R., Pizzo, D., VandenBerg, S. R., Thorne, A. H., Chen, C. C., Mischel, P. S., Gonias, S. L., Cavenee, W. K., and Furnari, F. B. (2015) A urokinase receptor-Bim signaling axis emerges during EGFR inhibitor resistance in mutant EGFR glioblastoma. *Cancer research* **75**, 394-404
- Ng, K. P., Hillmer, A. M., Chuah, C. T., Juan, W. C., Ko, T. K., Teo, A. S., Ariyaratne, P. N., Takahashi, N., Sawada, K., Fei, Y., Soh, S., Lee, W. H., Huang, J. W., Allen, J. C., Jr., Woo, X. Y., Nagarajan, N., Kumar, V., Thalamuthu, A., Poh, W. T., Ang, A. L., Mya, H. T., How, G. F., Yang, L. Y., Koh, L. P., Chowbay, B., Chang, C. T., Nadarajan, V. S., Chng, W. J., Than, H., Lim, L. C., Goh, Y. T., Zhang, S., Poh, D., Tan, P., Seet, J. E., Ang, M. K., Chau, N. M., Ng, Q. S., Tan, D. S., Soda, M., Isobe, K., Nothen, M. M., Wong, T. Y., Shahab, A., Ruan, X., Cacheux-Rataboul, V., Sung, W. K., Tan, E. H., Yatabe, Y., Mano, H., Soo, R. A., Chin, T. M., Lim, W. T., Ruan, Y., and Ong, S. T. (2012) A common BIM deletion polymorphism mediates intrinsic resistance and inferior responses to tyrosine kinase inhibitors in cancer. *Nature medicine* 18, 521-528
- 11. Nakagawa, T., Takeuchi, S., Yamada, T., Ebi, H., Sano, T., Nanjo, S., Ishikawa, D., Sato, M., Hasegawa, Y., Sekido, Y., and Yano, S. (2013) EGFR-TKI resistance due to BIM polymorphism can be circumvented in combination with HDAC inhibition. *Cancer research* **73**, 2428-2434
- 12. Huang, L., and Fu, L. (2015) Mechanisms of resistance to EGFR tyrosine kinase inhibitors. *Acta pharmaceutica Sinica. B* **5**, 390-401
- 13. Weihua, Z., Tsan, R., Huang, W. C., Wu, Q., Chiu, C. H., Fidler, I. J., and Hung, M. C. (2008) Survival of cancer cells is maintained by EGFR independent of its kinase activity. *Cancer Cell* **13**, 385-393
- 14. Network. (2008) Comprehensive genomic characterization defines human glioblastoma genes and core pathways. *Nature* **455**, 1061-1068
- Parsons, D. W., Jones, S., Zhang, X., Lin, J. C., Leary, R. J., Angenendt, P., Mankoo, P., Carter, H., Siu, I. M., Gallia, G. L., Olivi, A., McLendon, R., Rasheed, B. A., Keir, S., Nikolskaya, T., Nikolsky, Y., Busam, D. A., Tekleab, H., Diaz, L. A., Jr., Hartigan, J., Smith, D. R., Strausberg, R. L., Marie, S. K.,

Shinjo, S. M., Yan, H., Riggins, G. J., Bigner, D. D., Karchin, R., Papadopoulos, N., Parmigiani, G., Vogelstein, B., Velculescu, V. E., and Kinzler, K. W. (2008) An integrated genomic analysis of human glioblastoma multiforme. *Science* **321**, 1807-1812

- 16. Wen, P. Y., and Kesari, S. (2008) Malignant gliomas in adults. *The New England journal of medicine* **359**, 492-507
- 17. Bao, S., Wu, Q., McLendon, R. E., Hao, Y., Shi, Q., Hjelmeland, A. B., Dewhirst, M. W., Bigner, D. D., and Rich, J. N. (2006) Glioma stem cells promote radioresistance by preferential activation of the DNA damage response. *Nature* **444**, 756-760
- 18. Sanchez-Garcia, I., Vicente-Duenas, C., and Cobaleda, C. (2007) The theoretical basis of cancer-stem-cell-based therapeutics of cancer: can it be put into practice? *Bioessays* **29**, 1269-1280
- 19. Gilbert, L. A., and Hemann, M. T. (2010) DNA damage-mediated induction of a chemoresistant niche. *Cell* **143**, 355-366
- Shi, Z., Tiwari, A. K., Shukla, S., Robey, R. W., Singh, S., Kim, I. W., Bates, S. E., Peng, X., Abraham, I., Ambudkar, S. V., Talele, T. T., Fu, L. W., and Chen, Z. S. (2011) Sildenafil reverses ABCB1- and ABCG2-mediated chemotherapeutic drug resistance. *Cancer research* **71**, 3029-3041
- 21. Gong, X., Schwartz, P. H., Linskey, M. E., and Bota, D. A. Neural stem/progenitors and glioma stem-like cells have differential sensitivity to chemotherapy. *Neurology* **76**, 1126-1134
- 22. Guryanova, O. A., Wu, Q., Cheng, L., Lathia, J. D., Huang, Z., Yang, J., MacSwords, J., Eyler, C. E., McLendon, R. E., Heddleston, J. M., Shou, W., Hambardzumyan, D., Lee, J., Hjelmeland, A. B., Sloan, A. E., Bredel, M., Stark, G. R., Rich, J. N., and Bao, S. (2011) Nonreceptor tyrosine kinase BMX maintains self-renewal and tumorigenic potential of glioblastoma stem cells by activating STAT3. *Cancer Cell* **19**, 498-511
- Zhou, W., Ke, S. Q., Huang, Z., Flavahan, W., Fang, X., Paul, J., Wu, L., Sloan, A. E., McLendon, R. E., Li, X., Rich, J. N., and Bao, S. (2015) Periostin secreted by glioblastoma stem cells recruits M2 tumour-associated macrophages and promotes malignant growth. *Nature cell biology* 17, 170-182

- Ghajar, C. M., Peinado, H., Mori, H., Matei, I. R., Evason, K. J., Brazier, H., Almeida, D., Koller, A., Hajjar, K. A., Stainier, D. Y., Chen, E. I., Lyden, D., and Bissell, M. J. (2013) The perivascular niche regulates breast tumour dormancy. *Nature cell biology* 15, 807-817
- Heimberger, A. B., Kong, L. Y., Abou-Ghazal, M., Reina-Ortiz, C., Yang, D. S., Wei, J., Qiao, W., Schmittling, R. J., Archer, G. E., Sampson, J. H., Hiraoka, N., Priebe, W., Fuller, G. N., and Sawaya, R. (2009) The role of tregs in human glioma patients and their inhibition with a novel STAT-3 inhibitor. *Clin Neurosurg* 56, 98-106
- Johns, T. G., Perera, R. M., Vernes, S. C., Vitali, A. A., Cao, D. X., Cavenee, W. K., Scott, A. M., and Furnari, F. B. (2007) The efficacy of epidermal growth factor receptor-specific antibodies against glioma xenografts is influenced by receptor levels, activation status, and heterodimerization. *Clin Cancer Res* 13, 1911-1925
- 27. Weiner, L. M., Surana, R., and Wang, S. (2010) Monoclonal antibodies: versatile platforms for cancer immunotherapy. *Nat Rev Immunol* **10**, 317-327
- Fecci, P. E., Mitchell, D. A., Whitesides, J. F., Xie, W., Friedman, A. H., Archer, G. E., Herndon, J. E., 2nd, Bigner, D. D., Dranoff, G., and Sampson, J. H. (2006) Increased regulatory T-cell fraction amidst a diminished CD4 compartment explains cellular immune defects in patients with malignant glioma. *Cancer Res* 66, 3294-3302
- 29. Birner, P., Toumangelova-Uzeir, K., Natchev, S., and Guentchev, M. (2010) STAT3 tyrosine phosphorylation influences survival in glioblastoma. *Journal of neuro-oncology* **100**, 339-343
- 30. Lo, H. W., Hsu, S. C., Ali-Seyed, M., Gunduz, M., Xia, W., Wei, Y., Bartholomeusz, G., Shih, J. Y., and Hung, M. C. (2005) Nuclear interaction of EGFR and STAT3 in the activation of the iNOS/NO pathway. *Cancer cell* **7**, 575-589
- 31. Dalton, W. S. (1999) The tumor microenvironment as a determinant of drug response and resistance. *Drug Resist Updat* **2**, 285-288
- Kim, S. M., Kwon, O. J., Hong, Y. K., Kim, J. H., Solca, F., Ha, S. J., Soo, R.
 A., Christensen, J. G., Lee, J. H., and Cho, B. C. (2012) Activation of IL-6R/JAK1/STAT3 signaling induces de novo resistance to irreversible EGFR

inhibitors in non-small cell lung cancer with T790M resistance mutation. *Molecular cancer therapeutics* **11**, 2254-2264

- Yao, Z., Fenoglio, S., Gao, D. C., Camiolo, M., Stiles, B., Lindsted, T., Schlederer, M., Johns, C., Altorki, N., Mittal, V., Kenner, L., and Sordella, R. (2010) TGF-beta IL-6 axis mediates selective and adaptive mechanisms of resistance to molecular targeted therapy in lung cancer. *Proc Natl Acad Sci U S A* 107, 15535-15540
- Jin, X., Yin, J., Kim, S. H., Sohn, Y. W., Beck, S., Lim, Y. C., Nam, D. H., Choi, Y. J., and Kim, H. (2011) EGFR-AKT-Smad Signaling Promotes Formation of Glioma Stem-like Cells and Tumor Angiogenesis by ID3-Driven Cytokine Induction. *Cancer Res* **71**, 7125-7134
- 35. Pu, Y. S., Hour, T. C., Chuang, S. E., Cheng, A. L., Lai, M. K., and Kuo, M. L. (2004) Interleukin-6 is responsible for drug resistance and anti-apoptotic effects in prostatic cancer cells. *Prostate* **60**, 120-129
- Ricci-Vitiani, L., Pallini, R., Biffoni, M., Todaro, M., Invernici, G., Cenci, T., Maira, G., Parati, E. A., Stassi, G., Larocca, L. M., and De Maria, R. (2010) Tumour vascularization via endothelial differentiation of glioblastoma stem-like cells. *Nature* 468, 824-828
- 37. Bedard, P. L., Hansen, A. R., Ratain, M. J., and Siu, L. L. (2013) Tumour heterogeneity in the clinic. *Nature* **501**, 355-364
- Louis, D. N., Perry, A., Reifenberger, G., von Deimling, A., Figarella-Branger, D., Cavenee, W. K., Ohgaki, H., Wiestler, O. D., Kleihues, P., and Ellison, D. W. (2016) The 2016 World Health Organization Classification of Tumors of the Central Nervous System: a summary. *Acta neuropathologica*
- Verhaak, R. G., Hoadley, K. A., Purdom, E., Wang, V., Qi, Y., Wilkerson, M. D., Miller, C. R., Ding, L., Golub, T., Mesirov, J. P., Alexe, G., Lawrence, M., O'Kelly, M., Tamayo, P., Weir, B. A., Gabriel, S., Winckler, W., Gupta, S., Jakkula, L., Feiler, H. S., Hodgson, J. G., James, C. D., Sarkaria, J. N., Brennan, C., Kahn, A., Spellman, P. T., Wilson, R. K., Speed, T. P., Gray, J. W., Meyerson, M., Getz, G., Perou, C. M., Hayes, D. N., and Cancer Genome Atlas Research, N. (2010) Integrated genomic analysis identifies clinically relevant subtypes of glioblastoma characterized by abnormalities in PDGFRA, IDH1, EGFR, and NF1. *Cancer cell* **17**, 98-110

- Brennan, C. W., Verhaak, R. G., McKenna, A., Campos, B., Noushmehr, H., Salama, S. R., Zheng, S., Chakravarty, D., Sanborn, J. Z., Berman, S. H., Beroukhim, R., Bernard, B., Wu, C. J., Genovese, G., Shmulevich, I., Barnholtz-Sloan, J., Zou, L., Vegesna, R., Shukla, S. A., Ciriello, G., Yung, W. K., Zhang, W., Sougnez, C., Mikkelsen, T., Aldape, K., Bigner, D. D., Van Meir, E. G., Prados, M., Sloan, A., Black, K. L., Eschbacher, J., Finocchiaro, G., Friedman, W., Andrews, D. W., Guha, A., Iacocca, M., O'Neill, B. P., Foltz, G., Myers, J., Weisenberger, D. J., Penny, R., Kucherlapati, R., Perou, C. M., Hayes, D. N., Gibbs, R., Marra, M., Mills, G. B., Lander, E., Spellman, P., Wilson, R., Sander, C., Weinstein, J., Meyerson, M., Gabriel, S., Laird, P. W., Haussler, D., Getz, G., Chin, L., and Network, T. R. (2013) The somatic genomic landscape of glioblastoma. *Cell* 155, 462-477
- Phillips, H. S., Kharbanda, S., Chen, R., Forrest, W. F., Soriano, R. H., Wu, T. D., Misra, A., Nigro, J. M., Colman, H., Soroceanu, L., Williams, P. M., Modrusan, Z., Feuerstein, B. G., and Aldape, K. (2006) Molecular subclasses of high-grade glioma predict prognosis, delineate a pattern of disease progression, and resemble stages in neurogenesis. *Cancer cell* 9, 157-173
- 42. Parker, N. R., Khong, P., Parkinson, J. F., Howell, V. M., and Wheeler, H. R. (2015) Molecular heterogeneity in glioblastoma: potential clinical implications. *Frontiers in oncology* **5**, 55
- Lopez-Gines, C., Cerda-Nicolas, M., Gil-Benso, R., Pellin, A., Lopez-Guerrero, J. A., Callaghan, R., Benito, R., Roldan, P., Piquer, J., Llacer, J., and Barbera, J. (2005) Association of chromosome 7, chromosome 10 and EGFR gene amplification in glioblastoma multiforme. *Clinical neuropathology* 24, 209-218
- 44. Sottoriva, A., Spiteri, I., Piccirillo, S. G., Touloumis, A., Collins, V. P., Marioni, J. C., Curtis, C., Watts, C., and Tavare, S. (2013) Intratumor heterogeneity in human glioblastoma reflects cancer evolutionary dynamics. *Proceedings of the National Academy of Sciences of the United States of America* **110**, 4009-4014
- 45. Suda, K., Murakami, I., Sakai, K., Mizuuchi, H., Shimizu, S., Sato, K., Tomizawa, K., Tomida, S., Yatabe, Y., Nishio, K., and Mitsudomi, T. (2015) Small cell lung cancer transformation and T790M mutation: complimentary roles in acquired resistance to kinase inhibitors in lung cancer. *Scientific reports* **5**, 14447
- 46. Tsai, J. H., and Yang, J. (2013) Epithelial-mesenchymal plasticity in carcinoma metastasis. *Genes & development* **27**, 2192-2206

47. Wilson, C., Nicholes, K., Bustos, D., Lin, E., Song, Q., Stephan, J. P., Kirkpatrick, D. S., and Settleman, J. (2014) Overcoming EMT-associated resistance to anti-cancer drugs via Src/FAK pathway inhibition. *Oncotarget* **5**, 7328-7341

- Sequist, L. V., Waltman, B. A., Dias-Santagata, D., Digumarthy, S., Turke, A. B., Fidias, P., Bergethon, K., Shaw, A. T., Gettinger, S., Cosper, A. K., Akhavanfard, S., Heist, R. S., Temel, J., Christensen, J. G., Wain, J. C., Lynch, T. J., Vernovsky, K., Mark, E. J., Lanuti, M., lafrate, A. J., Mino-Kenudson, M., and Engelman, J. A. (2011) Genotypic and histological evolution of lung cancers acquiring resistance to EGFR inhibitors. *Science translational medicine* **3**, 75ra26
- 49. Xu, A. M., and Huang, P. H. (2010) Receptor tyrosine kinase coactivation networks in cancer. *Cancer research* **70**, 3857-3860
- Stommel, J. M., Kimmelman, A. C., Ying, H., Nabioullin, R., Ponugoti, A. H., Wiedemeyer, R., Stegh, A. H., Bradner, J. E., Ligon, K. L., Brennan, C., Chin, L., and DePinho, R. A. (2007) Coactivation of receptor tyrosine kinases affects the response of tumor cells to targeted therapies. *Science* **318**, 287-290
- Hu, J., Jo, M., Cavenee, W. K., Furnari, F., VandenBerg, S. R., and Gonias, S. L. (2011) Crosstalk between the urokinase-type plasminogen activator receptor and EGF receptor variant III supports survival and growth of glioblastoma cells. *Proc Natl Acad Sci U S A* **108**, 15984-15989
- Sharma, S. V., Lee, D. Y., Li, B., Quinlan, M. P., Takahashi, F., Maheswaran, S., McDermott, U., Azizian, N., Zou, L., Fischbach, M. A., Wong, K. K., Brandstetter, K., Wittner, B., Ramaswamy, S., Classon, M., and Settleman, J. (2010) A chromatin-mediated reversible drug-tolerant state in cancer cell subpopulations. *Cell* 141, 69-80
- Snuderl, M., Fazlollahi, L., Le, L. P., Nitta, M., Zhelyazkova, B. H., Davidson, C. J., Akhavanfard, S., Cahill, D. P., Aldape, K. D., Betensky, R. A., Louis, D. N., and lafrate, A. J. (2011) Mosaic Amplification of Multiple Receptor Tyrosine Kinase Genes in Glioblastoma. *Cancer Cell* 20, 810-817
- 54. Landi, L., and Cappuzzo, F. (2013) HER2 and lung cancer. *Expert review of anticancer therapy* **13**, 1219-1228

- 55. Clark, P. A., Iida, M., Treisman, D. M., Kalluri, H., Ezhilan, S., Zorniak, M., Wheeler, D. L., and Kuo, J. S. (2012) Activation of multiple ERBB family receptors mediates glioblastoma cancer stem-like cell resistance to EGFR-targeted inhibition. *Neoplasia* **14**, 420-428
- 56. Reardon, D. A., Nabors, L. B., Mason, W. P., Perry, J. R., Shapiro, W., Kavan, P., Mathieu, D., Phuphanich, S., Cseh, A., Fu, Y., Cong, J., Wind, S., Eisenstat, D. D., Group, B. I. T., and the Canadian Brain Tumour, C. (2015) Phase I/randomized phase II study of afatinib, an irreversible ErbB family blocker, with or without protracted temozolomide in adults with recurrent glioblastoma. *Neuro-oncology* **17**, 430-439
- Schoeberl, B., Faber, A. C., Li, D., Liang, M. C., Crosby, K., Onsum, M., Burenkova, O., Pace, E., Walton, Z., Nie, L., Fulgham, A., Song, Y., Nielsen, U. B., Engelman, J. A., and Wong, K. K. (2010) An ErbB3 antibody, MM-121, is active in cancers with ligand-dependent activation. *Cancer research* 70, 2485-2494
- 58. Kubo, T., Yamamoto, H., Lockwood, W. W., Valencia, I., Soh, J., Peyton, M., Jida, M., Otani, H., Fujii, T., Ouchida, M., Takigawa, N., Kiura, K., Shimizu, K., Date, H., Minna, J. D., Varella-Garcia, M., Lam, W. L., Gazdar, A. F., and Toyooka, S. (2009) MET gene amplification or EGFR mutation activate MET in lung cancers untreated with EGFR tyrosine kinase inhibitors. *International journal of cancer* **124**, 1778-1784
- Engelman, J. A., Zejnullahu, K., Mitsudomi, T., Song, Y., Hyland, C., Park, J. O., Lindeman, N., Gale, C. M., Zhao, X., Christensen, J., Kosaka, T., Holmes, A. J., Rogers, A. M., Cappuzzo, F., Mok, T., Lee, C., Johnson, B. E., Cantley, L. C., and Janne, P. A. (2007) MET amplification leads to gefitinib resistance in lung cancer by activating ERBB3 signaling. *Science* **316**, 1039-1043
- 60. Nakata, S., Tanaka, H., Ito, Y., Hara, M., Fujita, M., Kondo, E., Kanemitsu, Y., Yatabe, Y., and Nakanishi, H. (2014) Deficient HER3 expression in poorlydifferentiated colorectal cancer cells enhances gefitinib sensitivity. *International journal of oncology* **45**, 1583-1593
- Huang, P. H., Mukasa, A., Bonavia, R., Flynn, R. A., Brewer, Z. E., Cavenee, W. K., Furnari, F. B., and White, F. M. (2007) Quantitative analysis of EGFRvIII cellular signaling networks reveals a combinatorial therapeutic strategy for glioblastoma. *Proceedings of the National Academy of Sciences* of the United States of America **104**, 12867-12872

- Jun, H. J., Acquaviva, J., Chi, D., Lessard, J., Zhu, H., Woolfenden, S., Bronson, R. T., Pfannl, R., White, F., Housman, D. E., Iyer, L., Whittaker, C. A., Boskovitz, A., Raval, A., and Charest, A. (2011) Acquired MET expression confers resistance to EGFR inhibition in a mouse model of glioblastoma multiforme. *Oncogene*, 1-12
- Guix, M., Faber, A. C., Wang, S. E., Olivares, M. G., Song, Y., Qu, S., Rinehart, C., Seidel, B., Yee, D., Arteaga, C. L., and Engelman, J. A. (2008) Acquired resistance to EGFR tyrosine kinase inhibitors in cancer cells is mediated by loss of IGF-binding proteins. *The Journal of clinical investigation* 118, 2609-2619
- Peled, N., Wynes, M. W., Ikeda, N., Ohira, T., Yoshida, K., Qian, J., Ilouze, M., Brenner, R., Kato, Y., Mascaux, C., and Hirsch, F. R. (2013) Insulin-like growth factor-1 receptor (IGF-1R) as a biomarker for resistance to the tyrosine kinase inhibitor gefitinib in non-small cell lung cancer. *Cellular oncology* 36, 277-288
- 65. Jameson, M. J., Taniguchi, L. E., VanKoevering, K. K., Stuart, M. M., Francom, C. R., Mendez, R. E., Beckler, A. D., Carlson, H. T., Thomas, C. Y., and Khalil, A. A. (2013) Activation of the insulin-like growth factor-1 receptor alters p27 regulation by the epidermal growth factor receptor in oral squamous carcinoma cells. *Journal of oral pathology & medicine : official publication of the International Association of Oral Pathologists and the American Academy of Oral Pathology* **42**, 332-338
- 66. Bianco, R., Rosa, R., Damiano, V., Daniele, G., Gelardi, T., Garofalo, S., Tarallo, V., De Falco, S., Melisi, D., Benelli, R., Albini, A., Ryan, A., Ciardiello, F., and Tortora, G. (2008) Vascular endothelial growth factor receptor-1 contributes to resistance to anti-epidermal growth factor receptor drugs in human cancer cells. *Clinical cancer research : an official journal of the American Association for Cancer Research* 14, 5069-5080
- Akhavan, D., Pourzia, A. L., Nourian, A. A., Williams, K. J., Nathanson, D., Babic, I., Villa, G. R., Tanaka, K., Nael, A., Yang, H., Dang, J., Vinters, H. V., Yong, W. H., Flagg, M., Tamanoi, F., Sasayama, T., James, C. D., Kornblum, H. I., Cloughesy, T. F., Cavenee, W. K., Bensinger, S. J., and Mischel, P. S. (2013) De-repression of PDGFRbeta transcription promotes acquired resistance to EGFR tyrosine kinase inhibitors in glioblastoma patients. *Cancer discovery* **3**, 534-547

- 68. Di Cristofano, A., and Pandolfi, P. P. (2000) The multiple roles of PTEN in tumor suppression. *Cell* **100**, 387-390
- 69. Mellinghoff, I. K., Cloughesy, T. F., and Mischel, P. S. (2007) PTEN-mediated resistance to epidermal growth factor receptor kinase inhibitors. *Clinical cancer research : an official journal of the American Association for Cancer Research* **13**, 378-381
- 70. Fenton, T. R., Nathanson, D., Ponte de Albuquerque, C., Kuga, D., Iwanami, A., Dang, J., Yang, H., Tanaka, K., Oba-Shinjo, S. M., Uno, M., Inda, M. M., Wykosky, J., Bachoo, R. M., James, C. D., DePinho, R. A., Vandenberg, S. R., Zhou, H., Marie, S. K., Mischel, P. S., Cavenee, W. K., and Furnari, F. B. (2012) Resistance to EGF receptor inhibitors in glioblastoma mediated by phosphorylation of the PTEN tumor suppressor at tyrosine 240. *Proceedings of the National Academy of Sciences of the United States of America* 109, 14164-14169
- 71. Eimer, S., Belaud-Rotureau, M. A., Airiau, K., Jeanneteau, M., Laharanne, E., Veron, N., Vital, A., Loiseau, H., Merlio, J. P., and Belloc, F. (2011) Autophagy inhibition cooperates with erlotinib to induce glioblastoma cell death. *Cancer Biol Ther* **11**, 1017-1027
- 72. Fan, Q. W., and Weiss, W. A. (2011) Autophagy and Akt promote survival in glioma. *Autophagy* **7**, 536-538
- 73. Fan, Q. W., Cheng, C., Hackett, C., Feldman, M., Houseman, B. T., Nicolaides, T., Haas-Kogan, D., James, C. D., Oakes, S. A., Debnath, J., Shokat, K. M., and Weiss, W. A. (2010) Akt and autophagy cooperate to promote survival of drug-resistant glioma. *Sci Signal* **3**, ra81
- 74. Karakas, B., Bachman, K. E., and Park, B. H. (2006) Mutation of the PIK3CA oncogene in human cancers. *British journal of cancer* **94**, 455-459
- Bleeker, F. E., Lamba, S., Zanon, C., Molenaar, R. J., Hulsebos, T. J., Troost, D., van Tilborg, A. A., Vandertop, W. P., Leenstra, S., van Noorden, C. J., and Bardelli, A. (2014) Mutational profiling of kinases in glioblastoma. *BMC cancer* 14, 718
- 76. Stewart, E. L., Tan, S. Z., Liu, G., and Tsao, M. S. (2015) Known and putative mechanisms of resistance to EGFR targeted therapies in NSCLC patients with EGFR mutations-a review. *Translational lung cancer research* **4**, 67-81

- 77. Engel, J., Lategahn, J., and Rauh, D. (2016) Hope and Disappointment: Covalent Inhibitors to Overcome Drug Resistance in Non-Small Cell Lung Cancer. *ACS medicinal chemistry letters* **7**, 2-5
- 78. Hata, A. N., Niederst, M. J., Archibald, H. L., Gomez-Caraballo, M., Siddiqui, F. M., Mulvey, H. E., Maruvka, Y. E., Ji, F., Bhang, H. E., Krishnamurthy Radhakrishna, V., Siravegna, G., Hu, H., Raoof, S., Lockerman, E., Kalsy, A., Lee, D., Keating, C. L., Ruddy, D. A., Damon, L. J., Crystal, A. S., Costa, C., Piotrowska, Z., Bardelli, A., Iafrate, A. J., Sadreyev, R. I., Stegmeier, F., Getz, G., Sequist, L. V., Faber, A. C., and Engelman, J. A. (2016) Tumor cells can follow distinct evolutionary paths to become resistant to epidermal growth factor receptor inhibition. *Nature medicine* 22, 262-269
- Misale, S., Yaeger, R., Hobor, S., Scala, E., Janakiraman, M., Liska, D., Valtorta, E., Schiavo, R., Buscarino, M., Siravegna, G., Bencardino, K., Cercek, A., Chen, C. T., Veronese, S., Zanon, C., Sartore-Bianchi, A., Gambacorta, M., Gallicchio, M., Vakiani, E., Boscaro, V., Medico, E., Weiser, M., Siena, S., Di Nicolantonio, F., Solit, D., and Bardelli, A. (2012) Emergence of KRAS mutations and acquired resistance to anti-EGFR therapy in colorectal cancer. *Nature* **486**, 532-536

Chapter 3

Overcoming oxidative DNA damage supports escape from dependence on EGFRvIII

3.1 Introduction

Mutant EGFRvIII arises from of an in-frame deletion of 801 base pairs from the extracellular domain of EGFR and possesses ligand-independent constitutive tyrosine kinase activity (1). The ligand-independent nature of EGFRvIII causes defective endocytosis-mediated receptor internalization, resulting in increased stability of the receptor on the cell surface and amplified mitogenic effects (2). EGFRvIII has been detected in as many as 60% of GBMs at the protein level and in various other cancers including breast, lung, prostate, and non-small cell lung carcinoma; however, it has not been found in normal tissues (3). Furthermore, overexpression of EGFRvIII is a negative prognostic factor in GBM (2). Based on these findings, EGFRvIII has been pursued as a tumor-specific therapeutic target for GBM, but only in recent years has its requirement for the maintenance of GBM growth been established.

In 2010, Mukasa *et al.* demonstrated that EGFRvIII is crucial for the maintenance of glioma growth in vivo (4). In this work, U373MG GBM cells were engineered with doxycycline (DOX)-repressible EGFRvIII, and then engrafted subcutaneously into nude mice. In vivo tumorigenicity was only associated with EGFRvIII expression (-DOX), validating this model's dependence on EGFRvIII for tumorigenesis (4). However, when EGFRvIII expression was silenced (+DOX), a significant decline in tumor growth and subsequent cytostasis was observed, reflecting a classic example of oncogene addiction (4). The concept of "oncogenic addiction" was established by Weinstein in the early 2000s, and describes the reliance of tumor cells on continued expression of an oncogene or activated oncogenic signaling for the maintenance of tumorigenicity (5). As a result of this dependence, inactivation of the oncogene or oncoprotein impairs their growth and survival (6). Determinants for

targeted therapy development include demonstrated need of a specific molecule for carcinogenesis and tumor maintenance; therefore, the above results substantiated EGFRvIII as a rational therapeutic target.

Yet, despite sustained repression of EGFRvIII expression, some tumors eventually regained aggressive growth over the course of the experiment (4). The kinetics of this re-growth is analogous to the clinical situation where GBMs are initially sensitive to loss of receptor function, but eventually are able to overcome this loss. Therapeutic response to acute silencing of EGFRvIII (8 days DOX treatment) manifests as a significant reduction in proliferation and a slight increase in apoptosis (4). However, once the re-initiation of tumor growth in the absence of EGFRvIII expression occurs, both the decrease in proliferative capacity and the increase in apoptosis are reversed (4). In parallel with the latter phenotype, the expression of the anti-apoptotic molecule Bcl-xL is diminished at the protein level upon acute silencing of EGFRvIII, but is completely restored in the relapsed tumors (4). These data suggests that in our model and perhaps in GBM patients, overcoming the phenotypes associated with sensitivity to EGFR inhibition is required for escape.

In other solid tumors, response to EGFR inhibition also results in an accumulation of DNA damage in the form of double strand breaks (DSBs), and compromised DNA repair (7,8). For example, in breast cancer cells, treatment with erlotinib alone has been shown to lead to an accumulation of DSBs and suppression of the homology-directed recombinational repair (HDR) pathway (7). To this end, another study conducted in a variety of cancer types, demonstrated that erlotinib as part of a combinatorial therapy with arsenic trioxide leads to inhibition of EGFR-mediated DNA DSB repair and high levels of DNA damage (8). The phenotype in the

latter study was found to be secondary to erlotinib-mediated cell cycle arrest, whereas the results in the breast cancer study reflected a novel mechanism of action for erlotinib, as changes in the cell cycle were not observed (7,8).

DNA double strand breaks are the most lethal DNA lesions a cell can sustain because if left unrepaired, cell death, loss of genetic information and other genetic aberrations, such as chromosomal translocations can occur (9). In eukaryotes, two distinct repair pathways function to restore genomic integrity following DSB formation: non-homologous end-joining repair (NHEJ) and homology-directed recombinational repair (HDR) (10). The most common form of HDR is called homologous recombination (HR) and it requires an identical or very similar large stretch of DNA to serve as a template for repair, reflecting a high-fidelity repair process (11). NHEJ repair involves the rejoining of the broken ends of the fragmented DNA, and consequently, may be more error-prone (10). The decision to employ one repair pathway over the other is tightly regulated by the cell cycle; HR requires a sister chromatid to serve as the genetic template for repair, and so is restricted to the late S/G2 phases of the cell cycle (11). Alternatively, NHEJ is active in all stages of the cell cycle.

A number of reports linking unrepaired DNA DSBs and cancer development exist, but on the contrary, the hijacking of DNA repair pathways to eliminate DNA damage has also been implicated in therapeutic resistance (12-15). Additionally, as mentioned in the previous chapter, the mechanism of action for many cytotoxic chemotherapeutic agents is to inflict DNA damage in order to induce programmed cell death. Thus, DNA damage is causatively linked to tumorigenesis, is used for the treatment of cancer, but can also be circumvented by cancer cells for their survival. In mammalian cells, both EGFR and mutant EGFRvIII have been established as major regulators in the repair of radiation-induced DNA damage. DNA DSB repair capacity is enhanced by activated EGFR, and is reduced upon EGFR inhibition (16). MAPK signaling was established as the predominant downstream pathway that is engaged by EGFR to specifically facilitate its role in NHEJ repair (14). Previous reports established the involvement of EGFRvIII in radioresistance, and Mukherjee *et al.*, extended these findings by uncovering its ability to activate a key DNA repair enzyme, the DNA-dependent protein kinase, catalytic subunit (DNA-PKCs) (13,17). These results confirmed that similar to EGFR, NHEJ repair is also employed by EGFRvIII to circumvent radiation-induced DNA damage, since DNA-PKCs activity is blocked during HR (11).

Considering the above results and observations, blocking EGFRvIII in GBM should lead to an impairment in DNA repair, but this has not been directly tested, especially in a sans radiation setting. Thus, we sought to determine if a persistent DNA damage phenotype upon EGFR inhibition occurs in the context of GBM. Furthermore, we investigated if escape from dependence on EGFRvIII receptor function involves the exploitation of such damage or necessitates adaptive mechanisms to overcome it.
Results

3.2 Doxycycline diet sufficiently silences EGFRvIII and decreases in vivo tumor growth

In the previous study conducted by Mukasa *et al.*, doxycycline was administered via drinking water supplemented with sucrose, and although this is an effective method of regulation, superior approaches have been developed since this study. A number of doxycycline-containing diets are now commercially available with the following advantages over water delivery: protection from light via packaging, longer stability requiring only one change per week, and a reduced risk of dehydration, precluding the need for sucrose. In this study, we employed the diet delivery method to genetically silence EGFRvIII in vivo, and on the basis of the established response phenotypes observed upon acute silencing (8 days DOX treatment), we decided to assess the effectiveness of this approach at this time point.

Tumor-derived U373MG-tetO-EGFRvIII cells that are still dependent on EGFRvIII receptor function were engrafted subcutaneously into nude mice, and after tumors reached an approximate size of 500 mm³, half of the mice were switched to the doxycycline diet for 8 days. We observed a significant decrease in tumor growth in the group of mice switched to the doxycycline diet, reminiscent of the growth decrease observed with water delivery (**Figure 3.1A**). Additionally, we observed a complete silencing of EGFRvIII mRNA and protein expression (**Figure 3.1B**) upon doxycycline administration. These results demonstrated that the doxycycline diet approach is just as effective as the water delivery method and validated that 8 days of treatment is sufficient for in vivo blockade of EGFRvIII.



Figure 3.1. Doxycycline diet effectively silences EGFRvIII expression and glioma growth. (A) Growth of subcutaneous tumors from each population during the course of the experiment. Tumor Volume = ½ length x width^2 in unit of mm³ (+Doxycycline) indicates the start of doxycycline administration. n= 4 mice per group (B) qPCR analysis of EGFRvIII mRNA expression and Western blot analysis of EGFRvIII protein in tumor lysates for each condition.

3.3 Blockade of EGFRvIII alone increases the DNA damage burden in GBM

Although previous reports in other solid tumors have demonstrated that potent EGFR inhibition leads to impaired DNA DSB repair, this phenotype has not been documented in the context of GBM. Additionally, the previous findings suggest that the most appreciative impact on DNA damage and subsequent repair occurs in the presence of an exogenous physical agent, such as ionizing radiation (IR). In the case of the U373MG-tetO-EGFRvIII GBM model, mice were not exposed to any exogenous insult, but given the established role of both EGFR and EGFRvIII in the context of genome stability, we tested the hypothesis that in GBM, blockade of EGFRvIII alone yields a phenotype reflective of attenuated DNA DSB repair.

p53 binding protein 1 (53BP1) is an important regulator of chromatin based DSB signaling (11). It was first described as a binding partner for the tumor suppressor protein p53, from which its name was given (11,18). Functionally, 53BP1 acts as a molecular scaffold to recruit other DSB response proteins to damaged chromatin, and amplifies the activity of the ataxia telangiectasia mutated kinase (ATM) to promote checkpoint signaling in response to DNA damage (19). Additionally, it is known to specifically promote NHEJ repair over HR by antagonizing the resection extension step required for HR to take place (20).

In response to DNA damage in the form of DSBs, 53BP1 rapidly accumulates on chromatin surrounding the DSB sites and its localization persists until the DNA has been repaired (11). On the basis of these known functions for 53BP1, we conducted immunohistochemistry on the tumor samples that were generated in the in vivo experiment described in **section 3.2** to assess 53BP1 positive nuclei, as a marker for DNA DSBs. We observed a very low percentage of 53BP1 positive nuclei in the EGFRvIII-positive tumors (-DOX), but upon acute silencing of EGFRvIII, there was a significant increase in detectable 53BP1 nuclear staining (**Figure 3.2**). Given that proficient DNA repair is reflected by the clearance of 53BP1 nuclear foci (13), our results indicate that in the absence of EGFRvIII alone, DNA damage repair is attenuated in GBM. Additionally, the persistence in 53BP1 positive nuclei was observed in two different EGFRvIII-dependent populations, suggesting that this response phenotype may be a common phenomenon.

We next examined if this response phenotype also applies to the context of pharmacological inhibition of EGFRvIII, and used EGFRvIII expressing, PTEN wild-type primary *Ink4a/Arf^{-/-}* murine astrocytes to address this question. This model reflects cooperation between Ink4a/Arf deficiency and EGFRvIII signaling that promotes the formation of highly invasive intracranial tumors in mice that are similar to human GBMs (21). Additionally, these cells display sensitivity to the EGFR TKI's (22). To assess the DNA damage in this setting, we chose to examine the accumulation of phosphorylated γ H2A.X foci by immunofluorescence staining. ATM-mediated phosphorylation of γ H2A.X at serine 139 is the first step in cellular detection of DSBs, thus γ H2A.X foci formation has been established as a reliable indicator of DNA damage-induced DSBs (11).

In our control cells (DMSO-treated), we observed very few cells containing greater than eight γ H2A.X foci, which is the measure we used as our positive index (**Figure 3.3**). However, when treated with 5µM of the EGFR TKI gefitinib, we observed a 3.9-fold increase (25% versus 6.5%, p<0.001) in γ H2A.X foci formation at 48 hours post-treatment (**Figure 3.3**). These findings are consistent with the antitumor activity

89

of erlotinib previously described in breast cancer (7), suggesting that EGFR TKI's may act similarly to intercept DNA damage repair. Additionally, these data imply that the emerging phenotypes from our genetic model are clinically relevant and to our knowledge demonstrate for the first time, that EGFRvIII inhibition alone leads to an increase in the persistence of DNA DSBs in GBM.



Figure 3.2. An increase in the persistence of unrepaired DNA double strand breaks is associated with loss of EGFRvIII alone. (A) Representative immunohistochemical staining for the surrogate DNA double strand break marker, 53BP1, in resected tumors (B) Quantification of positively stained nuclei using microscopy cell counting software. EGFRvIII-Dep #1: (3.3-fold increase) and EGFRvIII-Dep #3 (13.2-fold increase) Data are means \pm SEM **p<0.01 **** p<0.0001



В





3.4 Increased ROS activity is observed with EGFRvIII inhibition

Based on the previous results, we next sought to determine the endogenous source of the DNA damage that we observe with blockade of EGFRvIII. It is known that endogenous DNA damage caused by reactive oxygen species (ROS) is the most frequently occurring damage (23). Reactive oxygen species in the most basic definition are chemically reactive chemical species, containing oxygen. Examples of ROS include superoxide, peroxides, hydroxyl radical, and singlet oxygen (24). Within most mammalian cells, ROS are primarily generated in the mitochondria as natural byproducts of oxidative metabolism (24). Premature electron leakage from the complexes I, II, and III of the electron transport chain can lead to the partial reduction of oxygen and the emergence of superoxide, which is the proximal ROS (23). In this manner, other highly reactive free radicals are derived, that are at leisure to attack DNA, proteins and lipids (23-25).

In the case of DNA, these chemical attacks result in lesions referred to as adducts. These DNA adducts can impair base pairing, block DNA replication and transcription, and promote the formation of DNA single-strand breaks (SSBs) (26). Moreover, when two SSBs emerge in close proximity or when the DNA-replication machinery encounters a SSB or other lesions, the very lethal DNA DSBs are formed. The potential toxic effects of ROS are usually prevented by inherent antioxidant defenses like superoxide dismutase, catalases and glutathione; however, when there is an imbalance between the production of ROS and the neutralizing ability of antioxidants, a state of oxidative stress ensues (27).

Oxidative stress can be defined as the cumulative accumulation of intracellular ROS, which can be both deleterious and beneficial to the cells (27). ROS participate in the intracellular signaling cascades that are commonly activated to maintain the oncogenic phenotype of cancer cells (25). For instance, during hypoxia, which refers to low oxygen conditions within the cell, mitochondrial ROS production is required for activation of the nuclear factor kappa B (NF κ -B) transcription factor to promote cell survival signaling (28). Alternatively, ROS can induce cellular senescence and apoptosis, and therefore function as anti-tumorigenic species. Indeed, the DNA-damaging effects imposed by many cancer therapeutics are mediated through the production of ROS (29). Furthermore, the overgeneration of ROS can cause sensitivity to hypoxia and subsequent cell death in cancer cells (28).

The established link between ROS activity and subsequent DNA damage, and the propensity of these species to antagonize tumorigenic potential, prompted us to interrogate if higher intracellular ROS levels, in the absence of EGFRvIII, was associated with the increase in DNA damage burden. We examined ROS activity by Flow Cytometry in our U373MG-tet-O-EGFRvIII populations in the presence and absence of doxycycline treatment, using two established ROS indicators. MitoSOX Red is a mitochondrial-localized, superoxide indicator and CellROX Orange is a cytosolic localized, general oxidative stress indicator. These two cell permeant dyes are non-fluorescent in their reduced states and become fluorescent upon oxidation by reactive oxygen species. We found that upon silencing of EGFRvIII, ROS activity is significantly elevated in both the cytosolic (8-fold increase) and mitochondrial (13-fold increase) cellular compartments in EGFRvIII-dependent populations (**Figure 3.4**). We next evaluated if the elevated ROS levels could be rescued by antioxidant treatment. We employed two antioxidants: EUK-134, a dual superoxide dismutase/catalase mimetic to scavenge ROS within mitochondria and Trolox, a vitamin E analogue to scavenge ROS within the cytosol. EGFRvIII-negative cells (+DOX) were treated with each antioxidant for 24 hours prior to labeling with the respective ROS indicator. Indeed, we observed a 50% decrease in mitochondrial ROS activity in the presence of EUK-134 (50 μ M) and a 45% decrease in cytosolic ROS activity in the presence of Trolox (100 μ M) (**Figure 3.4**). In the ROS activity experiments, doxycycline was administered for 6 days, which corresponds to 3 days of EGFRvIII silencing, and the 53BP1 in vivo experiment reflects 5 days of EGFRvIII silencing. Therefore, our data seems to be consistent with the predicted order of events, where the overgeneration of ROS would occur prior to the corresponding DNA damage.

Similar to our interest in the clinical relevance of the DNA damage phenotype seen in the genetic model, we also examined ROS activity upon pharmacological inhibition of EGFRvIII. EGFRvIII-positive U373MG glioma cells (-DOX) were treated with either DMSO or 5µM of lapatanib and *Ink4a/Arf^{-/-}* EGFRvIII murine astrocytes were treated with either DMSO or 5µM of gefitinib for 48 hours. Lapatanib was superior against EGFR tyrosine kinase activity (**Figure 3.5**). Cells were then labeled with the two ROS indicators. Mitochondrial ROS activity was significantly increased with lapatanib treatment in the U373MG glioma model (3-fold) and also with gefitinib treatment in the murine astrocytes (13-fold) (**Figure 3.5**). However, in both models, no significant difference in cytosolic ROS activity was observed (**Figure 3.5**). Taken

together, our data suggest that EGFRvIII regulates ROS activity levels in GBM, and in part, its receptor activity is required in the context of mitochondrial ROS.



Figure 3.4. EGFRvIII silencing leads to elevated ROS activity in GBM. (A) Representative Flow Cytometry plots depicting ROS activity as measured by the PE-positive population (Q3) for each condition. Blue=unlabeled cells; Red=labeled cells (B) Mean percentage of positive cells after background subtraction in three independent experiments. MitoSOX Red is a superoxide indicator; CellROX Orange is a general oxidative stress indicator. EUK-134 (50 μ M); Trolox (100 μ M). Data are means \pm SEM ****p<0.0001, ***p<0.001

Figure 3.5. Pharmacological inhibition of EGFRvIII is also associated with elevated ROS activity in GBM. Representative Western Blot analysis, confirming inhibition of EGFRvIII kinase activity by respective TKI's, Flow Cytometry plots depicting ROS activity as measured by the PE-positive population (Q3) in each cell compartment and the mean percentage of positive cells after background subtraction in (A) U373MG-tet-O-EGFRvIII cells and (B) *Ink4a/Arf^{/-}* EGFRvIII murine astrocytes. Lapatanib and Gefitinib treatment: 5μ M for 48 hours. C= DMSO, G=gefitinib, E=erlotinib, L=lapatanib Data are means \pm SEM **p<0.01



ROS-induced DNA damage is predominantly executed in the form of DNA oxidation, although DNA can undergo other DNA-damaging modifications, such as methylation, depurination and deamination (30). Out of the four nitrogen bases found in DNA, guanine is the most susceptible to DNA oxidation because of its low redox potential (31). The conversion of guanine to 8-oxoguanine is the most common transformation that occurs, arising from ROS-induced hydroxylation and cleavage reactions of the purine ring (32). The 8-oxoguanine DNA lesion is also one of the most characterized, making it a reliable biomarker to indicate the extent of oxidative DNA damage (30).

Based on this information, we next assessed oxidative stress-induced DNA damage by immunohistochemistry, using 8-oxoguanine as our index. This experiment was conducted on in vivo tissue from the U373MG-tetO-EGFRvIII model and also *Ink4a/Arf*^{-/-}EGFRvIII murine astrocytes in vivo samples from our previously published work, where in vivo growth in the presence or absence of gefitinib was measured (22). In the U373MG-tet-O-EGFRvIII model, we observed different intensity patterns for 8-oxoguanine staining between our EGFRvIII-positive populations, but overall, saw a significant increase in DNA oxidation upon EGFRvIII blockade (**Figure 3.6**). In the *Ink4a/Arf*^{-/-}EGFRvIII murine astrocytes, the results were similar, whereby an increase in the detection of 8-oxoguanine was associated with sensitivity to gefitinib treatment (**Figure 3.7**). These results demonstrate that in part, ROS is involved in promoting the DNA damaging effects seen with EGFRvIII inhibition.

Intensity Rubric



8-Oxoguanine





Figure 3.6 Increased DNA oxidation correlates with increased DNA damage burden upon EGFRvIII silencing. (A) Representative immunohistochemical staining for the biomarker of DNA oxidation, 8-oxoguanine, in tumors for each condition (B) Quantification of staining as determined by intensity rubric. Data reflect the individual cases per group and the respective extent of oxidative stress, represented as percentages. EGFRvIII-Dep #1: (-DOX) low-64%; medium-36%; high-0% vs (+DOX) low-27%; medium-73%; high-0% and EGFRvIII-Dep #3: (-DOX) low-47%; medium-53%; high-0% vs (+DOX) low-13%; medium-87%; high-0%



Figure 3.7. Increased DNA oxidation is associated with pharmacological inhibition of EGFRvIII. (A) Representative immunohistochemical staining for the 8-oxoguanine, in resected tumors for each condition (B) Quantification of staining as determined by intensity rubric (Figure 3.6). Data reflect the individual cases per group and the respective extent of oxidative stress, represented as percentages. Vehicle: low-66%; medium-34%; high-0% vs Gefitinib: low-3%; medium-90%; high-7%

3.5 EGFRvIII-independence is associated with overcoming therapeutic sensitivity

We noticed a pattern in our genetic model, whereby many of the phenotypes that were indicative of sensitivity to EGFRvIII inhibition, were reversed upon escape from EGFRvIII receptor function. Thus, we hypothesized that EGFRvIII-independent populations display a reversal in the percentage of DNA DSBs, ROS activity, and DNA oxidative damage. To test this hypothesis, we assessed EGFRvIII-independent relapse tumors and cultured cells for evidence of decreased ROS activity and oxidative DNA damage.

We first wanted to establish that a decrease in the persistence of unrepaired DNA DSBs was associated with overcoming EGFRvIII inhibition in our genetic model. Similar to the approach used for the EGFRvIII-dependent populations, we conducted immunohistochemical staining fro 53BP1 positive nuclei on tumor tissue from two EGFRvIII-independent populations (Escapers). The Escapers were chronically treated with the doxycycline diet, meaning that EGFRvIII is always silenced in this context. Just as we suspected, we observed a significant decrease in 53BP1 positive nuclei (~2-fold) in both populations (**Figure 3.8**).

Next, we examined ROS activity in cell lines that were generated from these Escaper tumors by Flow Cytometry using our established ROS indicators. Indeed, we observed a significant decrease in both mitochondrial (Esc 0: 2-fold; Esc 1: 3.6-fold) and cytosolic ROS activity (Esc 0: 2.2-fold; Esc1: 3.6-fold) (**Figure 3.9**). Consistent with these results, we also saw a decrease in the extent of oxidative DNA damage in Escaper tumor tissue stained for 8-oxoguanine (**Figure 3.10**).



53BP1





Figure 3.8. Overcoming EGFRvIII blockade is reflected by a significant reduction in the persistence of DNA DSBs. (A) Representative immunohistochemical staining for the surrogate DNA double strand break marker, 53BP1, in resected tumors (B) Quantification of positively stained nuclei using microscopy cell counting software. Esc 0: (45% reduction) and Esc 1 (48% reduction) Data are means \pm SEM **p<0.01



Figure 3.9. Escaper populations display significantly less ROS activity. (A) Representative Flow Cytometry plots depicting ROS activity as measured by the PE-positive population (Q3) for each condition. Blue=unlabeled cells; Red=labeled cells (B) Mean percentage of positive cells after background subtraction in three independent experiments. Data are means \pm SEM ****p<0.0001 ***p<0.001







Figure 3.10 A decrease in DNA oxidation accompanies diminished reliance on EGFRvIII (A) Representative 8-oxoguanine immunohistochemical staining in Escaper tumors (B) Quantification of staining as determined by intensity rubric (Figure 3.6). Data reflect the individual cases and their respective extent of oxidative stress, represented as percentages. EGFRvIII (+): low-54%; medium-46%; high-0% vs EGFRvIII (-): low-20%; medium-80%; high-0% vs EGFRvIII (-) Escapers: low-40%; medium-60%; high-0%

These data prompted us to examine the possible ROS levels in EGFR TKIresistant *Ink4a/Arf^{-/-}* EGFRvIII murine astrocytes that emerged following chronic treatment with gefitinib in vivo (22). Cell lines generated from these tumors are continually cultured in growth media containing 2μ M of gefitinib. For this experiment, we challenged the gefitinib resistant cells by treating with 5μ M of gefitinib for 48 hours prior to labeling with the ROS indicators. Similar to our Escaper populations, we observed a significant decrease in mitochondrial ROS activity in all four gefitinib resistant populations compared to the sensitive cells (control population) (**Figure 3.11**). These data further demonstrate that the phenotypes associated with genetic escape and pharmacological escape are analogous.

Lastly, we analyzed the extent of oxidative DNA damage in this model by immunohistochemistry. As suspected, we also saw a significant decrease in 8oxoguanine staining in this context (**Figure 3.12**). Taken together, these data suggest that escape from cytostasis involves recovery from DNA damage following oxidative stress. Furthermore, our findings demonstrate that another phenotype that is indicative of therapeutic sensitivity is also overturned in order for GBMs to overcome reliance on EGFRvIII.





Mitochondrial ROS (5 µM GEFITINIB)

Figure 3.11. Gefitinib resistance is associated with significantly less ROS activity. (A) Representative Flow Cytometry plots depicting mitochondrial ROS activity as measured by the PE-positive population (Q3) for each clone. Control population included as reference. Blue=unlabeled cells; Red=labeled cells; GR=gefitinib resistant (B) Mean percentage of positive cells after background subtraction. Data are means \pm SEM ****p<0.0001

Α







3.6 Summary

In summary, we found that EGFRvIII inhibition alone results in an increase in the persistence of DNA double strand breaks in vivo. This persistence was common to multiple "therapy-sensitive" populations and also correlated with an increase in reactive oxygen species (ROS) activity and oxidative DNA damage in vivo. On the contrary, populations that have escaped the need for EGFRvIII receptor function have significantly less DNA double strand breaks, decreased ROS activity and reduced oxidative DNA damage. Our results suggest that overcoming the effects of ROSinduced DNA damage is one mechanism utilized by GBMs to progressively escape blockade of EGFRvIII.

Increased ROS-induced DNA damage is often characterized as a tumorpromoting phenomenon, suggesting that tumors often capitalize on this damage for the maintenance of tumorigenicity. However, our findings imply that some tumors may be sensitive to ROS-induced DNA damage, necessitating potential adaptive mechanism(s) in order to support tumor survival and subsequent escape. It is known that the tight regulation of intracellular ROS levels is crucial for specific biological outcomes. For instance, constitutive low levels of ROS are necessary to support cell proliferation and differentiation (23). Yet as ROS levels increase, genetic adaptation is required for cell survival, including the transcriptional regulation of genes to antagonize potentially, deleterious oxidation. Based on this, we suspect that genetic changes that are unique to our EGFRvIII-independent populations give rise to their characteristic reduction in oxidative stress. In the next two chapters, we will discuss one of these genetic adaptations, the up-regulation of kelch domain containing 8A (KLHDC8A), and will establish its involvement in this adaptive response. Chapter 3 contains data that have not been published previously, and are currently being prepared for submission for publication. The work in chapter 3 was completed with the help of Jianhui Ma, M.D, PhD., and Antonia Boyer of the laboratory of Frank Furnari. This work was supported by the NIH R01 NS08939 to (F.B. Furnari).

3.7 Materials and Methods

Cell culture and reagents

U373MG-tetO-EGFRvIII-dependent were cultured in Dulbecco's Modified Eagle's Medium (DMEM) supplemented with 10% Tet-approved fetal bovine serum (FBS), Penicillin/streptomycin, G418 (2µg/ml), and puromycin (1µg/ml). Escaper cells were cultured in the same media, but with the addition of doxycycline (1µg/ml). Ink4a/arf^{-/-} EGFRvIII murine astrocytes were cultured in DMEM supplemented with 10% heat inactivated FBS, Penicillin/streptomycin, and puromycin (1µg/ml). Gefitinib-resistant Ink4a/arf^{-/-} EGFRvIII murine astrocytes were cultured in the same media, but with the addition of 2µM gefitinib. Gefitinib and lapatanib were purchased from LC Labs. Doxycycline was purchased from Clontech. Trolox, and EUK-134 were purchased from Sigma-Aldrich. Trolox was dissolved in DMSO (20 mg/ml) and EUK-134 (2 mg/ml) was dissolved in molecular grade water.

Xenograft Studies

All animal procedures were approved by the IACUC at the University of California, San Diego. A total of 5 x 10³ tumor-derived U373MG-tet-O-EGFRvIII cells were injected subcutaneously into the flanks of 6- to 8-week old female athymic mice. Mice were initially fed a regular diet and tumor volume was measured with calipers weekly beginning at day 14 post-injection. When average tumor volume reached approximately 500 mm³, half of the mice were switched to the doxycycline diet (200 mg/kg; Bio-Serv) for a total of 8 days. At the endpoint, tumors were resected from each group and appropriately prepared for the respective post-analysis experiments.

Quantitative real-time PCR

Subcutaneous tumors were resected from mice and then RNA was isolated from the tissue using the RNeasy Plus Kit (Qiagen). cDNA was synthesized using the RNA to cDNA EcoDry Premix (Clontech). qPCR was performed with a CFX96 Real-Time System (BioRad) using the following program: 95°C for 10 minutes, [95°C for 15 seconds; 58°C for 1 minute] x 40 cycles, 95°C for 1 min and 55°C for 1 min. EGFRvIII and actin gene expression were measured using SYBR Green (BioRad) with internal triplicate determinations for each sample.

Western blotting

Subcutaneous tumors were resected from mice and then tissue was lysed for protein harvesting using RIPA Buffer (50 mM Tris, pH 7.5, 150 mM NaCl, 1 mM EDTA pH 8.0, 1 mM NaF, 1% NP40, 0.1% SDS, 0.5% sodium deoxycholate and 1x Complete protease inhibitor [Roche]) and phosphatase inhibitors). Total protein was quantified using BCA Protein Assay Reagent (Thermo Scientific). Afterwards, 40µg of each protein sample was resolved on a 7.5% to 15% polyacrylamide gel. Proteins were transferred, blocked with 5% milk followed by primary antibody incubation overnight at 4°C. The following antibodies were used at specified concentrations for immunoblots: EGFR clone 13 (1:2500; BD Transduction Labs), EGFR Y1068 (1:1000, Cell Signaling), a-tubulin (1:500, Santa Cruz Technology), GAPDH (1:1000; Cell Signaling). Membranes were washed three times with TBS containing 0.1% Tween-20 and incubated with anti-mouse or anti-rabbit-HRP antibody (Sigma). Chemiluminescence was detected using Amersham ECL Western Blotting Detection Reagent (GE Healthcare) or SuperSignal West Pico Chemiluminescent Substrate

(Thermo Scientific). Images were obtained using ChemiDoc MP imaging system and Image Lab 4.1 software (Bio-Rad).

Immunohistochemistry

Tissue sections were cut from formalin-fixed paraffin-embedded xenograft tissue. Sections were deparaffanized, re-hydrated, and then stained with either anti-53BP1 (1:100; Cell Signaling) or anti-8-oxoguanine (1:1000; Rockland). Heat-induced antigen retrieval was performed for the 53BP1 staining using 10mM sodium citrate, pH 6.0. Antigen retrieval was omitted for 8-oxoguanine staining. Primary antibodies were incubated on slides overnight at 4°C in a humidified chamber. Secondary antibody incubation plus AEC (Scytek) for 53BP1 or plus DAB (Vectastain) for 8-oxoguanine detection, was done according to the manufacturer's instructions. Slides were dehydrated through a series of alcohol and xylene, then coverslips were mounted onto slides. Quantification of positively stained nuclei was conducted using microscopy cell counting software. Stain intensity scoring was determined by establishing an intensity rubric based on the samples displaying the highest, middle, and lowest staining across all samples.

<u>Immunofluorescence</u>

Ink4a/arf^{-/-} EGFRvIII murine astrocytes were seeded on coverslips and allowed to grow for 48 hours. At this point, cells were treated with 5µM of lapatanib for indicated time. Cells were fixed in 4% paraformaldehyde (pH 7.1) for 20 min at room temperature followed by a three PBS washes and permeabilization with 0.4% NP40 buffer for 15 min. After blocking with 5% heat inactivated donor calf serum in HBSS,

cells were incubated with pH2AX antibody (Millipore) followed by FITC-conjugated secondary antibody. Nuclei were counterstained with DAPI simultaneous to mounting coverslips on slides. Images were captured using a fluorescence microscope.

ROS Assays

A total of 50,000 to 80,000 cells were seeded into 6-well plates. After 48 hours, cells were labeled with either MitoSOX Red (superoxide indicator; Life Technologies) or CellROX Orange (general oxidative stress indicator; Life Technologies) according to the manufacturer's instructions. Samples were analyzed for cellular fluorescence on an LSR II (BD Biosciences) and data were analyzed with FlowJo software (Treestar). For the genetic silencing experiments, U373-tetO-EGFRvIII cells were treated with Doxycycline (1 μ g/ml) for 6 days prior to labeling and flow cytometry. For the pharmacological inhibition studies, Ink4a/arf^{-/-} EGFRvIII murine astrocytes were treated with 5 μ M of gefitinib for 48 hours prior to labeling and flow cytometry.

Statistical analysis

All data are presented with mean \pm SEM as determined by Prism software (GraphPad). Unpaired Student's t-test or ANOVA was used as appropriate. P values of 0.05 or less were considered significant for all experiments. Data are representative of results obtained in at least two to three independent experiments.

3.9. References

- Nishikawa, R., Ji, X. D., Harmon, R. C., Lazar, C. S., Gill, G. N., Cavenee, W. K., and Huang, H. J. (1994) A mutant epidermal growth factor receptor common in human glioma confers enhanced tumorigenicity. *Proceedings of the National Academy of Sciences of the United States of America* **91**, 7727-7731
- Huang, H. S., Nagane, M., Klingbeil, C. K., Lin, H., Nishikawa, R., Ji, X. D., Huang, C. M., Gill, G. N., Wiley, H. S., and Cavenee, W. K. (1997) The enhanced tumorigenic activity of a mutant epidermal growth factor receptor common in human cancers is mediated by threshold levels of constitutive tyrosine phosphorylation and unattenuated signaling. *The Journal of biological chemistry* 272, 2927-2935
- 3. Nagane, M., Lin, H., Cavenee, W. K., and Huang, H. J. (2001) Aberrant receptor signaling in human malignant gliomas: mechanisms and therapeutic implications. *Cancer letters* **162 Suppl**, S17-S21
- 4. Mukasa, A., Wykosky, J., Ligon, K. L., Chin, L., Cavenee, W. K., and Furnari, F. (2010) Mutant EGFR is required for maintenance of glioma growth in vivo, and its ablation leads to escape from receptor dependence. *Proceedings of the National Academy of Sciences of the United States of America* **107**, 2616-2621
- 5. Weinstein, I. B. (2000) Disorders in cell circuitry during multistage carcinogenesis: the role of homeostasis. *Carcinogenesis* **21**, 857-864
- 6. Weinstein, I. B., and Joe, A. (2008) Oncogene addiction. *Cancer research* **68**, 3077-3080; discussion 3080
- 7. Li, L., Wang, H., Yang, E. S., Arteaga, C. L., and Xia, F. (2008) Erlotinib attenuates homologous recombinational repair of chromosomal breaks in human breast cancer cells. *Cancer research* **68**, 9141-9146
- 8. Kryeziu, K., Jungwirth, U., Hoda, M. A., Ferk, F., Knasmuller, S., Karnthaler-Benbakka, C., Kowol, C. R., Berger, W., and Heffeter, P. (2013) Synergistic anticancer activity of arsenic trioxide with erlotinib is based on inhibition of EGFR-mediated DNA double-strand break repair. *Molecular cancer therapeutics* **12**, 1073-1084

- 9. Downs, J. A. (2007) Chromatin structure and DNA double-strand break responses in cancer progression and therapy. *Oncogene* **26**, 7765-7772
- Lieber, M. R. (2010) The mechanism of double-strand DNA break repair by the nonhomologous DNA end-joining pathway. *Annual review of biochemistry* 79, 181-211
- 11. Panier, S., and Boulton, S. J. (2014) Double-strand break repair: 53BP1 comes into focus. *Nature reviews. Molecular cell biology* **15**, 7-18
- Nitta, M., Kozono, D., Kennedy, R., Stommel, J., Ng, K., Zinn, P. O., Kushwaha, D., Kesari, S., Inda, M. M., Wykosky, J., Furnari, F., Hoadley, K. A., Chin, L., DePinho, R. A., Cavenee, W. K., D'Andrea, A., and Chen, C. C. (2010) Targeting EGFR induced oxidative stress by PARP1 inhibition in glioblastoma therapy. *PloS one* 5, e10767
- Mukherjee, B., McEllin, B., Camacho, C. V., Tomimatsu, N., Sirasanagandala, S., Nannepaga, S., Hatanpaa, K. J., Mickey, B., Madden, C., Maher, E., Boothman, D. A., Furnari, F., Cavenee, W. K., Bachoo, R. M., and Burma, S. (2009) EGFRvIII and DNA double-strand break repair: a molecular mechanism for radioresistance in glioblastoma. *Cancer research* 69, 4252-4259
- 14. Kriegs, M., Kasten-Pisula, U., Rieckmann, T., Holst, K., Saker, J., Dahm-Daphi, J., and Dikomey, E. (2010) The epidermal growth factor receptor modulates DNA double-strand break repair by regulating non-homologous end-joining. *DNA repair* **9**, 889-897
- 15. Kastan, M. B. (2008) DNA damage responses: mechanisms and roles in human disease: 2007 G.H.A. Clowes Memorial Award Lecture. *Molecular cancer research : MCR* 6, 517-524
- 16. Mukherjee, B., Choy, H., Nirodi, C., and Burma, S. (2010) Targeting nonhomologous end-joining through epidermal growth factor receptor inhibition: rationale and strategies for radiosensitization. *Seminars in radiation oncology* **20**, 250-257
- 17. Nagane, M., Levitzki, A., Gazit, A., Cavenee, W. K., and Huang, H. J. (1998) Drug resistance of human glioblastoma cells conferred by a tumor-specific mutant epidermal growth factor receptor through modulation of Bcl-XL and caspase-3-like proteases. *Proceedings of the National Academy of Sciences* of the United States of America **95**, 5724-5729

- 18. Iwabuchi, K., Bartel, P. L., Li, B., Marraccino, R., and Fields, S. (1994) Two cellular proteins that bind to wild-type but not mutant p53. *Proceedings of the National Academy of Sciences of the United States of America* **91**, 6098-6102
- 19. Wang, B., Matsuoka, S., Carpenter, P. B., and Elledge, S. J. (2002) 53BP1, a mediator of the DNA damage checkpoint. *Science* **298**, 1435-1438
- 20. Liu, T., and Huang, J. (2016) DNA End Resection: Facts and Mechanisms. *Genomics, proteomics & bioinformatics* **14**, 126-130
- Bachoo, R. M., Maher, E. A., Ligon, K. L., Sharpless, N. E., Chan, S. S., You, M. J., Tang, Y., DeFrances, J., Stover, E., Weissleder, R., Rowitch, D. H., Louis, D. N., and DePinho, R. A. (2002) Epidermal growth factor receptor and Ink4a/Arf: convergent mechanisms governing terminal differentiation and transformation along the neural stem cell to astrocyte axis. *Cancer cell* 1, 269-277
- 22. Wykosky, J., Hu, J., Gomez, G. G., Taylor, T., Villa, G. R., Pizzo, D., VandenBerg, S. R., Thorne, A. H., Chen, C. C., Mischel, P. S., Gonias, S. L., Cavenee, W. K., and Furnari, F. B. (2015) A urokinase receptor-Bim signaling axis emerges during EGFR inhibitor resistance in mutant EGFR glioblastoma. *Cancer research* **75**, 394-404
- 23. Hamanaka, R. B., and Chandel, N. S. (2010) Mitochondrial reactive oxygen species regulate cellular signaling and dictate biological outcomes. *Trends in biochemical sciences* **35**, 505-513
- 24. Sena, L. A., and Chandel, N. S. (2012) Physiological roles of mitochondrial reactive oxygen species. *Molecular cell* **48**, 158-167
- Weinberg, F., Hamanaka, R., Wheaton, W. W., Weinberg, S., Joseph, J., Lopez, M., Kalyanaraman, B., Mutlu, G. M., Budinger, G. R., and Chandel, N. S. (2010) Mitochondrial metabolism and ROS generation are essential for Kras-mediated tumorigenicity. *Proceedings of the National Academy of Sciences of the United States of America* **107**, 8788-8793
- 26. Jackson, S. P., and Bartek, J. (2009) The DNA-damage response in human biology and disease. *Nature* **461**, 1071-1078

- 28. Lluis, J. M., Buricchi, F., Chiarugi, P., Morales, A., and Fernandez-Checa, J. C. (2007) Dual role of mitochondrial reactive oxygen species in hypoxia signaling: activation of nuclear factor-{kappa}B via c-SRC and oxidant-dependent cell death. *Cancer research* **67**, 7368-7377
- 29. Alan Mitteer, R., Wang, Y., Shah, J., Gordon, S., Fager, M., Butter, P. P., Jun Kim, H., Guardiola-Salmeron, C., Carabe-Fernandez, A., and Fan, Y. (2015) Proton beam radiation induces DNA damage and cell apoptosis in glioma stem cells through reactive oxygen species. *Scientific reports* **5**, 13961
- 30. Wiseman, H., and Halliwell, B. (1996) Damage to DNA by reactive oxygen and nitrogen species: role in inflammatory disease and progression to cancer. *The Biochemical journal* **313 (Pt 1)**, 17-29
- 31. Crespo-Hernandez, C. E., Close, D. M., Gorb, L., and Leszczynski, J. (2007) Determination of redox potentials for the Watson-Crick base pairs, DNA nucleosides, and relevant nucleoside analogues. *The journal of physical chemistry. B* **111**, 5386-5395
- 32. Turk, P. W., Laayoun, A., Smith, S. S., and Weitzman, S. A. (1995) DNA adduct 8-hydroxyl-2'-deoxyguanosine (8-hydroxyguanine) affects function of human DNA methyltransferase. *Carcinogenesis* **16**, 1253-1255

Chapter 4

Introduction to the Kelch Repeat Superfamily

4.1 Kelch Repeat Superfamily

KLHDC8A is a member of the Kelch repeat superfamily, which is one of the largest evolutionarily conserved gene families, where the presence of the kelch motif that defines this family is found from *Drosophila melanogaster* to *Homo sapiens* (1). The founding member of this family, *kelch*, was first described in *Drosophila melanogaster*, where female homozygous *kelch* mutants produced very small, cup-shaped eggs for which the gene was named; "kelch" in the German language means "cup" (2). Future molecular cloning experiments together with the use of bioinformatics afforded the discovery of the kelch motif as a repeated element in the sequence of the *Drosophila* kelch open reading frame 1 (ORF1) protein (3,4).

The kelch motif is typically 44-56 amino acids in length, occurring in most members as five to seven tandem repeats that collectively fold into a β -propeller tertiary structure (5,6). The crystal structure of galactose oxidase, a converter of alcohol to aldehyde, from the fungus *Dactylium dendroides* was the first of the Kelch superfamily to be solved, becoming the prototype for subsequent analyses used to identify other Kelch superfamily members and to predict their tertiary structures (4,7). On the basis of this crystal structure, each repeat forms a twisted four-stranded antiparallel β -sheet that corresponds to a single blade of the propeller radially arranged around a central axis, and either an amino-terminal (N-terminal) or a carboxyl-terminal (C-terminal) strand closure mechanism links the first and last blades to complete the propeller (5-7). Intra- and inter-blade loops protrude above, below, or adjacent to the β -sheets, constituting the accessible regions of this structure and contribute to the characteristic binding properties of each β -propeller (6).
Multiple sequence alignment and pattern analysis uncovered a group of highly conserved residues now regarded as the consensus sequence of the kelch motif (4). These key residues include a series of four hydrophobic amino acids followed by a double glycine element (GG), and two characteristically spaced aromatic residues, tyrosine (Y) and tryptophan (W) (5,6,8). Aside from these eight residues, the sequence similarity between individual kelch motifs within a given protein is surprisingly low, rendering these residues indispensable to distinguish kelch-containing proteins from those of the similarly structured WD-40 repeat family, which also form β -propellers (4,6).

Within the human genome, the encoding sequences for kelch-repeat proteins are located on all chromosomes except chromosome 21 and the Y chromosome (5). The growing list of Kelch superfamily members currently includes 66 genes, of which sixty-three encode proteins, and three are non-coding (8). Each member is classified into one of three subfamilies: KLHL, KBTBD, and KLHDC, defined by the number and types of protein domains present (5,6,8).

4.2 Kelch Subfamilies and Domain Architecture

The classic molecular architecture of a kelch protein includes one N-terminal BTB/POZ domain, one BACK domain and a series of C-terminal kelch repeats (5,6,8). Members of the largest subfamily, KLHL, contain all of these domains, while the KBTBD members usually lack the BACK domain (5,6,8). Most KLHDC family members lack both the BTB/POZ and BACK domains and consists almost entirely of kelch repeats alone or with other domains such as transmembrane, Glycine rich, Lish or CTLH (CRA C-terminal to Lish) domains (8).

The KLHL subfamily is conserved throughout evolution and to date, 42 KLHL genes have been identified in H. sapiens and classified by the HUGO Gene Nomenclature Committee (HGNC) (1,8). The characterization of the BTB/POZ domain found at the N-terminal end of KLHL and KBTBD proteins, but not unique to the Kelch superfamily, was first explored in *Drosophila*. Its name is derived from a highly conserved, 120 amino acid motif that is shared between three Drosophila genes: bric à brac (bab), tramtrack (ttk), and Broad-Complex (BR-C) (BTB) (9). POZ reflects the concurrent name given to the related motif found in poxvirus that resembled portions of zinc finger proteins (1,10). The structure of this domain is compact and globular, consisting of a cluster of α -helices flanked by short β -sheet structures (11-13). It primarily participates in protein-protein interactions, and though the secondary structures of BTB/POZ proteins are highly similar, the primary sequences are not (8,11,14). This sequence variability permits differential proteinprotein interactions, and consequently, diverse functional abilities. Within the Kelch superfamily, the BTB domain typically functions as a substrate adaptor that is required for the formation of Cullin-dependent E3 ubiquitin ligase complexes (8,11).

Other known functions include transcriptional regulation, cytoskeletal organization, and ion conductance, yet given the large number of BTB/POZ proteins that have been identified, the functions of many proteins are still to be determined (6,8,11,12).

The BACK (for BTB and C-terminal Kelch) domain, also found in KLHL subfamily members, is the most conserved domain within the Kelch Superfamily (15). It is ~130 amino acids in length and typically occurs as a single copy between the N-terminal BTB/POZ domain and the C-terminal Kelch repeats (8,15). Its secondary structure is likely to be completely α-helical and its central core, hydrophobic, due to the many non-polar residues that make up its conserved features: an N-terminal Asn-Cys-Leu-Gly-Ile motif, a Val-Arg-[Leu/Met/Phe]-Pro-Leu-Leu motif, two arginine residues, four glutamic acids, and several hydrophobic residues (15). Similar to the BTB/POZ domain, the BACK domain is also found in non-Kelch proteins. However, the majority of BACK domains occur in BTB-Kelch proteins and are predominant among vertebrate species (15).

The predicted secondary structure of a classical BTB-BACK-Kelch (BBK) protein suggests possible structural and functional similarities with the multi-subunit E3 enzymes termed Cullin-Skp1-Cullin-F box (SCF) complexes (15). On the basis of the crystal structure of the Skp1-Cdc4 complex, the BTB domain of a BBK protein would correspond to the N-terminal half of Skp1, which interacts with the respective Cullin protein (15,16). The BACK domain would reflect both the C-terminal half of Skp1 and the subsequent N-terminal region of the F-box protein Cdc4, which are α -helical in structure (15,16). Lastly, Cdc4 recruits target substrates to the SCF via its C-terminal WD40-domain, a structural equivalent to the Kelch domain, which serves as the substrate recognition component of the complex (15,16). Thus, BBK proteins

are potentially able to accomplish the functions of Skp1 and F box proteins as a single polypeptide. The BACK domain is further speculated to mediate substrate orientation in Cullin-based E3 ubiquitin ligase complexes, but the true function for the BACK domain has not been defined (15). Nevertheless, missense mutations within this domain of the BBK proteins, KLHL40 and KLHL41, are associated with the onset of severe nemaline myopathy (a skeletal muscle disorder), substantiating its functional importance in blocking disease (17,18).

4.3 Kelch proteins and biological outcomes

The tertiary structure of the Kelch domain additionally classifies Kelch proteins as members of the β -propeller superfamily of proteins, and this β -propeller structure affords the fulfillment of a number of biological roles. Additionally, many kelch proteins can share a significant amount of sequence identity; yet can differ in subcellular localization and function within the same cell type (6). This highlights the versatility of this domain in the many aspects of cell function.

Several kelch proteins are actin-binding proteins (e.g., Mayven, α -scruin, IPP, AFKin, MRP2) and function in cytoskeletal organization and nervous system development via their β -propeller structures (19,20). Others facilitate germ cell maturation and cell morphology by promoting differentiation, cell spreading and cell polarity (6). In the context of the immune system, the kelch protein RAG-2 is involved in V(D)J recombination of immunoglobulin and T-cell receptors genes, and the glycoprotein attractin functions as a mediator of monocyte spreading (6). Specifically, activated T-cells secrete attractin into the extracellular environment, where it stimulates cell adhesion and migration that are required for the development of T-cell clusters and cell-mediated immunity (21).

Additional biological processes, in which members of the Kelch superfamily are implicated in, include signal transduction, cell division, RNA processing, vesicular trafficking and chemotaxis (6,20,22,23). The critical nature of these molecular functions necessitates tight regulation to impede the onset of disease, and consequently, the aberrant expression and regulation of many kelch repeat proteins are associated with Mendelian disorders and cancer (1). In fact, the elevated secretion of attractin into cerebrospinal fluid supports glioma cell migration and tumor progression (24). Additional documented cases, whereby the normal function of kelch proteins is usurped to benefit cancer cells, will be discussed below.

4.4 Kelch proteins and cancer

The nuclear-restricted protein/brain (NRP/B or KLHL37), which is also known as ectodermal-neural cortex (ENC-1) gene, is a BTB-Kelch protein that functions as a component of the nuclear matrix (22,25). The nuclear matrix is a three-dimensional insoluble framework of the nucleus, and is involved in modulating gene expression, the cell cycle, and the integrity of the nuclear structure (26). Changes to the nuclear architecture are associated with differentiation and cellular transformation, and consequently, these alterations are prevalent in cancer.

In normal cells, the expression of NRP/B is restricted to neurons, and this specificity has established it as an early marker for neurons during development of the central nervous system (25). However, elevated expression of NRP/B is detected in the non-neuronal cells that make up GBMs and astrocytomas (22). Additionally, NRP/B is normally expressed in the nucleus of neurons, but within brain tumors it is expressed in the cytoplasm (22,26). Thus, a shift in both subcellular localization and cell type-specific expression is associated with the oncogenic version of this protein.

Several brain tumor-specific mutant forms of NRP/B have been identified, with mutations predominantly occurring in the Kelch domain (22). These mutations augment MAPK/ERK-mediated cell proliferation, confer resistance to cisplatininduced apoptosis, and are sufficient to promote the same tumorigenic properties in other cancer types (22). The NRP/B promoter is activated by p53; and so perhaps the mutant forms of NRP/B are not functionally competent upon p53 recognition and binding, which leads to the cisplatin-resistant phenotype. Additionally, the kelch domain mutations in NRP/B reduce its binding affinity for actin, which would dramatically affect its contribution to the nuclear matrix architecture (22). Wild-type NRP/B associates with the retinoblastoma (pRb) tumor suppressor protein to promote neuronal differentiation, and this interaction leads to hypophosphorylation of pRb, which induces G1 cell cycle arrest (25). Defects in the pRb pathway are often essential for the development of GBMs, occurring in 70-80% of GBM cases (27). This percentage may in part be due to NRP/B mutations. Hence, the normal functions of NRP/B would suggest that it is a tumor suppressor: however, mutant NRP/B actually contributes to tumorigenesis by promoting cell proliferation, survival and nuclear structure alterations.

The Kelch-like ECH-associated protein 1 (Keap1 or KLHL19) is a six-bladed propeller BTB-Kelch protein that is localized to the cytoplasm (5). It primarily functions as a substrate recognition module within the Cullin3-dependent E3 ubiquitin ligase complex, and through this role, targets the nuclear factor (erythroid 2-related) factor 2 (Nrf2) for proteasome-dependent degradation under normal cellular conditions (28). Specifically, Keap1 negatively controls the transactivation function of Nrf2 by sequestering it in the cytoplasm, but in the presence of stress stimuli, Nrf2 is released from this interaction in its active form and translocates to the nucleus (29). There, it induces the expression of phase II detoxification and oxidative stress response genes, such as heme oxygenase 1 and NADPH dehydrogenase quinone, and onecarbon metabolism enzymes that drive the synthesis of the antioxidant, glutathione (30).

The expression of these genes yields a cellular defense system that neutralizes reactive oxygen species (ROS) and protects against other toxic insults (30). Oxidative/electrophilic stress sensing by KEAP1, initiates this cascade, wherein its cysteine residues act as biosensors for ROS or xenobiotic molecules (31). Upon cysteine modification, Keap1 undergoes a conformational change that releases bound Nrf2. Such an efficient mechanism of ROS detoxification would present a selective advantage for cancer cells, which are constantly in a pro-oxidant state oxidant, as a consequence of the genetic, metabolic and microenvironmentassociated alterations that characterize them (32). Not surprising, the protection function of Nrf2 is often appropriated by cancer cells, to create a pro-survival environment that supports growth and therapeutic resistance (33). Additionally, lossof-function mutations in Keap1 are also linked to cancer (31).

The majority of the alterations to the Nrf2-Keap1 axis favor a hyperfunctioning Nrf2 protein. Constitutive activation of Nrf2 is associated with poor prognosis in a variety of cancers (28). Additionally, Nrf2-target genes are associated with cancer cell proliferation through the induction of a glucose flux and the generation of purines, which are important for nucleic acid formation and accelerated growth (28). Nrf2-mediated glutathione production is also critical for cell proliferation (34). Furthermore, elevated Nrf2 confers inherent and acquired chemoresistance in many cancer types, and protects against ionizing radiation by decreasing protein oxidative damage (28,35).

A few somatic mutations in Nrf2 have been identified in cancer, which impair Keap1 recognition and binding, resulting in Nrf2 stability, increased nuclear localization and the constitutive activation of cytoprotective genes (28). Keap1 mutations are more frequently encountered, and predominantly occur within the Kelch domain, which is required for binding to Nrf2 (30,31). Epigenetic silencing of Keap1 via hypermethylation of its promoter is also associated with a growth advantage for many cancers, and this mechanism of Keap1 suppression is also associated with poor prognosis in lung and malignant glioma (36,37). MicroRNA-mediated gene suppression is also employed by cancer cells to eliminate Keap1, and specifically miR-200A acts as a negative regulator of Keap1 stability (38). Thus, the intended tumor suppressor function of the Keap1 is often antagonized by a variety of mechanisms that support cancer cell expansion.

KLHDC8B is one of the only two, seven kelch repeat only-containing proteins within the human genome; the other is KLHDC8A. It is only expressed in mitotic cells and functions as a midbody protein that is essential for proper cytokinesis (39,40). This protein has been studied extensively in the context of classical Hodgkin's lymphoma (cHL), where its deficiency is linked to this disease (40). Classical Hodgkin's lymphoma is unique, in that the cancerous Reed-Sternberg (RS) cells are the minor population within the tumor mass (1 in 100) (39). However, these cells promote the creation of the growing mass by recruiting benign, reactive inflammatory cells, such as eosinophils, fibroblasts, and lymphocytes (40). This disease is primarily associated with Epstein-Barr Virus (EBV) exposure (40% of cases), but other genetic features have been implicated as risk factors (39).

Indeed, the familial risk for cHL is very high, and a screen of several familial cases of cHL identified a common reciprocal translocation between chromosomes 2 and 3 (41). The specific breakpoint mapped to 3p21.31, a region where frequent somatic cytogenetic rearrangements had already been observed for cHL (41). This locus is also where KLHDC8B resides, and in affected individuals, its upstream regulatory elements and exon 1 are eliminated, as a result of the translocation (41). Additionally, a portion of its 5' UTR fused with an intergenic region on chromosome 2q11.2 and a single nucleotide polymorphism (SNP) at a phylogenetically conserved

position within its 5'UTR were also identified in some familial cases (41). The functional consequence of the fusion alteration is abrogated transcription, and the SNP causes attenuated translation of the protein (40,41).

The midbody is a small intracellular structure that functions as the final point of contact between dividing daughter cells before they undergo separation via cytokinesis. Studies that recapitulated the cellular context, whereby KLHDC8B function is disrupted, demonstrated that it protects against a multitude of mitotic errors: multinucleation, aberrant mitoses, centrosomal amplification, micronuclei production, delayed or failed abscission, aneuploidy, multipolar mitotic figures, and asymmetric segregation of daughter cells that lead to the formation of anuclear daughter cells (23). All of these errors are well documented in the pathogenesis of Hodgkin's lymphoma, substantiating the centrality of this protein in maintaining genetic stability. Furthermore, its primary function as a tumor suppressor adheres to the functional theme that appears to be common to this family of proteins.

4.6 KLHDC8A

In 2010, Mukasa *et al.* established the first substantial link between KLHDC8A and tumorigenesis. Gene expression profile analysis conducted on relapse GBM tumors (Escapers) that emerged from a chronic EGFRvIII-negative environment revealed that KLHDC8A was significantly enriched in these populations (42). The pattern of expression was particularly striking due to its absence in EGFRvIII-dependent populations. Furthermore, this study showed that when KLHDC8A is silenced in Escapers, their in vivo growth capacity is significantly impaired, suggesting that this protein is essential in tumor recurrence (42).

KLHDC8A is predicted to be a 7-bladed kelch repeat only-containing protein, based on its primary amino acid sequence (**Figure 4.1A**). Yet, aside from this information, very little is known about the mechanism of its function and the molecular consequences of its dysfunction. Intra-sequence alignment of each repeat highlighted putative residues that may be important for its function and subcellular localization, such as surface residues within the loop regions that can be phosphorylated to initiate signaling cascades or function as redox sensors (**Figure 4.1B**). Additionally, although KLHDC8A is presumably a cytoplasmic protein, a predicted nuclear localization sequence (NLS) exists within its sequence, suggesting that it may be able to shuttle between cellular compartments (**Figure 4.1B**).

Based on the roles of the other Kelch proteins described in this chapter, it is likely that KLHDC8A normally functions as a tumor suppressor; however, none of the functions established for the other members have been attributed to KLHDC8A. Moreover, if its demonstrated involvement in GBM so far is any indication of a "transformed" role, then like the others, it may be a victim of cancer hijacking. In the

134

next chapter, we will more clearly define a role for KLHDC8A in the context of therapeutic escape in GBM.



Figure 4.1. KLHDC8A is a 7-bladed β -propeller only containing protein. Schematic diagrams based on the predicted structure of KLHDC8A. (A) Top: linear representation of KLHDC8A depicting the amino acid numbers within each kelch repeat. Bottom: predicted crystal structure using the Phyre2 web portal for protein modeling, prediction and analysis (43), where each blade corresponds to the kelch repeat of the same color in the top linear representation. (B) Intra-sequence alignment for each of the kelch repeats, displaying the specific amino acids that reside in the loop and strand regions and the alignment for the conserved residues that make up the kelch consensus sequence. Note that the C-terminal and N-terminal ends combine to form the first blade. Y-tyrosine (phosphorylation); C-cysteine (redox sensing); KKRR-potential NLS

4.7. References

- 1. Dhanoa, B. S., Cogliati, T., Satish, A. G., Bruford, E. A., and Friedman, J. S. (2013) Update on the Kelch-like (KLHL) gene family. *Human genomics* **7**, 13
- 2. Schupbach, T., and Wieschaus, E. (1991) Female sterile mutations on the second chromosome of Drosophila melanogaster. II. Mutations blocking oogenesis or altering egg morphology. *Genetics* **129**, 1119-1136
- 3. Xue, F., and Cooley, L. (1993) kelch encodes a component of intercellular bridges in Drosophila egg chambers. *Cell* **72**, 681-693
- 4. Bork, P., and Doolittle, R. F. (1994) Drosophila kelch motif is derived from a common enzyme fold. *Journal of molecular biology* **236**, 1277-1282
- 5. Prag, S., and Adams, J. C. (2003) Molecular phylogeny of the kelch-repeat superfamily reveals an expansion of BTB/kelch proteins in animals. *BMC Bioinformatics* **4**, 42
- 6. Adams, J., Kelso, R., and Cooley, L. (2000) The kelch repeat superfamily of proteins: propellers of cell function. *Trends Cell Biol* **10**, 17-24
- 7. Ito, N., Phillips, S. E., Yadav, K. D., and Knowles, P. F. (1994) Crystal structure of a free radical enzyme, galactose oxidase. *Journal of molecular biology* **238**, 794-814
- 8. Gupta, V. A., and Beggs, A. H. (2014) Kelch proteins: emerging roles in skeletal muscle development and diseases. *Skeletal muscle* **4**, 11
- Zollman, S., Godt, D., Prive, G. G., Couderc, J. L., and Laski, F. A. (1994) The BTB domain, found primarily in zinc finger proteins, defines an evolutionarily conserved family that includes several developmentally regulated genes in Drosophila. *Proceedings of the National Academy of Sciences of the United States of America* **91**, 10717-10721
- 10. Bardwell, V. J., and Treisman, R. (1994) The POZ domain: a conserved protein-protein interaction motif. *Genes & development* **8**, 1664-1677

- 11. Perez-Torrado, R., Yamada, D., and Defossez, P. A. (2006) Born to bind: the BTB protein-protein interaction domain. *BioEssays : news and reviews in molecular, cellular and developmental biology* **28**, 1194-1202
- Albagli, O., Dhordain, P., Deweindt, C., Lecocq, G., and Leprince, D. (1995) The BTB/POZ domain: a new protein-protein interaction motif common to DNA- and actin-binding proteins. *Cell growth & differentiation : the molecular biology journal of the American Association for Cancer Research* 6, 1193-1198
- 13. Ahmad, K. F., Engel, C. K., and Prive, G. G. (1998) Crystal structure of the BTB domain from PLZF. *Proceedings of the National Academy of Sciences of the United States of America* **95**, 12123-12128
- Stogios, P. J., Downs, G. S., Jauhal, J. J., Nandra, S. K., and Prive, G. G. (2005) Sequence and structural analysis of BTB domain proteins. *Genome biology* 6, R82
- 15. Stogios, P. J., and Prive, G. G. (2004) The BACK domain in BTB-kelch proteins. *Trends in biochemical sciences* **29**, 634-637
- 16. Orlicky, S., Tang, X., Willems, A., Tyers, M., and Sicheri, F. (2003) Structural basis for phosphodependent substrate selection and orientation by the SCFCdc4 ubiquitin ligase. *Cell* **112**, 243-256
- Ravenscroft, G., Miyatake, S., Lehtokari, V. L., Todd, E. J., Vornanen, P., Yau, K. S., Hayashi, Y. K., Miyake, N., Tsurusaki, Y., Doi, H., Saitsu, H., Osaka, H., Yamashita, S., Ohya, T., Sakamoto, Y., Koshimizu, E., Imamura, S., Yamashita, M., Ogata, K., Shiina, M., Bryson-Richardson, R. J., Vaz, R., Ceyhan, O., Brownstein, C. A., Swanson, L. C., Monnot, S., Romero, N. B., Amthor, H., Kresoje, N., Sivadorai, P., Kiraly-Borri, C., Haliloglu, G., Talim, B., Orhan, D., Kale, G., Charles, A. K., Fabian, V. A., Davis, M. R., Lammens, M., Sewry, C. A., Manzur, A., Muntoni, F., Clarke, N. F., North, K. N., Bertini, E., Nevo, Y., Willichowski, E., Silberg, I. E., Topaloglu, H., Beggs, A. H., Allcock, R. J., Nishino, I., Wallgren-Pettersson, C., Matsumoto, N., and Laing, N. G. (2013) Mutations in KLHL40 are a frequent cause of severe autosomalrecessive nemaline myopathy. *American journal of human genetics* 93, 6-18
- 18. Gupta, V. A., Ravenscroft, G., Shaheen, R., Todd, E. J., Swanson, L. C., Shiina, M., Ogata, K., Hsu, C., Clarke, N. F., Darras, B. T., Farrar, M. A., Hashem, A., Manton, N. D., Muntoni, F., North, K. N., Sandaradura, S. A.,

Nishino, I., Hayashi, Y. K., Sewry, C. A., Thompson, E. M., Yau, K. S., Brownstein, C. A., Yu, T. W., Allcock, R. J., Davis, M. R., Wallgren-Pettersson, C., Matsumoto, N., Alkuraya, F. S., Laing, N. G., and Beggs, A. H. (2013) Identification of KLHL41 Mutations Implicates BTB-Kelch-Mediated Ubiquitination as an Alternate Pathway to Myofibrillar Disruption in Nemaline Myopathy. *American journal of human genetics* **93**, 1108-1117

- 19. Williams, S. K., Spence, H. J., Rodgers, R. R., Ozanne, B. W., Fitzgerald, U., and Barnett, S. C. (2005) Role of Mayven, a kelch-related protein in oligodendrocyte process formation. *Journal of neuroscience research* **81**, 622-631
- 20. Seng, S., Avraham, H. K., Jiang, S., Venkatesh, S., and Avraham, S. (2006) KLHL1/MRP2 mediates neurite outgrowth in a glycogen synthase kinase 3beta-dependent manner. *Molecular and cellular biology* **26**, 8371-8384
- 21. Duke-Cohan, J. S., Gu, J., McLaughlin, D. F., Xu, Y., Freeman, G. J., and Schlossman, S. F. (1998) Attractin (DPPT-L), a member of the CUB family of cell adhesion and guidance proteins, is secreted by activated human T lymphocytes and modulates immune cell interactions. *Proceedings of the National Academy of Sciences of the United States of America* **95**, 11336-11341
- 22. Liang, X. Q., Avraham, H. K., Jiang, S., and Avraham, S. (2004) Genetic alterations of the NRP/B gene are associated with human brain tumors. *Oncogene* **23**, 5890-5900
- 23. Krem, M. M., Luo, P., Ing, B. I., and Horwitz, M. S. (2012) The kelch protein KLHDC8B guards against mitotic errors, centrosomal amplification, and chromosomal instability. *The Journal of biological chemistry* **287**, 39083-39093
- 24. Khwaja, F. W., Duke-Cohan, J. S., Brat, D. J., and Van Meir, E. G. (2006) Attractin is elevated in the cerebrospinal fluid of patients with malignant astrocytoma and mediates glioma cell migration. *Clinical cancer research : an official journal of the American Association for Cancer Research* **12**, 6331-6336
- 25. Kim, T. A., Lim, J., Ota, S., Raja, S., Rogers, R., Rivnay, B., Avraham, H., and Avraham, S. (1998) NRP/B, a novel nuclear matrix protein, associates with

p110(RB) and is involved in neuronal differentiation. *The Journal of cell biology* **141**, 553-566

- Kim, T. A., Jiang, S., Seng, S., Cha, K., Avraham, H. K., and Avraham, S. (2005) The BTB domain of the nuclear matrix protein NRP/B is required for neurite outgrowth. *Journal of cell science* **118**, 5537-5548
- Ueki, K., Ono, Y., Henson, J. W., Efird, J. T., von Deimling, A., and Louis, D. N. (1996) CDKN2/p16 or RB alterations occur in the majority of glioblastomas and are inversely correlated. *Cancer research* 56, 150-153
- 28. Jaramillo, M. C., and Zhang, D. D. (2013) The emerging role of the Nrf2-Keap1 signaling pathway in cancer. *Genes & development* **27**, 2179-2191
- 29. Lo, S. C., Li, X., Henzl, M. T., Beamer, L. J., and Hannink, M. (2006) Structure of the Keap1:Nrf2 interface provides mechanistic insight into Nrf2 signaling. *The EMBO journal* **25**, 3605-3617
- Hast, B. E., Cloer, E. W., Goldfarb, D., Li, H., Siesser, P. F., Yan, F., Walter, V., Zheng, N., Hayes, D. N., and Major, M. B. (2014) Cancer-derived mutations in KEAP1 impair NRF2 degradation but not ubiquitination. *Cancer research* 74, 808-817
- Ohta, T., Iijima, K., Miyamoto, M., Nakahara, I., Tanaka, H., Ohtsuji, M., Suzuki, T., Kobayashi, A., Yokota, J., Sakiyama, T., Shibata, T., Yamamoto, M., and Hirohashi, S. (2008) Loss of Keap1 function activates Nrf2 and provides advantages for lung cancer cell growth. *Cancer research* 68, 1303-1309
- 32. Panieri, E., and Santoro, M. M. (2016) ROS homeostasis and metabolism: a dangerous liason in cancer cells. *Cell death & disease* **7**, e2253
- 33. Kensler, T. W., and Wakabayashi, N. (2010) Nrf2: friend or foe for chemoprevention? *Carcinogenesis* **31**, 90-99
- Ishimoto, T., Nagano, O., Yae, T., Tamada, M., Motohara, T., Oshima, H., Oshima, M., Ikeda, T., Asaba, R., Yagi, H., Masuko, T., Shimizu, T., Ishikawa, T., Kai, K., Takahashi, E., Imamura, Y., Baba, Y., Ohmura, M., Suematsu, M., Baba, H., and Saya, H. (2011) CD44 variant regulates redox status in cancer

cells by stabilizing the xCT subunit of system xc(-) and thereby promotes tumor growth. *Cancer cell* **19**, 387-400

- 35. Singh, A., Bodas, M., Wakabayashi, N., Bunz, F., and Biswal, S. (2010) Gain of Nrf2 function in non-small-cell lung cancer cells confers radioresistance. *Antioxidants & redox signaling* **13**, 1627-1637
- Muscarella, L. A., Barbano, R., D'Angelo, V., Copetti, M., Coco, M., Balsamo, T., la Torre, A., Notarangelo, A., Troiano, M., Parisi, S., Icolaro, N., Catapano, D., Valori, V. M., Pellegrini, F., Merla, G., Carella, M., Fazio, V. M., and Parrella, P. (2011) Regulation of KEAP1 expression by promoter methylation in malignant gliomas and association with patient's outcome. *Epigenetics* 6, 317-325
- Muscarella, L. A., Parrella, P., D'Alessandro, V., la Torre, A., Barbano, R., Fontana, A., Tancredi, A., Guarnieri, V., Balsamo, T., Coco, M., Copetti, M., Pellegrini, F., De Bonis, P., Bisceglia, M., Scaramuzzi, G., Maiello, E., Valori, V. M., Merla, G., Vendemiale, G., and Fazio, V. M. (2011) Frequent epigenetics inactivation of KEAP1 gene in non-small cell lung cancer. *Epigenetics* 6, 710-719
- 38. Eades, G., Yang, M., Yao, Y., Zhang, Y., and Zhou, Q. (2011) miR-200a regulates Nrf2 activation by targeting Keap1 mRNA in breast cancer cells. *The Journal of biological chemistry* **286**, 40725-40733
- 39. Krem, M. M., and Horwitz, M. S. (2013) Mitotic errors, aneuploidy and micronuclei in Hodgkin lymphoma pathogenesis. *Communicative & integrative biology* **6**, e23544
- 40. Krem, M. M., Salipante, S. J., and Horwitz, M. S. (2010) Mutations in a gene encoding a midbody protein in binucleated Reed-Sternberg cells of Hodgkin lymphoma. *Cell cycle* **9**, 670-675
- 41. Salipante, S. J., Mealiffe, M. E., Wechsler, J., Krem, M. M., Liu, Y., Namkoong, S., Bhagat, G., Kirchhoff, T., Offit, K., Lynch, H., Wiernik, P. H., Roshal, M., McMaster, M. L., Tucker, M., Fromm, J. R., Goldin, L. R., and Horwitz, M. S. (2009) Mutations in a gene encoding a midbody kelch protein in familial and sporadic classical Hodgkin lymphoma lead to binucleated cells. *Proceedings of the National Academy of Sciences of the United States of America* **106**, 14920-14925

- 42. Mukasa, A., Wykosky, J., Ligon, K. L., Chin, L., Cavenee, W. K., and Furnari, F. (2010) Mutant EGFR is required for maintenance of glioma growth in vivo, and its ablation leads to escape from receptor dependence. *Proceedings of the National Academy of Sciences of the United States of America* **107**, 2616-2621
- 43. Kelley, L. A., Mezulis, S., Yates, C. M., Wass, M. N., and Sternberg, M. J. (2015) The Phyre2 web portal for protein modeling, prediction and analysis. *Nature protocols* **10**, 845-858

Chapter 5

KLHDC8A attenuates the effects of elevated ROS in the absence of EGFRvIII

5.1 Introduction

The earliest evidence for chemoprotective agents that inhibit carcinogenesis arising from chemical exposure dates back to over 85 years ago (1). Since then, the mechanism of their protection has been more clearly defined, whereby these chemoprotectors are known to induce the expression of cellular enzymes that metabolize carcinogens to less reactive forms (2). The inducing activity of electrophilic chemoprotective agents was found to be dependent on the presence of a cis-acting transcriptional enhancer sequence called the antioxidant response element (ARE) within the respective response genes (1). In the previous chapter, we described a primary example of this mechanism: Nrf2 function, which binds the ARE of its respective target genes upon activation by oxidants or electrophiles.

The cellular adaptive response to elevated levels of reactive oxygen species is regulated in multiple ways. This response depends primarily on the transactivation of cellular defense genes, but also on the stability of the transcription factors required for the production of these genes and the oxidative/electrophilic sensing proteins. The stress sensing ability of Keap1 could be describe as the cell's first line of defense against deleterious levels of ROS. Likewise, other proteins within the Kelch superfamily may participate in this process as well. In this chapter, we establish a novel link between KLHDC8A and oxidative stress, based on our hypothesis that KLHDC8A gene up-regulation is a genetic adaptation utilized by GBMs in response to excessive ROS levels.

Results

5.2. Up-regulation of KLHDC8A is consistently observed in GBM models of resistance to EGFRvIII inhibition

KLHDC8A was the second most up-regulated gene within the mRNA profiles of Escaper populations that emerged in our EGFRvIII genetic silencing model. Based on this finding, we first wanted to determine if this genetic change was unique to the U373MG-tetO-EGFRvIII model or if this reflected a common response in different contexts where overcoming blockade of EGFRvIII is observed. To address this question, we first validated the microarray findings for all of the genes that were determined to be associated with "escape" by Mukasa *et al.*, using quantitative polymerase chain reaction (qPCR) analysis (full data not shown). With this approach, we confirmed that KLHDC8A mRNA expression is indeed most expressed in Escapers compared to the EGFRvIII-positive populations and the acute EGFRvIIInegative populations (**Figure 5.1**).

Next, we examined the expression of KLHDC8A mRNA in two established EGFR TKI-resistant models. The first model we examined was the *Ink4a/arf^{f-}* EGFRvIII murine astrocytes, a model in which both therapeutic sensitivity and acquired resistance to gefitinib phenotypically resembles our genetic model (**Chapter 3**). The second model was the patient derived xenograft (PDX) GBM39 neurosphere model. Resistance in this model is observed following chronic treatment with the EGFR TKI erlotinib in vivo (3). In both models, we found that KLHDC8A mRNA was significantly up-regulated in the resistant populations compared to the sensitive populations (Control) (**Figure 5.1**). Taken together, these data suggest that KLHDC8A is involved in multiple contexts of acquired resistance, and potentially reflects a common genetic adaptation in response to EGFRvIII inhibition.



Figure 5.1. Up-regulation of KLHDC8A is common to in several GBM models. Quantitative PCR analysis for KLHDC8A gene expression in three GBM models: (1) U373MG-tetO-EGFRvIII, (2) *Ink4a/arf*^{-/-} EGFRvIII murine astrocytes, (3) GBM39 PDX. Data are means \pm SEM ****p<0.0001, **p<0.01, n.s. = not significant

5.3. KLHDC8A interacts with the TRiC/CCT chaperonin complex

In order to delineate a functional role for KLHDC8A, we next conducted screening studies to identify putative binding partners. Currently, there is not a reliable KLHDC8A antibody to be used for interaction studies, so for this objective, we employed tagged versions of KLHDC8A. In our first study, we stably transduced HEK293T cells with an engineered C-terminal FLAG-tagged KLHDC8A construct and performed immunoprecipitation followed by liquid chromatography-coupled tandem mass spectrometry (LC-MS/MS) (**Figure 5.2A**). In this experiment, 39 proteins were identified as putative binding partners (**Figure 5.2C**). We followed up this study with a second interaction approach: proximity-dependent biotin identification (BioID). This technique utilizes a 35 kDa *Escherichia coli* DNA binding biotin ligase called BirA* (R118G) that can be fused to any protein of interest (4). Upon supplementation of culture medium with biotin, BirA biotinylates both direct interacting and vicinal proteins, affording the identification of both transient and permanent interactions (4). Subsequently, the biotinylated proteins can be isolated by streptavidin affinity capture, and then identified by mass spectrometry.

We generated an N-terminal BirA-Myc_KLHDC8A fusion protein and stably transduced HEK293 cells with this construct (**Figure 5.2B**). This time, we identified 163 putative binding partners, and so to triage our extensive list of binding partners, we compared the hits from both approaches to yield a final list of 15 common associations (**Figure 5.2C**). Within our final list, we noticed that there were six proteins that are individual subunits of the TRiC/CCT chaperonin complex. This complex is an ATP-dependent protein folding machine consisting of two stacked rings of eight different but related (CCT1-CCT8) subunits and the apical domain of each

subunit contributes to lid formation that covers the cavity in the form of a cap (5). Approximately, 5-15% of newly synthesized proteins rely on this complex for folding and assembly, and these substrates are processed in a very exclusive fashion, whereby the complex completely encapsulates them during folding (5). Determinants of substrate selection have been established for this complex and they include: 1) proteins that are components of oligomeric complexes, 2) proteins that are ~40-75 kDa in size 3) proteins that are enriched in hydrophobic sequences, and 4) proteins with high β -sheet propensity, particularly long stretches of 35-45 amino acids of β -sheet content (6).

Based on what is currently known about the molecular profile of KLHDC8A, three of the four determinants are fulfilled, and thus, it is likely a TRiC/CCT substrate. We confirmed that this complex directly interacts with KLHDC8A by coimmunoprecipitation. U373MG-tetO-EGFRvIII GBM cells were stably transduced with a C-terminal Myc-tagged KLHDC8A construct, and Myc-tag-associated proteins in the presence and absence of EGFRvIII expression were isolated by affinity capture (**Figure 5.2D**). We analyzed the captured immunoprecipitates by Western Blot, and indeed were able to isolate a substantial amount of the TCP1 subunit in our KLHDC8A_Myc cells, but not in the empty vector (EV) control cells (**Figure 5.2D**). Furthermore, our results demonstrate that this interaction did not require EGFRvIII, which is consistent with the mRNA expression data, whereby a role for KLHDC8A is predicted to be specific to EGFRvIII-independent contexts.



Figure 5.2. KLHDC8A interacts with the TRiC/CCT chaperonin complex. Multiple protein interaction studies revealed proteins commonly associated with KLHDC8A. (A) Co-immunoprecipitation-Mass spectrometry approach using FLAG-tagged KLHDC8A construct in HEK293T cells. (B) Proximity dependent biotin identification-Mass spectrometry approach using the BirA-Myc-KLHDC8A fusion construct in HEK293 cells. (C) Comparative analysis to establish common elements between the two lists identified several subunits of the TRiC/CCT chaperonin complex (in bold) as consistently interacting with KLHDC8A. (D) Immunoprecipitation studies conducted in U373MG-tetO-EGFRvIII GBM cells to validate the direct interaction between KLHDC8A and the TRiC/CCT chaperonin complex.

A great deal of work has been done to characterize how essential the TRiC/CCT complex is to cell viability and also to establish a list of its bonafide substrates. Initially its main substrates were only thought to be actin and tubulin, but a more defined TRiC/CCT interactome has been established through various experimental and computational efforts (6). Other notable TRiC/CCT substrates identified include: the histone deacetylase 3 (HDAC3), the telomerase cofactor TCAB1, the WD40 repeat protein WDR68, the von-Hippel Lindau tumor suppressor protein, and cyclin E (7-11), which have all been linked to oncogenesis.

Interestingly, Gallina *et al.* recently demonstrated that the TRiC/CCT complex is involved in protein quality control and recovery from genotoxic stress. In this work, both the yeast Cmr1 protein and its human orthologue WDR76 were shown to directly interact with components of the TRiC/CCT complex within the context of a novel intranuclear quality control (INQ) compartment (12). This INQ serves as a hub for the establishment of molecular adaptations required for resumption of DNA replication. More importantly, the tertiary structures of Cmr1 and WDR76 are very similar to KLHDC8A, in that they are predominantly made up B-propeller domains. In the absence of these domains, formation of the INQ is impaired, and consequently no recovery from DNA damage is observed (12).

With consideration for these findings, we suspect that the significance of the association between TRiC/CCT and KLHDC8A is linked to cellular response to genotoxic stress that is specifically caused by reactive oxygen species. Thus, we briefly examined the subcellular localization for KLHDC8A, to see if it can also localized to the nucleus, since the involvement of the TRiC/CCT complex to the cell recovery process took place in the nucleus (12). We took our U373MG-tetO-EGFRvIII

cells, stably expressing KLHDC8A-MYC, and performed immunofluorescence staining on these cells using a myc-tag specific antibody as an indicator for KLHDC8A localization. The myc-tag was primarily detected in the nuclear compartments of cells, as reflected in the merge with the DAPI nuclei stain (**Figure 5.3**). Interestingly, in some cases we saw co-localization with nuclear EGFRvIII, but within the same population, we also observed EGFRvIII-positive cells that were negative for KLHDC8A-MYC (**Figure 5.3**). When EGFRvIII was silenced in these cells by doxycycline treatment, we still observed nuclear localization, further supporting a possible EGFRvIII-independent function for KLHDC8A. Additionally, if KLHDC8A does function in cellular recovery from genotoxic stress, then a nuclear role for its involvement would be expected.



U373MG-tetO-EGFRvIII

Figure 5.3. KLHDC8A is found in the nucleus independent of EGFRvIII in GBM. Representative immunofluorescence staining for EGFRvIII and KLHDC8A-MYC localization in U373MG-tetO-EGFRvIII cells. Doxycycline treatment was done for 6 days prior to staining. EGFRvIII (FITC-channel); KLHDC8A-MYC (TRITC-channel)

5.4. Loss of KLHDC8A in Escapers leads to an increase in DNA double strand breaks

Our hypothesis is that up-regulation of KLHDC8A represents a genetic adaptation in response to changes in ROS levels. We reasoned then that targeting KLHDC8A should once again restore sensitivity to EGFRvIII inhibition, and this restored sensitivity would be reflected phenotypically. Thus, we investigated the requirement for KLHDC8A in the acquisition of phenotypes that support EGFRvIII independence.

Beginning with two Escapers, whereby loss of KLHDC8A led to impaired in vivo growth (13), we examined the extent of DNA damage burden in these populations by immunohistochemistry. We used 53BP1 positive nuclei as our standard readout for DNA damage, and observed a significant increase in DNA double strand breaks in the KLHDC8A knockdown samples compared to the negative control samples (**Figure 5.4**). This increase was evident in both escapers, suggesting that the role of KLHDC8A in the maintenance of these individual populations is the same. Furthermore, this data represents the first evidence that links KLHDC8A to DNA damage.



Figure 5.4. siRNA-mediated knockdown of KLHDC8A promotes the persistence of unrepaired DNA double strand breaks in Escapers. Representative immunohistochemical staining for 53BP1 nuclei in Escaper siKLHDC8A populations. Quantification of positively stained nuclei was determined using microscopy cell counting software. Escaper 0 siK8A No.1: (2.5-fold increase), Escaper 0 siK8A No.2: (3.5-fold increase), Escaper 1 siK8A No.1: (12.9-fold increase), Escaper 1 siK8A No.2: (16.3-fold increase) Data are means <u>+</u> SEM **** p<0.0001, **p<0.01, *p<0.05

5.5. Combined loss of EGFRvIII and KLHDC8A results in elevated ROS activity and oxidative DNA damage

We next moved into examining if reactive oxygen species was the source of the increase in DNA damage observed in KLHDC8A silenced Escaper tumors. To address this question, we took cultured Escaper lines that were generated from the original tumors and stably expressed the two KLHDC8A-specific siRNAs in these cells. We confirmed the knockdown efficiency by quantitative PCR analysis (**Figure 5.5A**), and then assessed the ROS activity in these cells using our two ROS indicators. Overall, we saw a significant increase in mitochondrial ROS activity when KLHDC8A was inhibited as reflected in the percentage of MitoSOX Red positive cells (**Figure 5.5B-C**). This was observed for one siRNA in Escaper 0 cells and for both siRNAs in Escaper 1 cells.

Next, we examined the impact of inhibiting KLHDC8A on cytoplasmic ROS activity. In majority of the cases, we did not observe an appreciable increase in cytoplasmic ROS activity (**Figure 5.6**). The lack of a change in cytoplasmic ROS activity is reminiscent of the findings from the EGFRvIII pharmacological inhibition studies shown in chapter 3. This implies that KLHDC8A can substitute for EGFRvIII, but specifically for the EGFRvIII kinase-dependent tumorigenic activities.

Lastly, we assessed if the impact on mitochondrial ROS activity alone would be sufficient for an observable increase in oxidative DNA damage in vivo. We took tumor tissue from the KLHDC8A knockdown populations and stained for 8oxoguanine. In every condition, we observed an increase in the detection of 8oxoguanine (**Figure 5.7**). Taken together, these data suggest that targeting KLHDC8A alone is sufficient to restore sensitivity to EGFRvIII inhibition in Escapers. Figure 5.5. Inhibition of KLHDC8A restores sensitivity to EGFRvIII inhibition in the form of increased mitochondrial ROS levels. (A) Knockdown efficiency for KLHDC8A mRNA was validated by quantitative PCR in both Escaper cell lines. (B) Representative Flow Cytometry plots depicting ROS activity as measured by the PE-positive population (Q3) for each condition. Blue=unlabeled cells; Red=labeled cells (C) Mean percentage of positive cells after background subtraction. MitoSOX Red is a superoxide indicator. Data are means \pm SEM ****p<0.0001,**p<0.01





Mitochondrial ROS



С

Escaper 0








Figure 5.6. Cytoplasmic ROS activity is not affected when KLHDC8A is inhibited. (A) Representative Flow Cytometry plots depicting ROS activity as measured by the PE-positive population (Q3) for each condition. Blue=unlabeled cells; Red=labeled cells (C) Mean percentage of positive cells after background subtraction. Cell ROX Orange is general oxidative stress indicator. Data are means + SEM *p<0.05

siLuciferase

siK8A no.2

siLuciferase



Figure 5.7. Loss of KLHDC8A is associated with an increase in oxidative DNA damage. (A) Representative immunohistochemical staining for 8-oxoguanine, in resected tumors (B) Quantification of staining intensity. Data reflect the individual cases and their respective extent of oxidative stress, represented as percentages. Esc 0 siNEG: low-50%; medium-50%; high-0% vs Esc 0 siK8A No.1: low-40%; medium-33%; high-27% and Esc 0 siK8A No.2: low-0%; medicum-37%; high-63% Esc 1 siNEG: low-55%; medium-27%; high-18% vs Esc 1 siK8A No.1: low-10%; medium-80%; high-10% and Esc 1 siK8A No.2: low-0%; medium-50%, high-50%

5.6 Summary

To date, the only established role for KLHDC8A was for the maintenance of in vivo tumorigenicity in the absence of EGFRvIII in GBM. Here, we extend its role to the context of mediating ROS levels and oxidative DNA damage. KLHDC8A is consistently up-regulated when acquired resistance emerges, and so far our work suggests that the significance of this may be to protect against oxidative stress overload.

KLHDC8A was confirmed to interact with the Group II chaperonin complex, TRiC/CCT. Recently, a role for this complex has extended beyond protein folding and assembly into mediating recovery from oxidative stress as part of a novel intranuclear quality control compartment (INQ). Our preliminary localization studies on KLHDC8A revealed that it localizes in the nucleus in an EGFRvIII-independent fashion. Thus, we suspect that the direct association between KLHDC8A and TRiC/CCT complex may not be simply due to it being a substrate, but also due to their involvement in cellular stress responses.

Similar to EGFRvIII receptor activity, KLHDC8A seems to be required specifically in the context of mitochondrial ROS activity. Although, we did not observe any impact on cytoplasmic ROS activity, we still saw a significant increase in oxidative DNA damage in the respective in vivo samples. This would suggest that mitochondrial ROS activity is the primary determinant for DNA oxidation, but additional studies will be required to establish this claim.

In summary, we have established a novel link between KLHDC8A, ROS and oxidative DNA damage. To our knowledge, this is the first evidence demonstrating such an association and characterization of this protein. We believe that this may be a common occurrence in multiple contexts of therapeutic resistance to EGFRvIII targeting. The broader implications of this work and the required future studies will be discussed in the next chapter.

Chapter 5 contains data that have not been published previously, and are currently being prepared for submission for publication. The work in chapter 5 was completed with the help of Jill Wykosky, PhD., a former member of the laboratory of Frank Furnari and Jason Liang, a fellow PhD Candidate in the laboratory of Huilin Zhou at the Ludwig Institute for Cancer Research. This work was supported by the NIH F31 NS076343 to (T.E. Taylor) and the NIH R01 NS080939 to (F. B. Furnari).

5.7 Materials and Methods

Cell culture and reagents

HEK 293T cells were cultured in Dulbecco's Modified Eagle's Medium (DMEM) supplemented with 10% heat-inactivated fetal bovine serum (FBS) and Penicillin/streptomycin (1:1000). HEK293 cells were cultured in Minimum Essential Medium (MEM) supplemented with 10% FBS, penicillin/streptomycin (1:1000), and sodium pyruvate (2.2 g/L). U373MG-tetO-EGFRvIII-dependent cells were cultured in (DMEM) supplemented with 10% Tet-approved FBS, Penicillin/streptomycin, G418 (2µg/ml), puromycin (1µg/ml) and blasticidin (2µg/ml). Escaper cells were cultured in the same media, but with the addition of doxycycline (1µg/ml). Tet-approved FBS was purchased from Genesee Scientific. Doxycycline was purchased from Clontech. Blasticidin was purchased from Invitrogen. Biotin was purchased form Sigma-Aldrich.

Construction of Plasmids

The pcDNA3.1mycBioID plasmid was obtained from Addgene. KLHDC8A cDNA was cloned into plasmid backbone between the EcoRI and HindIII restriction sites to generate the final myc-BirA*-KLHDC8A fusion protein. The KLHDC8A-myc cDNA insert was generated by PCR and cloned into the Nhel and BamHI restriction sites of the pLV-t2A-Blasticidin lentiviral vector to use for stable transduction. The KLHDC8A-FLAG cDNA insert was generated by PCR and cloned by PCR and cloned into the Sall and NotI restriction sites of the pDREF lentiviral vector to use for stable transduction. The Sall and NotI restriction sites of the pDREF lentiviral vector to use for stable transduction. The kLHDC8A-FLAG cDNA insert was generated by PCR and cloned into the Sall and NotI restriction sites of the pDREF lentiviral vector to use for stable transduction. The pSUPER.retro.blasticidin retroviral vector was used to generate stable constructs for knockdown of KLHDC8A. To generate new cells, the same constructs were used as

previously described in (13). The specific siRNA sequences for human KLHDC8A are: No.1: 5'-GCAGCAGCACAATGATTAA-3'; No.2: 5'-AGCGAGAATTGGACATGAA-3'; negative control: 5'-TTCTCTGAACGTGTCACGT-3'; and negative control (siLuciferase): 5'- CTTACGCTGAGTACTTCGA -3'

Stable Cell Line Generation

The HEK 293T packaging cell line was cotransfected with the pSUPER.retro.blast retroviral constructs and pCL10A1 or with the lentirviral construct pLV-t2A-Blast plus pCMV-VSV.G and delta 8.9 using Lipofectamine 2000 (Invitrogen), following the manufacturer's instructions. Lentiviral supernatants were harvested at 48, 72, and 96 hr after transfection, filtered (0.45 µm), then used for overnight infections of U373MG-tet-O-EGFRvIII and Escaper cells in the presence of 8µg/mL polybrene. Cells were allowed to recover in fresh media for 24 hours and were then selected in DMEM media containing 2µg/mL neomycin (G418), 1 µg/mL puromycin, and 4µg/mL blasticidin S for KLHDC8A-MYC and siKLHDC8A stable expression, respectively. HEK293 cells were cotransfected with the pcDNA3.1mycBioID lentiviral constuct and pCMV-VSV.G and delta 8.9 using Lipofectamine 2000, then selected in complete expression. HEK293T were cotransfected with the pDREF lentiviral constuct and pCMV-VSV.G and delta 8.9 using Lipofectamine 2000, then selected in complete DMEM media containing 4µg/mL neomycin (G418) for myc-BirA*-KLHDC8A stable expression.

Co-Immunoprecipitation

Cells were lysed for protein harvesting using RIPA lysis buffer (50 mM Tris, pH 7.5, 150 mM NaCl, 1 mM EDTA pH 8.0, 1 mM NaF, 1% NP40, 0.1% SDS, 0.5% sodium deoxycholate and 1x Complete protease inhibitor [Roche]) and phosphatase inhibitors) for the Anti-c-Myc approach and lysis buffer (50 mM Tris HCl, pH7.4, with 150 mM NaCl, 1 mM EDTA, and 1% Triton X-100) for the Anti-FLAG approach. Lysates were incubated with either 20µl of Anti-c-Myc or Anti-FLAG agarose based on the respective tag, and incubated overnight at 4°C on end-over-end rotator. Agarose resin was pelleted with a 10-second pulse at 12,000 x g. Resin was washed three times with 500µl of TBST (TBS with 0.05% Tween-20 Detergent), pelleting resin in between washes. Bound proteins were eluted by boiling at 98°C in sample buffer for KLHDC8A-myc, and eluted fractions were analyzed by Western Blotting. Bound protein were eluted using a 3XFLAG peptide for FLAG-KLHDC8A, and eluted fractions were analyzed by mass spectrometry to identify putative binding partners.

Affinity Capture of Biotinylated Proteins

Cells were grown until 90% confluency then incubated for 24 hr in complete media supplemented with 50µM biotin. After one PBS wash, cells were harvested off dish by scraping in 10 ml of 1X PBS on ice. Cells were pelleted by centrifuging at 1200 rpm for 5 min at 4°C. After resuspending in 1 ml of 1X PBS, cells were transferred to a 1.5 ml eppendorf tube, pelleted once more, then lysed at 25°C in 1.5 ml RIPA lysis buffer (50 mM Tris, pH 7.5, 150 mM NaCl, 1 mM EDTA pH 8.0, 1 mM NaF, 1% NP40, 0.1% SDS, 0.5% sodium deoxycholate and 1x Complete protease inhibitor [Roche]) and phosphatase inhibitors) and sonicated on ice (2 sec on/2 sec off for 16 seconds at

10% amplitude). After sonication, lysates were filtered through an Amicon Ultra 3K (2 ml) device to remove free biotin and to concentrate the sample. Lysates were precleared with 250µl Protein G agarose beads, and incubation overnight at 4°C on rotator. Supernatants were transferred to a new eppendorf tube and lysate was incubated again with 250µl agarose beads (MyOne Steptavadin C1; Invitrogen) overnight at 4°C on rotator. Beads were collected and washed twice for 8 min at 25°C (all subsequent steps at 25°C) in 1 ml wash buffer 1 (2% SDS in dH₂O). This was repeated once with wash buffer 2 (0.1% deoxycholate, 1% Triton X-100, 500 mM NaCl, 1 mM EDTA, and 50 mM Hepes, pH 7.5), once with wash buffer 3 (250 mM LiCl, 0.5% NP-40, 0.5% deoxycholate, 1 mM EDTA, and 10 mM Tris, pH 8.1) and twice with wash buffer 4 (50 mM Tris, pH 7.4, and 50 mM NaCl). 25% of the sample was reserved for Western blot analysis. Bound proteins were removed from the agarose beads with 50µl of 4X SDS-sample buffer, no bromophenol blue, saturated with biotin (10mM) at 98°C. 75% of the sample was analyzed by mass spectrometry.

Western blotting

Cells were lysed for protein for harvesting using RIPA Buffer (50 mM Tris, pH 7.5, 150 mM NaCl, 1 mM EDTA pH 8.0, 1 mM NaF, 1% NP40, 0.1% SDS, 0.5% sodium deoxycholate and 1x Complete protease inhibitor [Roche]) and phosphatase inhibitors). Total protein was quantified using BCA Protein Assay Reagent (Thermo Scientific). Afterwards, 40µg of each protein sample was resolved on a 7.5% to 15% polyacrylamide gel. Proteins were transferred, blocked with 5% milk followed by primary antibody incubation overnight at 4°C. The following antibodies were used at specified concentrations for immunoblots: EGFR clone 13 (1:2500; BD Transduction

Labs), GAPDH (1:1000; Cell Signaling), α-tubulin (1:500, Santa Cruz Technology), βactin-HRP (1:25000, Sigma-Aldrich), FLAG-tag (1:2000, OriGene), myc-tag (1:1000, Cell Signaling) and TCP1-α (1:500, Santa Cruz Technology. Membranes were washed three times with TBS containing 0.1% Tween-20 and incubated with antimouse or anti-rabbit-HRP antibody (Sigma). Chemiluminescence was detected using Amersham ECL Western Blotting Detection Reagent (GE Healthcare) or SuperSignal West Pico Chemiluminescent Substrate (Thermo Scientific). Images were obtained using ChemiDoc MP imaging system and Image Lab 4.1 software (Bio-Rad).

Protein Identification by Mass Spectrometry

A fraction of the proteins eluted from the streptavidin agarose beads by SDS-sample buffer were reduced and alkylated and separated by SDS-PAGE. Separated proteins were visualized by Silver staining to confirm uniqueness prior to mass spectrometry analysis. Next, on bead tryptic digests were analyzed by 1D LC/MS/MS. Dithiothreitol (DTT) (15 mM) was added to beads suspension mix and proteins were reduced at 42°C for 30 min. lodoacetamide was added (to 30 mM) and proteins were alkylated at 25°C for 30 min in the dark. Mass spectrometry grade trypsin (Promega) was added (~1:50 ratio) for overnight digestion at 37°C. Agarose beads were removed by centrifugation. Peptides were desalted using a C18 SepPak and lyophilized under vacuum followed by resuspension in 80% acetonitrile/20% water before on-line analysis of peptides by high-resolution, high-accuracy LC-MS/MS, consisting of a TSK gel HILIC, a 15-cm Amide-80 column, a low-flow ADVANCED Michrom MS source, and a LTQ-Orbitrap XL (Thermo Fisher Scientific). A 120-min gradient (20% water, 80% acetonitrile) was used to separate the peptides. Samples were then analyzed using a LTQ tandem mass spectrometer (Thermo Fisher Scientific, San Jose, CA). Raw MS/MS spectra were searched by SEQUEST on a Sorcerer system for for protein identification against the IPI human protein database v3.80 from EBI (European Bioinformatics Institute), which contains semi-tryptic peptide sequences with the allowance of up to two missed cleavages and precursor mass tolerance of 50.0 ppm. The following search parameters were used: a molecular mass of 57.021465 Da was added to all cysteines to account for alkylation, semi-tryptic, mass tolerance of 1.2 Da for precursor ions, and 226 D on N terminus and lysine for biotinylation. Search results were sorted, filtered, statically analyzed, and displayed using PeptideProphet and ProteinProphet (Institute for Systems Biology). The minimum trans-proteomic pipeline (TPP) probability score for proteins was set to 0.95, to assure TPP error rate of lower than 0.01.

Quantitative real-time PCR

A total of 100,000 transduced cells were seeded in triplicate wells of a 6-well plate. After 24 hours, the cells were put into serum-free DMEM. The following day, RNA was isolated from the cells using the RNeasy Plus Kit (Qiagen). cDNA was synthesized using the RNA to cDNA EcoDry Premix (Clontech). qPCR was performed with a CFX96 Real-Time System (BioRad) using the following program: 95°C for 10 minutes, [95°C for 15 seconds; 58°C for 1 minute] x 40 cycles, 95°C for 1 min and 55°C for 1 min. KLHDC8A and actin gene expression were measured using SYBR Green (BioRad) with internal triplicate determinations for each sample.

<u>Immunohistochemistry</u>

Tissue sections were cut from formalin-fixed paraffin-embedded xenograft tissue. Sections were deparaffanized, re-hydrated, and then stained with either anti-53BP1 (1:100; Cell Signaling) or anti-8-oxoguanine (1:1000; Rockland). Heat-induced antigen retrieval was performed for the 53BP1 staining using 10mM sodium citrate, pH 6.0. Antigen retrieval was omitted for 8-oxoguanine staining. Primary antibodies were incubated on slides overnight at 4°C in a humidified chamber. Secondary antibody incubation plus AEC (Scytek) for 53BP1 or plus DAB (Vectastain) for 8-oxoguanine detection, was done according to the manufacturer's instructions. Slides were dehydrated through a series of alcohol and xylene, then coverslips were mounted onto slides. Quantification of positively stained nuclei was conducted using microscopy cell counting software. Stain intensity scoring was determined by establishing an intensity rubric based on the samples displaying the highest, middle, and lowest staining across all samples.

<u>Immunofluorescence</u>

U373MG-tetO-EGFRvIII-dependent cells were seeded on coverslips and allowed to grow for 48 hours. At this point, cells were fixed in 4% paraformaldehyde (pH 7.1) for 20 min at room temperature followed by a three PBS washes and permeabilization with 0.3% Triton X-100 in PBS buffer for 10 min. After blocking with PBS containing, 2% BSA IgG free, 10% normal goat serum, and 0.3M glycine, cells were incubated with Myc-tag antibody (1:8000, Cell Signaling (9B11)) followed by TRITC-conjugated secondary antibody. Nuclei were counterstained with DAPI simultaneous to mounting coverslips on slides. Images were captured using a fluorescence microscope.

ROS Assays

A total of 50,000 to 80,000 cells were seeded into 6-well plates. After 48 hours, cells were labeled with either MitoSOX Red (superoxide indicator; Life Technologies) or CellROX Orange (general oxidative stress indicator; Life Technologies) according to the manufacturer's instructions. Escaper cells were treated with Doxycycline (1 μ g/ml) throughout the entire course of the experiment. Samples were analyzed for cellular fluorescence on an LSR II (BD Biosciences) and data were analyzed with FlowJo software (Treestar).

Statistical analysis

All data are presented with mean \pm SEM as determined by Prism software (GraphPad). Unpaired Student's t-test or ANOVA was used as appropriate. P values of 0.05 or less were considered significant for all experiments. Data are representative of results obtained in at least two to three independent experiments.

5.8. References

- 1. Wasserman, W. W., and Fahl, W. E. (1997) Functional antioxidant responsive elements. *Proceedings of the National Academy of Sciences of the United States of America* **94**, 5361-5366
- 2. Miller, E. C., Miller, J. A., Brown, R. R., and Macdonald, J. C. (1958) On the protective action of certain polycyclic aromatic hydrocarbons against carcinogenesis by aminoazo dyes and 2-acetylaminofluorene. *Cancer research* **18**, 469-477
- Nathanson, D. A., Gini, B., Mottahedeh, J., Visnyei, K., Koga, T., Gomez, G., Eskin, A., Hwang, K., Wang, J., Masui, K., Paucar, A., Yang, H., Ohashi, M., Zhu, S., Wykosky, J., Reed, R., Nelson, S. F., Cloughesy, T. F., James, C. D., Rao, P. N., Kornblum, H. I., Heath, J. R., Cavenee, W. K., Furnari, F. B., and Mischel, P. S. (2014) Targeted therapy resistance mediated by dynamic regulation of extrachromosomal mutant EGFR DNA. *Science* 343, 72-76
- 4. Roux, K. J., Kim, D. I., Raida, M., and Burke, B. (2012) A promiscuous biotin ligase fusion protein identifies proximal and interacting proteins in mammalian cells. *The Journal of cell biology* **196**, 801-810
- Cong, Y., Baker, M. L., Jakana, J., Woolford, D., Miller, E. J., Reissmann, S., Kumar, R. N., Redding-Johanson, A. M., Batth, T. S., Mukhopadhyay, A., Ludtke, S. J., Frydman, J., and Chiu, W. (2010) 4.0-A resolution cryo-EM structure of the mammalian chaperonin TRiC/CCT reveals its unique subunit arrangement. *Proceedings of the National Academy of Sciences of the United States of America* **107**, 4967-4972
- Yam, A. Y., Xia, Y., Lin, H. T., Burlingame, A., Gerstein, M., and Frydman, J. (2008) Defining the TRiC/CCT interactome links chaperonin function to stabilization of newly made proteins with complex topologies. *Nature structural* & molecular biology 15, 1255-1262
- 7. Feldman, D. E., Thulasiraman, V., Ferreyra, R. G., and Frydman, J. (1999) Formation of the VHL-elongin BC tumor suppressor complex is mediated by the chaperonin TRiC. *Molecular cell* **4**, 1051-1061
- 8. Freund, A., Zhong, F. L., Venteicher, A. S., Meng, Z., Veenstra, T. D., Frydman, J., and Artandi, S. E. (2014) Proteostatic control of telomerase function through TRiC-mediated folding of TCAB1. *Cell* **159**, 1389-1403

- 9. Guenther, M. G., Yu, J., Kao, G. D., Yen, T. J., and Lazar, M. A. (2002) Assembly of the SMRT-histone deacetylase 3 repression complex requires the TCP-1 ring complex. *Genes & development* **16**, 3130-3135
- 10. Miyata, Y., Shibata, T., Aoshima, M., Tsubata, T., and Nishida, E. (2014) The molecular chaperone TRiC/CCT binds to the Trp-Asp 40 (WD40) repeat protein WDR68 and promotes its folding, protein kinase DYRK1A binding, and nuclear accumulation. *The Journal of biological chemistry* **289**, 33320-33332
- 11. Won, K. A., Schumacher, R. J., Farr, G. W., Horwich, A. L., and Reed, S. I. (1998) Maturation of human cyclin E requires the function of eukaryotic chaperonin CCT. *Molecular and cellular biology* **18**, 7584-7589
- Gallina, I., Colding, C., Henriksen, P., Beli, P., Nakamura, K., Offman, J., Mathiasen, D. P., Silva, S., Hoffmann, E., Groth, A., Choudhary, C., and Lisby, M. (2015) Cmr1/WDR76 defines a nuclear genotoxic stress body linking genome integrity and protein quality control. *Nature communications* 6, 6533
- Mukasa, A., Wykosky, J., Ligon, K. L., Chin, L., Cavenee, W. K., and Furnari, F. (2010) Mutant EGFR is required for maintenance of glioma growth in vivo, and its ablation leads to escape from receptor dependence. *Proceedings of the National Academy of Sciences of the United States of America* **107**, 2616-2621

Chapter 6 Discussion

6.1 Summary

EGFR and mutant EGFRvIII represent common genetic abnormalities encountered in primary GBM. Over the past several decades, comprehensive roles for their involvement in a number of tumorigenic properties have been established. Based on these tumor-promoting roles, several therapeutic agents have been developed to disrupt their function. However, the efficacy of these agents has been limited by both upfront and acquired resistance. GBM tumors invariably recur, and a great deal of work has been done to characterize new resistance mechanisms that support relapse, in order to inform the development of more durable therapies.

Here, we have shown that another mechanism of acquired resistance to EGFRvIII inhibition involves overcoming oxidative DNA damage. In our primary GBM model, we observed a pattern, whereby the phenotypes that were associated with sensitivity to EGFRvIII inhibition, were overturned upon relapse. These phenotypes included impaired proliferative capacity, increased apoptosis, and an increase in DNA damage burden. The latter phenotype was established in this work, and was found to be the result of elevated ROS activity. On the contrary, GBM populations that have overcome blockade of EGFRvIII displayed both reduced ROS activity and DNA oxidation. Thus, it appears that the very qualities that commonly contribute to tumor initiation and progression (i.e., the tolerance of DNA damage and subsequent mutagenesis) may also threaten the fitness of cancer cells and impede therapeutic escape.

These results imply that a dependence of cancer cells on antioxidant systems is a specific vulnerability that can be exploited to induce targeted cell death. Normal cells are characterized at having lower intracellular ROS levels, and so one way to achieve this would be to increase oxidative stress above the toxicity threshold in a tumor-specific manner. We have shown that the up-regulation of KLHDC8A is unique to populations that have acquired resistance to therapies targeted against EGFR and EGFRvIII, and that by inhibiting this gene, therapeutic sensitivity is restored. However, it remains to be determined if these induced levels reflect a tumor-specific signature.

KLHDC8A was also found to interact with the TRiC/CCT chaperonin complex, and fulfills many of the determinants of substrate recognition for this complex. The TRiC/CCT interactome encompasses many functional classes of cytoplasmic proteins, including those involved in cytoskeleton organization, DNA replication and repair, cell-cycle progression, RNA processing, and protein trafficking (1,2). Our work demonstrated that KLHDC8A is also found in the nucleus, yet it is still unknown if its interaction with the TRiC/CCT complex occurs in the cytoplasm or the nucleus.

More recently, the TRiC/CCT chaperonin complex, along with other components of the intranuclear quality control (INQ) compartment, have been implicated in cellular capacity to recover from genotoxic stress (3). Furthermore, it has previously been linked to the context of cytoplasmic proteotoxic stress through its interaction with HSF1A, which activates the transcription of HSF1 stress response target genes (4). Given that many of the key regulators of cell growth and differentiation are clients of molecular chaperones, including the TRiC/CCT complex, chaperones are potential mediators of oncogenesis. Thus, it is plausible that the significance of the interaction between TRiC/CCT and KLHDC8A is integral to its role in conferring resistance to EGFRvIII inhibition, even if it simply modulates its folding.

In total, our studies have extended the role for KLHDC8A in gliomagenesis, which was previously only known to be required for in vivo tumor growth in the absence of EGFRvIII (5). We have additionally established that it is required to maintain endogenous ROS levels below the toxic threshold. Consequently, this leads to a decrease in persistent DNA damage and presumably therapeutic escape (**Figure 6.1**). However, there are still remaining questions regarding the exact mechanism for how KLHDC8A protects cancer cells from stress toxicity and stress-induced apoptosis.



Figure 6.1. Predicted model of KLHDC8A-mediated resistance to EGFRvIII inhibition. In the context of therapeutic sensitivity, EGFRvIII targeting leads to elevated intracellular ROS activity. Unchallenged by antioxidant defenses (e.g., superoxide dismutase, catalase, glutathione), these reactive species can oxidize DNA bases (i.e., guanine) causing double-strand breaks (DSBs). If these breaks are deemed irreparable, then either cell death or growth arrest/senescence results. In the absence of EGFRvIII receptor function, up-regulation of KLHDC8A can serve as an alternative way to regulate ROS levels. The interaction between the TRiC/CCT complex and KLHDC8A may mediate this role. Consequently, cancer cells are able to recover from oxidative stress, which affords the re-initiation of tumor growth (acquired resistance).

6.2 KLHDC8A and the DNA damage burden

EGFR and EGFRvIII have established roles in DNA double-strand break repair. This role is particularly linked to conferring radioresistance in GBM (6-9). Several reports have demonstrated that the primary DNA repair pathway that is employed by these two molecules is the non-homologous end-joining pathway (NHEJ), although a regulatory role in homologous recombination has been observed (10-12). Furthermore, both EGFR nuclear translocation and the activation of the DNAdependent protein kinase catalytic subunit (DNA-PKCs) repair protein were found to be essential to this process (6,8,13).

Interestingly, within our interaction study using the MycBirA* proximitydependent biotinylation approach, DNA-PKCs was one of the top five hits determined to interact with KLHDC8A, as established by number of peptides recovered. Moreover, additional components of the stress-induced INQ structure, besides Cmr1/WDR76 and the TRiC/CCT chaperonin complex, included Ku70 and Ku80, which together with DNA-PKCs form the NHEJ-dependent DNA-PK complex (3,14). The INQ was also concluded to be a site of rapid turnover and recycling of phosphorylated, ubiquitinated and sumoylated targets from stalled replication forks, in order to promote an efficient response to DNA replication stress (3).

A predicted functional role for KLHDC8A based on IBA (inferred biological aspect of ancestry) computational analysis (Gene Ontology) is that it is involved in ubiquitin-dependent protein catabolism. This implies that KLHDC8A may indirectly promote the recycling of amino acids for new protein synthesis by facilitating protein turnover. Thus, the role of KLHDC8A may be coupled to the TRiC/CCT complex by way of supplying the building blocks needed for protein synthesis. Furthermore,

chaperones function to impede the accumulation of damaged and misfolded proteins, which would otherwise induce cellular oxidative stress. Therefore, if KLHDC8A stimulates protein quality control mechanisms in an effort to dampen oxidative stress within the cell, this would presumably also decrease the extent of oxidative DNA damage.

A number of WD40 repeat proteins have been shown to directly bind DNA histones at the center of their β-propeller structures (15). A priori, KLHDC8A can alternatively play a more direct role in reducing the DNA damage burden via chromatin binding. Our work demonstrates that KLHDC8A is capable of nuclear entry, and so this would appropriately position it to similarly bind to DNA in order to facilitate DNA repair-promoting modifications. Indeed, nuclear EGFR was shown to drive DNA synthesis and DNA repair by phosphorylating Histone H4 at tyrosine 72, and the disruption of this interaction impaired tumor growth (13). Hence, KLHDC8A could substitute for EGFR in this manner.

Lastly, the fact that an accumulation of DNA damage does not always support tumor progression was quite a surprise to us, given that one of the fundamental features of cancer is genome instability. In fact, another characterized Escaper reflected this feature, displaying no evidence of a significant decrease in DNA damage and thriving tumor growth under these conditions (**Figure 6.2A-C**). Of note, this particular Escaper expresses the lowest levels of KLHDC8A, and so would not be expected to be dependent on its function (**Figure 6.2D**). Thus, some cancer cells are more reliant on an intact DNA repair pathway, which may provide a therapeutic window, whereby the disruption of these pathways could selectively target tumor cells. However, the factors that determine when the accumulation of DNA damage is a friend or a foe to cancer are still unknown. It would be interesting if the up-regulation of KLHDC8A were one of these tumor-specific determinants.



Figure 6.2. Some escapers do not rely on intact DNA repair pathways. (A) Representative immunohistochemical staining for 53BP1 in the EGFRvIII-independent population #5. (B) Quantification of positive nuclei, represented as percentages. (C) In vivo growth curve for this population in the presence of a significant DNA damage burden. (D). qPCR analysis of KLHDC8A mRNA levels in each individual population. Data are <u>+</u> SEM ****p<0.0001

6.3 KLHDC8A and elevated reactive oxygen species (ROS)

The genetic aberrations that are prevalent in cancer cells promote continuous and excessive production of reactive oxygen species to mediate proliferation (16). However, our work and that of others have shown that ROS can in turn cause DNA damage, triggering growth arrest or apoptosis. Thus, another aspect of oncogenedependence is reflected in the reliance of the cancer cells on the primary oncogene in order to propagate within such a pro-oxidant environment.

Upon inactivation of the oncogene, cancer cells experience a state of "oncogenic shock", which is defined by a shift in the levels of pro-survival signals and the pro-apoptotic signals, in favor of cell death (17). This shift could also be perpetuated by the loss of a key modulator of excessive cellular ROS. In our work, we found that blocking the receptor activity of EGFRvIII led to an increase in mitochondrial ROS activity, but cytoplasmic ROS levels remained low. It has been reported that elevated cytoplasmic ROS results in a decrease in lifespan (18). Thus, EGFRvIII kinase activity may regulate mitochondrial ROS levels, while a kinase-independent function may be responsible for maintaining low cytoplasmic ROS levels. Cytoplasmic ROS function as part of signaling pathways; therefore, this could explain why simply blocking EGFRvIII receptor activity is insufficient, since EGFRvIII kinase-independent functions may still interfere with ROS signaling that would decrease cell fitness. Hence, our approach to genetically silence EGFRvIII could result in a more durable response.

Additionally, our work has established a new link between KLHDC8A and mitochondrial ROS homeostasis. Though we have not delineated the mechanism by which KLHDC8A is involved in this process, we suspect that the antioxidant enzyme systems are recruited. These include: superoxide dismutase 1 and 2, catalase, glutathione peroxidase, peroxiredoxin and thioredoxin. It remains to be determined if loss of KLHDC8A results in a change in either the transcription or translation of these molecules, and how their expression levels align with our current results. Additionally, elevated ROS activity, particularly from peroxynitrite, is known to negatively affect mitochondrial integrity and cause irreversible modifications to proteins, and so it would be interesting to see if these aberrations persist in the absence of KLHDC8A (19). Likewise, it is known that an intact actin cytoskeleton is required for the formation of the juxtanuclear quality control compartment (JUNQ), one of the major cytosolic inclusion bodies that participates in cellular stress response by removing misfolded or aggregated proteins (20). A number of kelch proteins are known to bind actin and function in actin organization (21). Thus, in the event that KLHDC8A does not modulate mitochondrial integrity, it may instead promote the elimination of the resulting damaged proteins by participating in JUNQ architecture.

Based on the significance of amino acid residues within other kelch proteins in the adaptive response to oxidative stress (i.e. Keap1), it is conceivable that functional residues within the accessible loop regions of KLHDC8A may be important in this context. Cysteine residues are predominant targets of redox-linked regulation, and thus can function as sensors of oxidative stress. Moreover, these oxidation modifications can result in changes in conformation, activity, localization and proteinprotein interactions. Hence, determining the functional relevance of key amino acid residues, particularly the cysteines, would provide a framework to understand the requirement for KLHDC8A in ROS homeostasis.

6.4 Future Studies

While we have established that there is an association between KLHDC8A and oxidative DNA damage, there are still a number of unanswered questions regarding the specific biology behind this process. Future studies should focus on: (1) Establishing the significance of the interaction between KLHDC8A and the TRiC/CCT complex (2) Determining if within the nucleus, KLHDC8A directly associates with chromatin, and (3) if KLHDC8A possesses post-translational modification activity.

The significance of its interaction with the chaperonin complex would establish more clarity on what appears to be a type of symbiotic relationship. Does KLHDC8A require the TRiC/CCT complex to properly fold, and if so would the TRiC/CCT complex then support tumor progression by folding a client protein that is related to oncogenesis? Additionally, does KLHDC8A partner with the TRiC/CCT complex due to its potential role in protein catabolism, or does the entire interaction operate as a positive feedback loop? Answers to these questions would shed light on the involvement of protein quality control to therapeutic escape.

As it relates to DNA repair, it will be interesting to identify nuclear binding partners of KLHDC8A. Does KLHDC8A bind chromatin, and if so would the disruption of this interaction be sufficient to impact the DNA damage burden. Furthermore, the binding partners would determine if the role for KLHDC8A in the nucleus is different from its role in the cytoplasm. If so, would KLHDC8A then reflect another example, whereby a change in subcellular localization (i.e., NRP/B) results in the impartation of an oncogenic function. KLHDC8A is a kelch-domain containing only protein, meaning that it lacks the BTB domain. The BTB domain has been substantiated as the structure required for Kelch proteins to function as components of CUL3-RING E3 ubiquitin ligase complexes; establishing a protein-modifying role for these members. KLHDC8B is the most characterized kelch-only containing protein of the two, yet the contribution of its structure to its functional role is not known beyond the ability to bind actin. It would be interesting to see then, if by way of the kelch domains, these two kelch-only containing proteins are also capable of modifying substrates. Studies that would determine this possibility, could uncover the specific mechanisms that are either exploited or disrupted in brain and lymphatic neoplasia, respectively.

In sum, we have found that KLHDC8A promotes the maintenance of EGFRvIII-independent populations by modulating ROS homeostasis, and consequently the extent of DNA damage. Additionally, we have shown that by characterizing therapeutically sensitive phenotypes, one may predict possible escape routes. Interestingly, our work also defined a novel therapeutic window, whereby greater tumor response to radiation may be achievable. We hope that these findings will provoke more efforts toward characterizing tumor sensitivity in addition to identifying novel mechanisms of acquired resistance as a basis for GBM therapeutic development. Future studies that are specific to this project will seek to delineate how the protein structure of KLHDC8A defines its functional role in gliomagenesis.

Chapter 6 contains data that have not been published previously, and are currently being prepared for submission for publication. The dissertation author solely completed the work presented in chapter 6. This work was supported by NIH R01 NS080939 to (F. B. Furnari).

6.5 References

- Dekker, C., Stirling, P. C., McCormack, E. A., Filmore, H., Paul, A., Brost, R. L., Costanzo, M., Boone, C., Leroux, M. R., and Willison, K. R. (2008) The interaction network of the chaperonin CCT. *The EMBO journal* 27, 1827-1839
- Yam, A. Y., Xia, Y., Lin, H. T., Burlingame, A., Gerstein, M., and Frydman, J. (2008) Defining the TRiC/CCT interactome links chaperonin function to stabilization of newly made proteins with complex topologies. *Nature structural* & molecular biology 15, 1255-1262
- Gallina, I., Colding, C., Henriksen, P., Beli, P., Nakamura, K., Offman, J., Mathiasen, D. P., Silva, S., Hoffmann, E., Groth, A., Choudhary, C., and Lisby, M. (2015) Cmr1/WDR76 defines a nuclear genotoxic stress body linking genome integrity and protein quality control. *Nature communications* 6, 6533
- 4. Neef, D. W., Jaeger, A. M., Gomez-Pastor, R., Willmund, F., Frydman, J., and Thiele, D. J. (2014) A direct regulatory interaction between chaperonin TRiC and stress-responsive transcription factor HSF1. *Cell reports* **9**, 955-966
- Mukasa, A., Wykosky, J., Ligon, K. L., Chin, L., Cavenee, W. K., and Furnari, F. (2010) Mutant EGFR is required for maintenance of glioma growth in vivo, and its ablation leads to escape from receptor dependence. *Proceedings of the National Academy of Sciences of the United States of America* **107**, 2616-2621
- Mukherjee, B., McEllin, B., Camacho, C. V., Tomimatsu, N., Sirasanagandala, S., Nannepaga, S., Hatanpaa, K. J., Mickey, B., Madden, C., Maher, E., Boothman, D. A., Furnari, F., Cavenee, W. K., Bachoo, R. M., and Burma, S. (2009) EGFRvIII and DNA double-strand break repair: a molecular mechanism for radioresistance in glioblastoma. *Cancer research* 69, 4252-4259
- 7. Bao, S., Wu, Q., McLendon, R. E., Hao, Y., Shi, Q., Hjelmeland, A. B., Dewhirst, M. W., Bigner, D. D., and Rich, J. N. (2006) Glioma stem cells promote radioresistance by preferential activation of the DNA damage response. *Nature* **444**, 756-760
- 8. Dittmann, K., Mayer, C., Fehrenbacher, B., Schaller, M., Raju, U., Milas, L., Chen, D. J., Kehlbach, R., and Rodemann, H. P. (2005) Radiation-induced epidermal growth factor receptor nuclear import is linked to activation of DNA-

dependent protein kinase. The Journal of biological chemistry 280, 31182-31189

- 9. Liccardi, G., Hartley, J. A., and Hochhauser, D. (2011) EGFR nuclear translocation modulates DNA repair following cisplatin and ionizing radiation treatment. *Cancer research* **71**, 1103-1114
- 10. Li, L., Wang, H., Yang, E. S., Arteaga, C. L., and Xia, F. (2008) Erlotinib attenuates homologous recombinational repair of chromosomal breaks in human breast cancer cells. *Cancer research* **68**, 9141-9146
- 11. Kriegs, M., Kasten-Pisula, U., Rieckmann, T., Holst, K., Saker, J., Dahm-Daphi, J., and Dikomey, E. (2010) The epidermal growth factor receptor modulates DNA double-strand break repair by regulating non-homologous end-joining. *DNA repair* **9**, 889-897
- 12. Mukherjee, B., Choy, H., Nirodi, C., and Burma, S. (2010) Targeting nonhomologous end-joining through epidermal growth factor receptor inhibition: rationale and strategies for radiosensitization. *Seminars in radiation oncology* **20**, 250-257
- Chou, R. H., Wang, Y. N., Hsieh, Y. H., Li, L. Y., Xia, W., Chang, W. C., Chang, L. C., Cheng, C. C., Lai, C. C., Hsu, J. L., Chang, W. J., Chiang, S. Y., Lee, H. J., Liao, H. W., Chuang, P. H., Chen, H. Y., Wang, H. L., Kuo, S. C., Chen, C. H., Yu, Y. L., and Hung, M. C. (2014) EGFR modulates DNA synthesis and repair through Tyr phosphorylation of histone H4. *Developmental cell* **30**, 224-237
- 14. Smith, G. C., and Jackson, S. P. (1999) The DNA-dependent protein kinase. *Genes & development* **13**, 916-934
- 15. Suganuma, T., Pattenden, S. G., and Workman, J. L. (2008) Diverse functions of WD40 repeat proteins in histone recognition. *Genes & development* **22**, 1265-1268
- 16. Panieri, E., and Santoro, M. M. (2016) ROS homeostasis and metabolism: a dangerous liason in cancer cells. *Cell death & disease* **7**, e2253
- 17. Sharma, S. V., Fischbach, M. A., Haber, D. A., and Settleman, J. (2006) "Oncogenic shock": explaining oncogene addiction through differential signal

- Schaar, C. E., Dues, D. J., Spielbauer, K. K., Machiela, E., Cooper, J. F., Senchuk, M., Hekimi, S., and Van Raamsdonk, J. M. (2015) Mitochondrial and cytoplasmic ROS have opposing effects on lifespan. *PLoS genetics* 11, e1004972
- 19. Li, X., Fang, P., Mai, J., Choi, E. T., Wang, H., and Yang, X. F. (2013) Targeting mitochondrial reactive oxygen species as novel therapy for inflammatory diseases and cancers. *Journal of hematology & oncology* **6**, 19
- 20. Ogrodnik, M., Salmonowicz, H., Brown, R., Turkowska, J., Sredniawa, W., Pattabiraman, S., Amen, T., Abraham, A. C., Eichler, N., Lyakhovetsky, R., and Kaganovich, D. (2014) Dynamic JUNQ inclusion bodies are asymmetrically inherited in mammalian cell lines through the asymmetric partitioning of vimentin. *Proceedings of the National Academy of Sciences of the United States of America* **111**, 8049-8054
- 21. Adams, J., Kelso, R., and Cooley, L. (2000) The kelch repeat superfamily of proteins: propellers of cell function. *Trends Cell Biol* **10**, 17-24